Published for SISSA by 2 Springer

RECEIVED: December 16, 2020 REVISED: March 4, 2021 ACCEPTED: March 16, 2021 PUBLISHED: April 29, 2021

# Top, Higgs, diboson and electroweak fit to the Standard Model effective field theory

John Ellis,<sup>*a,b,c*</sup> Maeve Madigan,<sup>*d*</sup> Ken Mimasu,<sup>*a*</sup> Veronica Sanz<sup>*e,f*</sup> and Tevong You<sup>*b,d,g*</sup>

- <sup>c</sup>National Institute of Chemical Physics & Biophysics, Rävala 10, 10143 Tallinn, Estonia
- <sup>d</sup>DAMTP, University of Cambridge, Wilberforce Road, Cambridge CB3 0WA, U.K.
- <sup>e</sup>Instituto de Física Corpuscular (IFIC), Universidad de Valencia-CSIC,
- E-46980 Valencia, Spain
- <sup>f</sup> Department of Physics and Astronomy, University of Sussex, Brighton BN1 9QH, U.K.
- <sup>g</sup> Cavendish Laboratory, University of Cambridge, J.J. Thomson Avenue, Cambridge CB3 0HE, U.K.

*E-mail:* john.ellis@cern.ch, mum20@cam.ac.uk, ken.mimasu@kcl.ac.uk, veronica.sanz@uv.es, tevong.you@cern.ch

ABSTRACT: The search for effective field theory deformations of the Standard Model (SM) is a major goal of particle physics that can benefit from a global approach in the framework of the Standard Model Effective Field Theory (SMEFT). For the first time, we include LHC data on top production and differential distributions together with Higgs production and decay rates and Simplified Template Cross-Section (STXS) measurements in a global fit, as well as precision electroweak and diboson measurements from LEP and the LHC, in a global analysis with SMEFT operators of dimension 6 included linearly. We present the constraints on the coefficients of these operators, both individually and when marginalised, in flavour-universal and top-specific scenarios, studying the interplay of these datasets and the correlations they induce in the SMEFT. We then explore the constraints that our linear SMEFT analysis imposes on specific ultra-violet completions of the Standard Model, including those with single additional fields and low-mass stop squarks. We also present a model-independent search for deformations of the SM that contribute to between two and





<sup>&</sup>lt;sup>a</sup> Theoretical Particle Physics and Cosmology Group, Department of Physics, King's College London, London WC2R 2LS, U.K.

<sup>&</sup>lt;sup>b</sup> Theoretical Physics Department, CERN, CH-1211 Geneva 23, Switzerland

five SMEFT operator coefficients. In no case do we find any significant evidence for physics beyond the SM. Our underlying Fitmaker public code provides a framework for future generalisations of our analysis, including a quadratic treatment of dimension-6 operators.

KEYWORDS: Effective Field Theories, Beyond Standard Model, Higgs Physics

ARXIV EPRINT: 2012.02779

# Contents

1	Int	roduction	1		
<b>2</b>	2 SMEFT framework				
	2.1	Flavour symmetry in the Standard Model	4		
	2.2	SMEFT operators	5		
		2.2.1 Flavour-universal scenario	6		
		2.2.2 Top-specific flavour scenario	7		
3	Dat	taset description	9		
<b>4</b>	$\mathbf{SM}$	IEFT predictions	12		
	4.1	General strategy	12		
	4.2	Higgs production	13		
	4.3	Gauge boson and Higgs decays	15		
	4.4	Diboson data	16		
	4.5	Top data	16		
<b>5</b>	Fitt	ting procedure	17		
6	Glo	obal results	18		
	6.1	Higgs, diboson and electroweak fit	18		
	6.2	Top fit	22		
	6.3	Combined top, Higgs, diboson and electroweak fit	26		
		6.3.1 Operator sensitivities in individual and marginalised fits	26		
		6.3.2 Sensitivities in 'Higgs-only' operator planes	28		
		6.3.3 The triple-gluon operator $C_G$	30		
		6.3.4 Correlation matrix and principal component analysis	33		
7	Сот	nstraints on UV completions	38		
	7.1	Simple tree-level-induced SMEFTs	38		
	7.2	Tree-level SMEFT patterns	42		
	7.3	R-parity-conserving stop squarks at the 1-loop level	44		
	7.4	Survey of combinations of multiple operators	47		
8	Sur	mmary and conclusions	49		
$\mathbf{A}$	Dat	tasets	55		
в	Nu	merical fits with nested sampling	59		

#### 1 Introduction

Experimental tests of the Standard Model (SM) and probes of possible new physics beyond it have been taken to a new level by LHC measurements during its Runs 1 and 2. Previous to the LHC, the most precise tests of the SM were those provided by measurements at LEP, notably at the Z peak [1] and in  $W^+W^-$  diboson production [2–5], and at the Tevatron, notably in measurements of the W [6] and top quark [7]. The LHC has added new classes of precise measurements, notably those of Higgs production and decays [8–17] and the production of the top quark in various modes at Run 1 [18–41] and Run 2 [42–64], the W mass [65], and also triple-gauge coupling measurements in diboson production [66– 70] and Zjj production [71]. The SM is a tight theoretical framework that connects all these measurements. For this reason, it is necessary to take a global approach to the interpretation of its tests and to searches for deviations from its predictions that may be signatures of possible new physics beyond the SM (BSM).

The interactions within the SM are largely constrained by its  $SU(3) \times SU(2) \times U(1)$ gauge invariance, the exceptions being the magnitudes of the Yukawa couplings of fermions to the Higgs boson, and its self-interactions. Currently, all experimental data are consistent with the  $SU(3) \times SU(2) \times U(1)$  assignments of particles that are specified in the SM. It is therefore natural to assume that any BSM interactions must be invariant under  $SU(3) \times SU(2) \times U(1)$  and respect these established representation assignments. The absence of any evidence for additional particles at the LHC or elsewhere suggests that any BSM particles may be significantly heavier than the electroweak scale, as represented by the masses of the W, Z, top quark and Higgs boson. In this case, the low-energy effects of the more massive BSM particles may be approximated by integrating them out to obtain higher-dimensional interactions between the SM fields [72]. In this approach the SM is regarded as an effective theory whose known renormalizable interactions are supplemented by higher-order terms scaled by inverse powers of the BSM mass scale(s) [73, 74]. This SM Effective Field Theory (SMEFT) is a powerful tool for analyzing the consistency of the SM and searching indirectly for possible BSM physics.<sup>1</sup>

There have been intensive theoretical efforts in recent years to formulate consistently the SMEFT and to classify the sets of independent operators appearing at each dimension [73, 74, 81–90]. The first non-redundant basis of dimension-6 operators, known as the Warsaw basis, was laid out in ref. [81], and the matrix of one-loop anomalous dimensions of these operators has been computed in refs. [91–93]. These and other theoretical developments have laid the groundwork for phenomenological SMEFT analyses of available data to go beyond fitting subsets of specific operators towards global fits in a complete basis of operators, beginning with early global SMEFT fits to electroweak precision data [94] that started to include triple-gauge coupling measurements and the Higgs boson [95–97] soon after its discovery [98, 99]. Other LHC fits focused on triple-gauge couplings and/or Higgs measurements for subsets of operators [100–105]. Since then here have been a va-

<sup>&</sup>lt;sup>1</sup>A more general EFT parametrisation in which the electroweak  $SU(2) \times U(1)$  symmetry is non-linearly realised and the Higgs is added as a singlet scalar, known as the Higgs EFT (HEFT), is described, e.g., in ref. [75] with fits given in refs. [76–78]. Its geometric interpretation is developed in refs. [79, 80].

riety of studies of the SMEFT, for example, in electroweak processes [106–111], flavour physics [112–120], low-energy precision data [121–123], diboson measurements [124–131], at dimension 8 [87, 88, 132–136] (where collider positivity constraints are particularly relevant [136–140]), and its connection with UV-complete models, both at tree-level [141–147] and one-loop [148–159]. In particular, the most recent global analyses have set constraints on dimension-6 SMEFT operator coefficients imposed by precision electroweak data from LEP and the Tevatron, together with Higgs and diboson data from the LHC including some from Run 2 [160–163],<sup>2</sup> while separate SMEFT fits of data on the top quark have also been performed [165–171].

We present here the first global dimension-6 SMEFT analysis to include top data and operators in a simultaneous combination of the constraints from the Higgs, electroweak, diboson and top sectors. We use a full set of data from LHC Run 2, in particular the latest Higgs Simplified Template Cross Section (STXS) measurements, differential distributions in WW diboson and  $Z_{jj}$  measurements, and updated top observables including kinematic distributions,  $t\bar{t}$ , single-top and  $t\bar{t}W/Z$  production. In addition to expanding our dataset, improvements over previous fits include a proper computation using SMEFT@NLO [172] of the dimension-6 contributions to Higgs gluon fusion in STXS bins and incorporating the full SMEFT dependence in off-shell Higgs to 4 lepton decays [173]. We also provide a self-consistent treatment of the triple-gluon operator at linear order that had been omitted from our previous fit [97] on the basis of strong constraints at quadratic order [174-176]. We discuss two possible options for the fermion flavour structure, one assuming a flavouruniversal symmetry and the other allowing the coefficients of operators containing thirdgeneration fermions to vary independently through a top-specific flavour symmetry, both of which are explicitly broken only by the dimension-6 operators that induce shifts in Yukawa couplings.

Figure 1 provides a schematic representation of the dependences of these datasets on the 34 dimension-6 operator coefficients in the top-specific flavour scenario introduced in section 2 and defined in tables 1 and 2. The overlaps between the different oblongs show explicitly how a given operator may contribute to several classes of measurements. For example,  $\mathcal{O}_{Ht}, \mathcal{O}_{tW}, \mathcal{O}_{tB}$  and  $\mathcal{O}_{Qq}^{3,1}$  contribute to top EW observables, i.e., single-top and W helicity fraction measurements, as well as  $t\bar{t}V$  observables. In addition to these observables,  $\mathcal{O}_{Hu}, \mathcal{O}_{Hd}, \mathcal{O}_{Hq}^{(3)}$  and  $\mathcal{O}_{Hq}^{(1)}$  also contribute to diboson, electroweak precision and Higgs observables, exemplifying the interplay between the various data sectors.

Our analysis is based on a newly-developed public code called Fitmaker that is to be continuously updated in the future, with the goal of continuing to provide an adaptable, flexible and extensible framework for global SMEFT fits over the longer term. We have implemented both a fast analytical method for linear-order fits and a Markov Chain Monte Carlo (MCMC) procedure that is necessary to incorporate positivity priors in operator coefficients, as appear in specific BSM scenarios.

The global analysis we present here is performed at linear order in the SMEFT operator coefficients. It makes manifest the interplay between Higgs and top observables, as well as the electroweak and diboson data, and its calculational speed makes possible a broad-band

<sup>&</sup>lt;sup>2</sup>See refs. [15, 164] for recent SMEFT interpretations of the Higgs by ATLAS and CMS.



**Figure 1**. Schematic representation of the datasets and their overlapping dependences on the 34 Wilson coefficients included in our analysis.

search for physics beyond the Standard Model. The linear constraints can be viewed as provisional for operators where quadratic contributions are non-negligible. Nevertheless, keeping those operators in the global fit typically yields conservative marginalised limits and allows one to assess where the impact on other operators can be significant to a first approximation. Our framework is not limited in principle to linear order: we discuss in the text the possible impact of quadratic effects in the top sector and illustrate in an appendix focusing on Higgs data how our procedure can be used for quadratic fits as well, though a global analysis at quadratic order is beyond the scope of this paper. We note that the quadratic contributions can be dominant for strongly-coupled new physics, though dimension-8 operators would in general be relevant at the same order in the new physics scale, and could be included relatively economically in our framework to linear order.

The layout of our paper is as follows. Section 2 reviews the SMEFT framework we use, and sets out the flavour-universal scenario with  $SU(3)^5$  symmetry and the top-specific scenario with  $SU(2)^2 \times SU(3)^5$  symmetry that we analyze. Section 3 summarizes the dataset that we use in the global fit, which is described in more detail in appendix A. Section 4 describes how theory predictions are calculated, and section 5 sets out our fitting procedure. This is described in more detail in appendix B, including a nested-sampling calculational method that we illustrate in an analysis of Higgs signal strengths to quadratic order in the dimension-6 operator coefficients. The results from our global linear fit are presented in section 6, where we display results from the electroweak, Higgs and top sectors separately and in combination. In both the flavour-universal  $SU(3)^5$  symmetric and top-specific  $SU(2)^2 \times SU(3)^5$  scenario, we derive constraints for all individual operators and also constraints including all dimension-6 operators and marginalising. Our principal results are displayed in figure 6, with numerical results for the  $SU(2)^2 \times SU(3)^3$  top-specific scenario presented in table 6. We include dedicated discussions of sensitivities in 'Higgs-only' operator planes that illustrates the impact of the top data, and of the triple-gluon operator coefficient  $C_G$ . We also analyze correlations between the coefficients of the 34 operators in the top-specific scenario and perform a principal component analysis, identifying the most and least constrained combinations of SMEFT operators. Applications of our analysis to some specific BSM scenarios are presented in section 7, including single-field extensions of the SM and a light-stop scenario, and a survey of fits with contributions from any combination of two, three, four or five dimension-6 operators, none of which provide any significant evidence for physics beyond the SM. Our conclusions are summarised in section 8.

# 2 SMEFT framework

We describe in this section the dimension-6 SMEFT framework that we use for our analysis.<sup>3</sup> After reviewing flavour symmetry in the SM, we classify accordingly the dimension-6 operators of the SMEFT [119] and describe two flavour symmetry assumptions that we adopt for our global fit, namely flavour-universal and top-specific scenarios.

# 2.1 Flavour symmetry in the Standard Model

In the absence of the Yukawa interactions, the SM Lagrangian has a  $U(3)^5$  global symmetry [180, 181] associated with independent unitary transformations among the flavour components of its five types of fermions: the left-handed quark doublets, right-handed charge +2/3 and -1/3 quarks, left-handed lepton doublets and right-handed charged leptons. This symmetry can be decomposed into five SU(3) flavour rotations and five U(1) symmetries: baryon and lepton numbers, hypercharge, a Peccei-Quinn symmetry and independent rotations of the right-handed leptons:

$$\mathcal{G} = \mathrm{SU}(3)_q \times \mathrm{SU}(3)_u \times \mathrm{SU}(3)_d \times \mathrm{SU}(3)_l \times \mathrm{SU}(3)_e \times \mathrm{U}(1)^5.$$
(2.1)

The fermion representations transform as follows under the SU(3) rotations:

$$q^i \to U_q^{ij} q^j \,, \, u^i \to U_u^{ij} u^j \,, \, d^i \to U_d^{ij} d^j \,, \, l^i \to U_l^{ij} l^j \,, \, e^i \to U_e^{ij} e^j \,, \tag{2.2}$$

with  $U_x \in SU(3)_x$ . However, the SM Yukawa Lagrangian

$$\mathcal{L}_{\text{Yuk.}} = \lambda_d^{ij}(\bar{q}_i H)d_j + \lambda_u^{ij}(\bar{q}_i \tilde{H})u_j + \lambda_e^{ij}(\bar{l}_i H)e_j + \text{h.c.}$$
(2.3)

is manifestly not invariant under this set of transformations. The down-type Yukawa interactions break  $SU(2)_q \times SU(2)_d$ , the up-type interactions break  $SU(2)_q \times SU(2)_u$  and the lepton Yukawa interactions break  $SU(2)_l \times SU(2)_e$ .

One can recover the complete flavour symmetry in the presence of the Yukawa interactions by introducing three spurions that take the places of the Yukawa matrices and transform as follows:

$$Y_u \to U_q Y_u U_u^{\dagger}, \qquad Y_d \to U_q Y_d U_d^{\dagger}, \qquad Y_e \to U_l Y_e U_e^{\dagger}, \qquad (2.4)$$

such that the Yukawa Lagrangian becomes

$$\mathcal{L}_{\text{Yuk.}} = Y_d^{ij}(\bar{q}_i H) d_j + Y_u^{ij}(\bar{q}_i \tilde{H}) u_j + Y_e^{ij}(\bar{l}_i H) e_j + \text{h.c.}$$
(2.5)

<sup>&</sup>lt;sup>3</sup>For some reviews of the SMEFT, see, e.g., refs. [177–179].

The spurions take background values that can be rotated using flavour symmetry to a basis in which the down-type quark and lepton mass terms are diagonal and the up-type quark mass term is diagonal up to a factor of the CKM matrix, V:

$$\langle Y_d \rangle^{ij} = y_d^{ij} \propto m_d^{ij}, \ \langle Y_e \rangle^{ij} = y_e^{ij} \propto m_e^{ij}, \ \langle Y_u \rangle^{ij} = (V^{\dagger} y_u)^{ij} \propto (V^{\dagger})^{ik} m_u^{kj}.$$
(2.6)

After electroweak symmetry breaking (EWSB), the remaining V factor can be moved to the charged-current gauge interaction terms by an independent rotation of the left-handed up-type quark field,<sup>4</sup>  $u_L^i \to (V^{\dagger})^{ik} u_L^k$ .

According to the Minimal Flavour Violation (MFV) hypothesis [182], no new sources of flavour violation exist beyond the SM Yukawa couplings. It assumes that flavour-violating effects in, e.g., higher-dimensional operators, appear with an associated Yukawa matrix via spurion insertions. This can be taken as a prior for the expected sizes of the operator coefficients. Often the operators are defined to absorb a Yukawa factor determined by the spurions but here we shall consider the coefficients of the operators that break  $SU(3)^5$ , such as those that induce shifts in the Yukawas, as explicit breakings of the flavour symmetry, without these spurion insertions. This simplifies the normalisation of the operators to make the interpretation of their coefficients clear and unambiguous.

# 2.2 SMEFT operators

The SM Lagrangian is only the leading approximation, consisting of operators up to dimension 4, to an EFT that also includes higher-dimensional operators. The lepton numberviolating Weinberg operator [73] is the unique one at dimension 5. At dimension 6, we may write the SMEFT Lagrangian as

$$\mathcal{L}_{\text{SMEFT}} = \mathcal{L}_{\text{SM}} + \sum_{i=1}^{2499} \frac{C_i}{\Lambda^2} \mathcal{O}_i \,, \qquad (2.7)$$

where  $\Lambda$  is a dimensionful scale and  $C_i$  are the dimensionless Wilson coefficients. The dimension-6 operators  $\mathcal{O}_i$  in the Warsaw basis are given in table 1 (adapted from ref. [81]). Those with only bosonic fields are invariant under flavour symmetry, whereas those containing fermions can be classified in terms of their transformation properties under SU(3)<sup>5</sup>. Any off-diagonal flavour entry violates SU(3)<sup>5</sup> explicitly, as do terms with both a left- and a right-handed field, even in their flavour-diagonal entries. For terms involving two fermions, these include the Yukawa ( $\psi^2 H^3$ ) and dipole ( $\psi^2 XH$ ) operators as well as the righthanded charged-current operator ( $\mathcal{O}_{Hud}$ ). The four-fermion sector includes the ( $\bar{L}R$ )( $\bar{R}L$ ), ( $\bar{L}R$ )( $\bar{L}R$ ) and *B*-violating classes of operators. All other operators containing fermionic currents are SU(3)<sup>5</sup>-invariant in the flavour-universal case, i.e., when the diagonal entries have a common Wilson coefficient and off-diagonal entries vanish.

<sup>&</sup>lt;sup>4</sup>The opposite convention is sometimes used, with a diagonal up-type mass matrix before EWSB.

$\subseteq$
H
E
Ю
$\bigcirc$
ł
$\frown$
$\mathbb{N}$
$\bigcirc$
$\mathbb{N}$
$\vdash$
$\checkmark$
$\mathbb{N}$
$\bigtriangledown$
9

$X^3$		$H^6$ and $H^4D^2$		$\psi^2 H^3$		
$\mathcal{O}_{G}$	$f^{ABC}G^{A u}_{\mu}G^{B ho}_{ u}G^{C\mu}_{ ho}$	$\mathcal{O}_{H}$	$(H^{\dagger}H)^3$	$\mathcal{O}_{eH}$	$(H^{\dagger}H)(\bar{l}_{p}e_{r}H)$	
$\mathcal{O}_{\widetilde{G}}$	$f^{ABC} \tilde{G}^{A\nu}_{\mu} G^{B\rho}_{\nu} G^{C\mu}_{\rho}$	$\mathcal{O}_{H\square}$	$(H^{\dagger}H)\Box(H^{\dagger}H)$	$\mathcal{O}_{uH}$	$(H^{\dagger}H)(\bar{q}_{p}u_{r}\tilde{H})$	
$\mathcal{O}_W$	$\varepsilon^{IJK}W^{I\nu}_{\mu}W^{J\rho}_{\nu}W^{K\mu}_{\rho}$	$\mathcal{O}_{HD}$	$\left(H^{\dagger}D^{\mu}H\right)^{\star}\left(H^{\dagger}D_{\mu}H\right)$	$\mathcal{O}_{_{dH}}$	$(H^{\dagger}H)(\bar{q}_p d_r H)$	
$\mathcal{O}_{\widetilde{W}}$	$\varepsilon^{IJK}\widetilde{W}^{I\nu}_{\mu}W^{J\rho}_{\nu}W^{K\mu}_{\rho}$					
$X^2H^2$		$\psi^2 X H$		$\psi^2 H^2 D$		
$\mathcal{O}_{HG}$	$H^{\dagger}HG^{A}_{\mu\nu}G^{A\mu\nu}$	$\mathcal{O}_{eW}$	$(\bar{l}_p \sigma^{\mu\nu} e_r) \tau^I H W^I_{\mu\nu}$	$\mathcal{O}_{Hl}^{(1)}$	$(H^{\dagger}i\overset{\leftrightarrow}{D}_{\mu}H)(\bar{l}_{p}\gamma^{\mu}l_{r})$	
$\mathcal{O}_{_{H\widetilde{G}}}$	$H^{\dagger}H\widetilde{G}^{A}_{\mu u}G^{A\mu u}$	$\mathcal{O}_{eB}$	$(\bar{l}_p \sigma^{\mu\nu} e_r) H B_{\mu\nu}$	$\mathcal{O}_{Hl}^{(3)}$	$(H^{\dagger}i \overleftrightarrow{D}^{I}_{\mu} H)(\bar{l}_{p}\tau^{I}\gamma^{\mu}l_{r})$	
$\mathcal{O}_{HW}$	$H^{\dagger}HW^{I}_{\mu\nu}W^{I\mu\nu}$	$\mathcal{O}_{uG}$	$(\bar{q}_p \sigma^{\mu\nu} T^A u_r) \tilde{H} G^A_{\mu\nu}$	$\mathcal{O}_{_{He}}$	$(H^{\dagger}i \overleftrightarrow{D}_{\mu} H)(\bar{e}_p \gamma^{\mu} e_r)$	
$\mathcal{O}_{_{H\widetilde{W}}}$	$H^{\dagger}H\widetilde{W}^{I}_{\mu\nu}W^{I\mu\nu}$	$\mathcal{O}_{uW}$	$(\bar{q}_p \sigma^{\mu\nu} u_r) \tau^I \tilde{H} W^I_{\mu\nu}$	$\mathcal{O}_{Hq}^{(1)}$	$(H^{\dagger}i \overleftrightarrow{D}_{\mu} H)(\bar{q}_p \gamma^{\mu} q_r)$	
$\mathcal{O}_{HB}$	$H^{\dagger}HB_{\mu u}B^{\mu u}$	$\mathcal{O}_{uB}$	$(\bar{q}_p \sigma^{\mu\nu} u_r) \widetilde{H} B_{\mu\nu}$	$\mathcal{O}_{Hq}^{(3)}$	$(H^{\dagger}i \overleftrightarrow{D}_{\mu}^{I} H)(\bar{q}_{p}\tau^{I}\gamma^{\mu}q_{r})$	
$\mathcal{O}_{_{H\widetilde{B}}}$	$H^{\dagger}H\widetilde{B}_{\mu u}B^{\mu u}$	$\mathcal{O}_{dG}$	$(\bar{q}_p \sigma^{\mu\nu} T^A d_r) H G^A_{\mu\nu}$	$\mathcal{O}_{Hu}$	$(H^{\dagger}i \overleftrightarrow{D}_{\mu} H)(\bar{u}_p \gamma^{\mu} u_r)$	
$\mathcal{O}_{HWB}$	$H^{\dagger}\tau^{I}HW^{I}_{\mu\nu}B^{\mu\nu}$	$\mathcal{O}_{dW}$	$(\bar{q}_p \sigma^{\mu\nu} d_r) \tau^I H W^I_{\mu\nu}$	$\mathcal{O}_{_{Hd}}$	$(H^{\dagger}i \overset{\rightarrowtail}{D}_{\mu} H)(\bar{d}_p \gamma^{\mu} d_r)$	
$\mathcal{O}_{_{H\widetilde{W}B}}$	$H^{\dagger}\tau^{I}H\widetilde{W}^{I}_{\mu\nu}B^{\mu\nu}$	$\mathcal{O}_{dB}$	$(\bar{q}_p \sigma^{\mu\nu} d_r) H B_{\mu\nu}$	$\mathcal{O}_{Hud}$	$i(\tilde{H}^{\dagger}D_{\mu}H)(\bar{u}_{p}\gamma^{\mu}d_{r})$	
$(\bar{L}L)(\bar{L}L)$			$(ar{R}R)(ar{R}R)$		$(\bar{L}L)(\bar{R}R)$	
$\mathcal{O}_{ll}$	$(\bar{l}_p \gamma_\mu l_r) (\bar{l}_s \gamma^\mu l_t)$	$\mathcal{O}_{ee}$	$(\bar{e}_p \gamma_\mu e_r) (\bar{e}_s \gamma^\mu e_t)$	$\mathcal{O}_{le}$	$(\bar{l}_p \gamma_\mu l_r) (\bar{e}_s \gamma^\mu e_t)$	
$\mathcal{O}_{qq}^{(1)}$	$(\bar{q}_p \gamma_\mu q_r)(\bar{q}_s \gamma^\mu q_t)$	$\mathcal{O}_{uu}$	$(\bar{u}_p \gamma_\mu u_r)(\bar{u}_s \gamma^\mu u_t)$	$\mathcal{O}_{lu}$	$(\bar{l}_p \gamma_\mu l_r) (\bar{u}_s \gamma^\mu u_t)$	
$\mathcal{O}_{qq}^{(3)}$	$(\bar{q}_p \gamma_\mu \tau^I q_r) (\bar{q}_s \gamma^\mu \tau^I q_t)$	$\mathcal{O}_{dd}$	$(\bar{d}_p \gamma_\mu d_r) (\bar{d}_s \gamma^\mu d_t)$	$\mathcal{O}_{ld}$	$(\bar{l}_p \gamma_\mu l_r) (\bar{d}_s \gamma^\mu d_t)$	
$\mathcal{O}_{lq}^{(1)}$	$(\bar{l}_p \gamma_\mu l_r)(\bar{q}_s \gamma^\mu q_t)$	$\mathcal{O}_{eu}$	$(\bar{e}_p \gamma_\mu e_r)(\bar{u}_s \gamma^\mu u_t)$	$\mathcal{O}_{qe}$	$(\bar{q}_p \gamma_\mu q_r) (\bar{e}_s \gamma^\mu e_t)$	
$\mathcal{O}_{lq}^{(3)}$	$(\bar{l}_p \gamma_\mu \tau^I l_r)(\bar{q}_s \gamma^\mu \tau^I q_t)$	$\mathcal{O}_{ed}$	$(\bar{e}_p \gamma_\mu e_r) (\bar{d}_s \gamma^\mu d_t)$	$\mathcal{O}_{qu}^{(1)}$	$(\bar{q}_p \gamma_\mu q_r)(\bar{u}_s \gamma^\mu u_t)$	
		$\mathcal{O}_{ud}^{(1)}$	$(\bar{u}_p \gamma_\mu u_r) (\bar{d}_s \gamma^\mu d_t)$	$\mathcal{O}_{qu}^{(8)}$	$(\bar{q}_p \gamma_\mu T^A q_r)(\bar{u}_s \gamma^\mu T^A u_t)$	
		$\mathcal{O}_{ud}^{(8)}$	$(\bar{u}_p \gamma_\mu T^A u_r) (\bar{d}_s \gamma^\mu T^A d_t)$	$\mathcal{O}_{qd}^{(1)}$	$(\bar{q}_p\gamma_\mu q_r)(\bar{d}_s\gamma^\mu d_t)$	
				$\mathcal{O}_{qd}^{(8)}$	$(\bar{q}_p \gamma_\mu T^A q_r) (\bar{d}_s \gamma^\mu T^A d_t)$	
$(\bar{L}R)(\bar{R}L)$ and $(\bar{L}R)(\bar{L}R)$			<i>B</i> -violating			
$\mathcal{O}_{ledq}$	$(ar{l}_p^j e_r) (ar{d}_s q_t^j)$	$\mathcal{O}_{duq}$	$\varepsilon^{\alpha\beta\gamma}\varepsilon_{jk}\left[(d_p^{\alpha})^T C u_r^{\beta}\right]\left[(q_s^{\gamma j})^T C l_t^k\right]$			
$\mathcal{O}_{quqd}^{(1)}$	$(\bar{q}_p^j u_r) \varepsilon_{jk} (\bar{q}_s^k d_t)$	$\mathcal{O}_{qqu}$	$\varepsilon^{\alpha\beta\gamma}\varepsilon_{jk}\left[(q_p^{\alpha j})^T C q_r^{\beta k}\right] \left[(u_s^{\gamma})^T C e_t\right]$			
$\mathcal{O}_{quqd}^{(8)}$	$(\bar{q}_p^j T^A u_r) \varepsilon_{jk} (\bar{q}_s^k T^A d_t)$	$\mathcal{O}_{qqq}$	$\varepsilon^{\alpha\beta\gamma}\varepsilon_{jn}\varepsilon_{km}\left[(q_p^{\alpha j})^T C q_r^{\beta k}\right]\left[(q_s^{\gamma m})^T C l_t^n\right]$			
$\mathcal{O}_{lequ}^{(1)}$	$(\bar{l}_p^j e_r) \varepsilon_{jk} (\bar{q}_s^k u_t)$	$\mathcal{O}_{duu}$	$arepsilon^{lphaeta\gamma}\left[(d_p^{lpha})^T C u_r^{eta} ight]\left[(u_s^{\gamma})^T C e_t ight]$			
$\mathcal{O}_{lequ}^{(3)}$	$(\bar{l}_p^j \sigma_{\mu\nu} e_r) \varepsilon_{jk} (\bar{q}_s^k \sigma^{\mu\nu} u_t)$			J L		

**Table 1**. Dimension-6 operators in the Warsaw basis, adapted from ref. [81]. The grey cells indicate operators that break flavour  $SU(3)^5$  explicitly.

# 2.2.1 Flavour-universal scenario

Assuming  $SU(3)^5$  symmetry reduces the Warsaw basis to the flavour-universal scenario in which only the operators in cells not shaded grey in table 1 are allowed, with common flavour-diagonal Wilson coefficients and no off-diagonal entries. Neglecting CP-violating interactions, one is left with 31 degrees of freedom, of which 16 are relevant for a leadingorder fit to electroweak precision, diboson and Higgs data. To this we also add 4 operators that explicitly break the flavour symmetry and affect Higgs physics through a shift of the tau, muon, b-quark and top-quark Yukawa couplings. The 20 operators in our "flavour-universal" scenario are then

EWPO:	$\mathcal{O}_{HWB},\mathcal{O}_{HD},\mathcal{O}_{ll},\mathcal{O}_{Hl}^{(3)},\mathcal{O}_{Hl}^{(1)},\mathcal{O}_{He},\mathcal{O}_{Hq}^{(3)},\mathcal{O}_{Hq}^{(1)},\mathcal{O}_{Hd},\mathcal{O}_{Hu},$	
Bosonic:	$\mathcal{O}_{H\Box},\mathcal{O}_{HG},\mathcal{O}_{HW},\mathcal{O}_{HB},\mathcal{O}_{W},\mathcal{O}_{G},$	
Yukawa:	$\mathcal{O}_{ au H},\mathcal{O}_{\mu H},\mathcal{O}_{b H},\mathcal{O}_{t H}.$	(2.8)

We have categorised these operators roughly, into sets that are mostly constrained by electroweak precision observables (EWPO), those that can only be constrained at treelevel by Higgs and diboson measurements (bosonic), and operators that induce shifts in the Yukawa couplings (Yukawa).

## 2.2.2 Top-specific flavour scenario

The minimal flavour scenario that singles out top-quark couplings relaxes the  $SU(3)^5$  symmetry as follows [183]:

$$\begin{split} \mathrm{SU}(3)^5 &\to \mathrm{SU}(2)^2 \times \mathrm{SU}(3)^3 \\ &= \mathrm{SU}(2)_q \times \mathrm{SU}(2)_u \times \mathrm{SU}(3)_d \times \mathrm{SU}(3)_l \times \mathrm{SU}(3)_e \,. \end{split}$$

This allows chirality-flipping interaction terms involving the third-generation quark doublet and right-handed up-type fields, notably the top-quark Yukawa interaction. The following three additional dimension-6 operators in the SMEFT are now allowed:<sup>5</sup>

$$[\mathcal{O}_{uG}]^{33} = \mathcal{O}_{tG}, \quad [\mathcal{O}_{uB}]^{33} = \mathcal{O}_{tB}, \quad [\mathcal{O}_{uW}]^{33} = \mathcal{O}_{tW}.$$
 (2.9)

The flavour-universality conditions on operators in the  $\psi^2 H^2 D$ ,  $\bar{L}L\bar{L}L$ ,  $\bar{R}R\bar{R}R$  and  $\bar{L}L\bar{R}R$  classes that contain q or u are also relaxed. Schematically,

$$C_{\text{univ.}} \sum_{i=1,2,3} \mathcal{K}^{\mu} \bar{f}_{i} \gamma_{\mu} f_{i} \rightarrow \begin{cases} C_{3} \mathcal{K}^{\mu} \bar{f}_{3} \gamma_{\mu} f_{3}, \\ C_{\text{univ.}} \sum_{j=1,2(,3)} \mathcal{K}^{\mu} \bar{f}_{j} \gamma_{\mu} f_{j}, \end{cases}$$
(2.10)

where  $\mathcal{K}^{\mu}$  is a combination of other fields. Here a choice must be made in the second line of eq. (2.10) whether to split the degrees of freedom into a fully-universal operator that preserves the full flavour symmetry or an operator that respects only the reduced symmetry  $SU(2)^2 \times SU(3)^3$ , corresponding to keeping or removing the index in red, respectively. The two are related by a basis rotation. We adopt the second option, since it better separates the degrees of freedom that affect only top measurements from the rest. The  $\psi^2 H^2 D$  class

<sup>&</sup>lt;sup>5</sup>The analogous operator  $[\mathcal{O}_{uH}]^{33} = \mathcal{O}_{tH}$  is already included in the set of Yukawa operators listed in eq. (2.8).

grows to

$$\begin{bmatrix} \mathcal{O}_{Hq}^{(1)} \end{bmatrix} \to \begin{bmatrix} \mathcal{O}_{Hq}^{(1)} \end{bmatrix}^{j,j} \text{ and } \begin{bmatrix} \mathcal{O}_{Hq}^{(1)} \end{bmatrix}^{3,3} = \{ \mathcal{O}_{Hq_i}^{(1)}, \mathcal{O}_{HQ}^{(1)} \} , \\ \begin{bmatrix} \mathcal{O}_{Hq}^{(3)} \end{bmatrix} \to \begin{bmatrix} \mathcal{O}_{Hq}^{(3)} \end{bmatrix}^{j,j} \text{ and } \begin{bmatrix} \mathcal{O}_{Hq}^{(3)} \end{bmatrix}^{3,3} = \{ \mathcal{O}_{Hq_i}^{(3)}, \mathcal{O}_{HQ}^{(3)} \} , \\ \begin{bmatrix} \mathcal{O}_{Hu} \end{bmatrix} \to \begin{bmatrix} \mathcal{O}_{Hu} \end{bmatrix}^{j,j} \text{ and } \begin{bmatrix} \mathcal{O}_{Hq} \end{bmatrix}^{3,3} = \{ \mathcal{O}_{Hu_i}, \mathcal{O}_{HQ} \} ,$$
 (2.11)

where Q denotes here the third-generation quark doublet. Four-fermion operators are split generically into 'four-light', 'two-heavy-two-light' and 'four-heavy' flavour components, where 'light' and 'heavy' denote the first two and the third generations, respectively. The four-light degrees of freedom are the same as the four-fermion operators of the SU(3)<sup>5</sup> scenario, except that they are flavour-universal over three generations for d, l and e, and only the first two generations for q and u. The classification of the additional four-fermion operators under this generalisation is slightly more involved, due to the permutation symmetries on the flavour indices as well as the equivalence of certain degrees of freedom via Fierz identities. This is discussed in ref. [183], where a 'dim6top' basis is chosen for the operators involving top fields with LHC top physics observables in mind. The new operators are shown in table 2 with their definitions in terms of the Warsaw basis coefficients.

A total of 31 new CP-conserving degrees of freedom are introduced by this relaxation of the flavour symmetry. However, our analysis is only sensitive to a subset of these, for two main reasons. First, our chosen dataset does not constrain a number of the operators allowed by this flavour assumption. These include all flavour-universal four-light fermion operators, which are constrained by electron-positron collider data and numerous low-energy scattering and decay experiments (see ref. [122] for a recent compilation of constraints), as well as high-energy Drell-Yan and dijet observables at hadron colliders [184–189]. Furthermore, it is also effectively blind to the two-heavy quark two-lepton class of operators listed in table 2. Although many of them mediate the same final states as those selected by, e.g.,  $t\bar{t}V$  measurements, they do so in the absence of a resonant intermediate W or Z boson decaying into the lepton pair. So far, searches have implemented selections to enhance this resonant contribution, and are therefore not sensitive to the non-resonant phase space populated by the operators in question. The second important feature of our analysis is that it is restricted to the linear,  $\mathcal{O}(1/\Lambda^2)$ , level in the EFT expansion. This restricts the sensitivity to the set of operators that interfere appreciably with the dominant SM amplitudes for the processes of interest. This is not the case for the six neutral-current mediating, two-heavy two-light operators in the upper left section of table 2. These operators mediate  $q\bar{q} \rightarrow t\bar{t}$  production in the colour-singlet channel, which does not interfere at LO with the strongly-dominant SM QCD contribution. In contrast, the corresponding charged-current operator affects single-top quark production that, being an EW process in the SM, does have such an interference term. Finally, we also omit the four-heavy operators that would mainly be constrained by  $t\bar{t}bb$  measurements and four-top production searches. These data have been shown to be largely sensitive, at present, to the quadratic EFT contributions [168, 190–192], and our analysis would not yield meaningful bounds in these directions. We therefore only include in our analysis 8 two-heavy two-light quark degrees of freedom: the colour-singlet, charged-current operator and seven neutral, colour-octet operators.

To summarise, the 34 operators relevant for our leading-order, linear fit in the topspecific flavour scenario are the 20 listed in eqs. (2.8) plus the 14 discussed above (three in (2.9), three more in (2.11) and eight two-light two-heavy quark operators):

EWPO:	$\mathcal{O}_{HWB},\mathcal{O}_{HD},\mathcal{O}_{ll},\mathcal{O}_{Hl}^{(3)},\mathcal{O}_{Hl}^{(1)},\mathcal{O}_{He},\mathcal{O}_{Hq}^{(3)},\mathcal{O}_{Hq}^{(1)},\mathcal{O}_{Hd},\mathcal{O}_{Hu}$	,
Bosonic:	$\mathcal{O}_{H\Box},\mathcal{O}_{HG},\mathcal{O}_{HW},\mathcal{O}_{HB},\mathcal{O}_{W},\mathcal{O}_{G},$	
Yukawa:	$\mathcal{O}_{ au H},\mathcal{O}_{\mu H},\mathcal{O}_{b H},\mathcal{O}_{t H},$	
Top 2F:	${\cal O}_{HQ}^{(3)},{\cal O}_{HQ}^{(1)},{\cal O}_{Ht},{\cal O}_{tG},{\cal O}_{tW},{\cal O}_{tB},$	
Top 4F:	$\mathcal{O}_{Qq}^{3,1},\mathcal{O}_{Qq}^{3,8},\mathcal{O}_{Qq}^{1,8},\mathcal{O}_{Qu}^{8},\mathcal{O}_{Qd}^{8},\mathcal{O}_{tQ}^{8},\mathcal{O}_{tu}^{8},\mathcal{O}_{td}^{8}.$	(2.12)

These are grouped into top operators involving two (top 2F) and four (top 4F) heavy fermions, respectively.

#### 3 Dataset description

In this section we describe the data used in our global fit. We summarise here the main categories of data and refer the reader to appendix A for a complete list of the observables that have been implemented in Fitmaker, together with their source references.

The most precise electroweak measurements, other than the W mass, remain those from LEP and the SLC.<sup>6</sup> The Higgs boson discovery at the LHC enabled the possibility of a closed global SMEFT fit to a complete set of dimension-6 operators for the first time. Higgs physics has since progressed rapidly to include more channels and sub-categories beyond signal strengths. In particular, the STXS categorisations of the various Higgs production sub-channels provide further sensitivity to different directions in the fit, as illustrated, for example, in figure 2 below for the case of gluon fusion and described further in the next section.

The higher energies at the LHC also allow certain measurements of diboson and dilepton final states to become competitive with LEP [126, 185–189, 196], enable complementary probes of higher-dimensional operators [111, 197], and, moreover, give access to top physics with higher statistics than ever before, including the previously unreachable  $t\bar{t}W/Z/H$  and other, rare production processes such as four-top production [168, 190–192]. More operators, under less restrictive flavour assumptions, can then be included in a global SMEFT fit. This is particularly motivated since the top quark is often expected to be more sensitive to BSM physics.

The following is a summary of the different categories of observables that we consider — see appendix A for more details and references. We build our selected dataset by combining statistically independent measurements, including correlation information by means of published covariance/correlation matrices, when available.<sup>7</sup> In general, for LHC

 $<sup>^{6}</sup>$ We note that global SMEFT fits would benefit greatly from a future Z-pole run [193–195].

<sup>&</sup>lt;sup>7</sup>See ref. [198] for a study of the impact of correlations in global fits.

$\mathcal{O}_i$	$C_i$ Definition	$\mathcal{O}_i$	$C_i$ Definition	
4 quark (2 heavy 2 light)				
$\mathcal{O}_{Qq}^{1,1}$	$\sum_{i=1,2} \left( [C_{qq}^{(1)}]^{ii33} + \frac{1}{6} [C_{qq}^{(1)}]^{i33i} + \frac{1}{2} [C_{qq}^{(3)}]^{i33i} \right)$	$\mathcal{O}_{Qq}^{1,8}$	$\sum_{i=1,2} \left( [C_{qq}^{(1)}]^{i33i} + 3[C_{qq}^{(3)}]^{i33i} \right)$	
$\mathcal{O}_{Qq}^{3,1}$	$\sum_{i=1,2} \left( [C_{qq}^{(3)}]^{ii33} + \frac{1}{6} [C_{qq}^{(1)}]^{i33i} - \frac{1}{6} [C_{qq}^{(3)}]^{i33i} \right)$	$\mathcal{O}_{Qq}^{3,8}$	$\sum_{i=1,2} \left( [C_{qq}^{(1)}]^{i33i} - [C_{qq}^{(3)}]^{i33i} \right)$	
$\mathcal{O}_{tu}^1$	$\sum_{i=1,2} \left( [C_{uu}]^{ii33} + \frac{1}{3} [C_{uu}]^{i33i} \right)$	$\mathcal{O}_{tu}^{8}$	$\sum_{i=1,2} 2[C_{uu}]^{i33i}$	
$\mathcal{O}_{td}^1$	$\sum_{i=1,2(,3)} [C_{ud}^{(1)}]^{33ii}$	$\mathcal{O}_{td}^{8}$	$\sum_{i=1,2(,3)} [C_{ud}^{(8)}]^{33ii}$	
$\mathcal{O}_{tq}^{1}$	$\sum_{i=1,2} [C_{qu}^{(1)}]^{ii33}$	$\mathcal{O}^{8}_{tq}$	$\sum_{i=1,2} [C_{qu}^{(8)}]^{ii33}$	
$\mathcal{O}_{Qu}^{1}$	$\sum_{i=1,2} [C_{qu}^{(1)}]^{33ii}$	$\mathcal{O}_{Qu}^{8}$	$\sum_{i=1,2} [C_{qu}^{(8)}]^{33ii}$	
${\cal O}^{1}_{Qd}$	$\sum_{i=1,2(,3)} [C_{qd}^{(1)}]^{33ii}$	$\mathcal{O}^{8}_{Qd}$	$\sum_{i=1,2(,3)} [C_{qd}^{(8)}]^{33ii}$	
4 quark	k (4 heavy)			
$\mathcal{O}_{QQ}^{1}$	$2[C_{qq}^{(1)}]^{3333} - \frac{2}{3}[C_{qq}^{(3)}]^{3333}$	$\mathcal{O}^8_{QQ}$	$8[C_{qq}^{(3)}]^{3333}$	
$\mathcal{O}_{Qt}^1$	$[C_{qu}^{(1)}]^{3333}$	$\mathcal{O}_{Qt}^{8}$	$[C_{qu}^{(8)}]^{3333}$	
$\mathcal{O}_{tt}$	$[C_{uu}^{(1)}]^{_{3333}}$			
2 heavy 2 lepton				
$\mathcal{O}_{Ql}^{-(1)}$	$\sum_{i=1,2,3} [C_{lq}^1]^{ii33} - [C_{lq}^3]^{ii33}$	$\mathcal{O}_{tl}^{(1)}$	$\sum_{i=1,2,3} [C_{lu}]^{ii33}$	
${\cal O}_{Ql}^{3(1)}$	$\sum_{i=1,2,3} [C_{lq}^3]^{ii33}$	$\mathcal{O}_{te}^{(1)}$	$\sum_{i=1,2,3} [C_{eu}]^{ii33}$	
$\mathcal{O}_{Qe}^{(1)}$	$\sum_{i=1,2,3} [C_{eQ}]^{ii33}$			

Table 2. Four-fermion operators containing at least one third-generation bilinear in the 'dim6top' basis [183] assuming an  $SU(2)^2 \times SU(3)^3$  flavour symmetry. The relations of the corresponding Wilson coefficients with those of the Warsaw basis are also shown. The shaded entries indicate operators that are not included in our analysis because significant constraints cannot be obtained from the chosen dataset at leading order and linear level in the EFT expansion, as discussed in the text.

data, this amounts to a single ATLAS and CMS measurement in a particular final state for each LHC run. When multiple measurements, e.g., differential distributions, are reported, a single one is chosen based on maximising the sensitivity of our fit.

• The set of electroweak precision observables (EWPOs) include the pseudo-observables measured on the Z resonance by LEP and SLD, together with the W boson mass measurements by CDF and D0 at the Tevatron and ATLAS at the LHC:

EWPO: 
$$\{\Gamma_Z, \sigma_{\text{had}}^0, R_l^0, A_{FB}^l, A_l, R_b^0, R_c^0, A_{FB}^b, A_{FB}^c, A_b, A_c, M_W\}.$$
 (3.1)

We include a total of 14 electroweak measurements.

• For diboson measurements, we include the  $W^+W^-$  measurements of total crosssections at different energies and angular distributions at LEP, the fiducial differential cross-section in leading lepton  $p_T$  by ATLAS at the LHC, and ATLAS and CMS fiducial differential cross-section measurements of the Z-boson  $p_T$  in leptonic  $W^{\pm}Z$  production. We also incorporate the differential distribution in  $\Delta \varphi_{jj}$  for the Zjj measurement given by ATLAS, which we include in the diboson category because it is sensitive to related physics. We include a total of 118 diboson measurements.

- The Higgs dataset at the LHC includes the combination of Higgs signal strengths by ATLAS and CMS for Run 1, and for Run 2 both signal strengths and STXS measurements are used. ATLAS in particular provide the combined stage 1.0 STXS for  $4l, \gamma\gamma, WW^*, \tau^+\tau^-$  and  $b\bar{b}$ , while for CMS we use the signal strengths of  $4l, \gamma\gamma$ ,  $WW^*, \tau^+\tau^-$ ,  $b\bar{b}$  and  $\mu^+\mu^-$ . We also include the  $Z\gamma$  signal strength from ATLAS and a differential  $WW^*$  cross-section measurement from CMS. We include a total of 72 Higgs measurements.
- The top data consists of differential distributions in various  $t\bar{t}$  channels and crosssection measurements of top pair production in association with a W/Z boson or a photon (the  $t\bar{t}V$  dataset), as well as various single top differential and inclusive cross-section measurements, for both Runs 1 and 2. We include a total of 137 top measurements.

Overall, we include a total of 341 measurements in our analysis.

If not already given in such a form, each measurement is converted into a corresponding 'signal strength',  $\mu$ , defined as the ratio of the observed value to the best available theory prediction, usually quoted in the experimental publication. Differential data is taken from the publication and its associated entry in HEPdata, where available, using absolute differential cross section measurements,  $\vec{\sigma}_{abs}$ , and their associated covariance matrices,  $\Sigma_{abs}$ . If only normalised differential cross sections ( $\vec{\sigma}_{norm}$ ) are published, they are converted to absolute ones using the best available measurement of the inclusive cross section for that process in the same channel. Covariance matrices are then updated to reflect the correlations between the bins induced by the common rescaling of the total cross section,  $\sigma_{tot} \pm \delta \sigma_{tot}$ . The absolute differential measurement and its covariance matrix are then

$$\begin{split} \vec{\sigma}_{\rm abs} &= \vec{\sigma}_{\rm norm} \sigma_{\rm tot} \,, \\ \mathbf{\Sigma}_{\rm abs} &= \mathbf{\Sigma}_{\rm norm} \sigma_{\rm tot}^2 + \delta \sigma_{\rm tot}^2 \vec{\sigma}_{\rm norm} \otimes \vec{\sigma}_{\rm norm} \,. \end{split}$$

Where available, fastnlo tables [199–201] were used to obtain NNLO QCD predictions for the differential  $t\bar{t}$  data. The SM theoretical errors are taken to be uncorrelated and the relative signal strength covariance matrix is obtained by adding the relative experimental and theoretical covariances as follows:

$$\Sigma_{\mu} = \left(\frac{\Sigma_{\exp}}{\vec{\sigma}_{\exp} \otimes \vec{\sigma}_{\exp}} + \operatorname{diag}(\vec{\delta}_{\mathrm{th}}/\vec{\sigma}_{\mathrm{th}})^2\right) (\vec{\mu} \otimes \vec{\mu}), \text{ where } \vec{\mu} \equiv \frac{\vec{\sigma}_{\exp}}{\vec{\sigma}_{\mathrm{th}}}.$$
 (3.2)

This corresponds to adding the relative experimental and theory uncertainties in quadrature. The observables are stored in the Fitmaker database in json format, together with metadata and information about how each signal strength was obtained.

#### 4 SMEFT predictions

#### 4.1 General strategy

The SMEFT predictions for all of the included observables were computed using the code MadGraph5\_aMC@NLO together with the SMEFTsim [202] and/or the SMEFT@NLO [172] UFO models. These are used to extract the linear contribution,  $a_i^X$ , of a given Wilson coefficient,  $C_i$ , to a physical quantity, X, such as a production cross-section, partial or total decay width, or asymmetry:

$$\mu_X \equiv \frac{X}{X_{\rm SM}} = 1 + \sum_i a_i^X \frac{C_i}{\Lambda^2} + \mathcal{O}\left(\frac{1}{\Lambda^4}\right). \tag{4.1}$$

The  $a_i^X$  can usually be obtained with a single, high statistics Monte-Carlo (MC) run for each coefficient. In some cases, non-linear contributions from Wilson coefficients can arise in MC predictions due to the W-mass shift and modifications of total widths of intermediate particles. The former is a consequence of using the EW { $\alpha_{EW}, G_F, m_Z$ } input scheme, while the latter may modify branching ratios of narrow resonances such as the Higgs, W, Z or top. When such states are produced on shell, factorising a given process into production and decay via the narrow-width approximation (NWA) allows the decay contributions to be computed separately, then added to the prediction. If these effects cannot be factorised, as in the case of W-mass modifications or off-shell vector bosons in Higgs decays, the  $a_i^X$  are obtained by generating the predictions over a range of coefficient values and numerically fitting for the linear dependence. This is done, in particular, for Higgs production in association with a W-boson (WH), vector boson fusion (VBF),  $pp \to WW \to \ell \nu \ell \nu$  and  $pp \to W^{\pm}Z \to \ell^+ \ell^- \ell^{\pm} \nu$ .

Predictions are obtained at Leading Order (LO) in perturbation theory throughout, which, in almost all cases, corresponds to tree-level computations. Unless stated otherwise, we generate and analyse our events at the parton level and apply analysis-specific selection criteria to obtain the relevant fiducial regions of phase space. We use the following values in the aforementioned EW input scheme:

$$\alpha_{EW}^{-1} = 127.95, \qquad G_F = 1.16638 \times 10^{-5} \,\text{GeV}^{-2}, 
m_Z = 91.1876 \,\text{GeV}, \qquad m_H = 125.09 \,\text{GeV}, \qquad m_t = 173.2 \,\text{GeV}.$$
(4.2)

All other fermions are taken to be massless, which implies the use of five-flavour scheme PDFs, for which we use the default NNPDF23\_nlo\_as\_0119 sets [203] provided by MadGraph5\_aMC@NLO. The one exception is when the lighter fermions appear as Higgs boson decay products, when we assume that they interact with the Higgs via their Yukawa couplings,  $y_f \equiv 2^{\frac{3}{2}} m_f \sqrt{G_F}$ , with masses taken to be

$$m_{\mu} = 0.106 \,\text{GeV}\,, \quad m_{\tau} = 1.77 \,\text{GeV}\,, \quad m_c = 0.907 \,\text{GeV}\,, \quad \text{and} \quad m_b = 3.237 \,\text{GeV}\,.$$
 (4.3)

The latter two have been run up to the Higgs mass scale. In some cases, the 3rd generationspecific operators independently modify b-quark initiated contributions to EW processes such as VBF and diboson. We do not take these contributions into account as they are highly suppressed by the *b* PDFs. We do not assign a theory uncertainty to our predictions of the SMEFT contributions, assuming that they will be subdominant with respect to other uncertainties such as the baseline SM theory predictions. We also neglect other theoretical uncertainties inherent to the SMEFT framework itself, such as omitting quadratic dimension-6 or linear dimension-8 contributions and other higher-order effects (see, e.g., refs. [204–209] for discussions of these and related uncertainties). We note that operator mixing from RGE running and loops can also induce extra constraints [113, 118, 167, 210] and the effects of including SMEFT operators in parton distribution functions are also starting to be investigated [211].

# 4.2 Higgs production

We computed LO predictions in the full parameter space of our basis for the five main Higgs production modes: gluon-fusion (ggF), VBF, associated production with a W or Z(WH, ZH), and associated production with a top quark pair,  $t\bar{t}H$ . Some results are taken from ref. [160], after being cross checked by independent computations. Predictions in STXS bins are compared to and found to be in agreement with the predictions presented in refs. [164, 212]. The only one-loop calculations that we employ in our analysis involve the Higgs coupling to gluons, for which the LO contribution in the SM arises at one-loop level. Despite being a loop-induced coupling it mediates the  $gg \rightarrow H$  Higgs boson production mode, which is dominant at hadron colliders. Since many Higgs measurements are very sensitive to this production mechanism and the associated,  $qq \rightarrow H(+\text{jets})$  processes, we include as leading effects in the SMEFT both the tree-level contribution from  $C_{HG}$  and the leading effects from the operators that modify the top-loop contribution to the SM coupling:  $C_{tH}$ ,  $C_{tG}$ , and  $C_{H\square}$ . One final operator,  $C_G$ , modifies the gluon self-interaction allowing for a contribution to gluon fusion Higgs production in association with one or more jets, which we also include at one-loop order in this channel for the first time. The computations of linear contributions involve extracting the interference between loop diagrams of the SM with tree-level diagrams from  $C_{HG}$  as well as loop diagrams with a single operator insertion, and are made possible by SMEFT@NLO. We use a fixed renormalisation and factorisation scale of  $m_H$  in all such computations.

Figure 2 illustrates the predictions we obtain for a selection of stage 1.1 STXS gluon fusion bins [213], highlighting the potential additional discriminating power offered by the inclusion of H+jet(s). These predictions were obtained by parton-level generation of Higgs production in association with one or two additional jets. Specific predictions for, e.g., the 0-jet gluon-fusion bin with Higgs  $p_T > 10$  GeV or the '3-jet like'  $\geq 2$ -jet bins with  $p_T(Hjj) > 25$  GeV would require a matching/merging procedure interfaced with parton showering that goes beyond the level of sophistication of our analysis. Instead, we take the same dependence on the coefficients as we find for the associated parton-level bin ( $p_T^H = 0$ and  $p_T(Hjj) = 0$ ). This corresponds to assuming that the main effect on the population of the non-zero  $p_T$  bins will come from the parton shower, which does not depend on the EFT coefficients. Comparing our results with merged sample analysis of ref. [164] (tables 10–14), we find this assumption to be excellent for the two 0-jet bin, which have almost identical linear coefficients. Furthermore, the relevant 2-jet bins are compatible with the 'parton



Figure 2. Illustration of the effects of selected SMEFT operators on representative gluon-fusion simplified template cross-sections  $\sigma_i$  relative to the corresponding SM cross-sections  $\sigma_{\rm SM}$ , for  $C_i/\Lambda = 1 \,\text{TeV}^{-2}$ . The various Higgs  $p_T$  and  $m_{jj}$  bins provide complementary sensitivities and hence discriminating power between the operators.

shower only' assumption within about 10%. We note that the comparison of individual coefficients between the two analyses is not completely possible due to the different STXS binnings and EW input schemes used; additional information on the MC generation would be needed for a detailed cross-check. Nevertheless, we compare numbers for each operator where reasonable, and find that our  $C_{HG}$  contributions agree within 10–20%, while  $C_{tG}$  displays larger differences on the order of 20–60%. The other operators do not induce kinematics-dependent effects, contributing overall rescalings for which we also find good agreement with ref. [164].

If one were to use only Higgs signal strength data, they would be limited to the sensitivity shown in the left-most '0-jet' entry, and measurements of gluon fusion would exhibit a degeneracy in the four relevant coefficients. Instead, allowing for associated production with jets offers sensitivity to  $C_G$  and, most importantly, breaks the aforementioned degeneracy by exploiting the different energy dependences of the five operators. Besides  $\mathcal{O}_{H\Box}$  and  $\mathcal{O}_{tH}$ , there are 3 additional operators that globally rescale gluon fusion rates by shifting the top Yukawa interaction:  $C_{HD}$ ,  $C_{Hl}^{(3)}$  and  $C_{ll}$ . We do not show their impact because their contributions can simply be obtained from that of  $\mathcal{O}_{H\Box}$  by multiplicative factors of  $-\frac{1}{4}$ , -1 and  $\frac{1}{2}$ , respectively. Furthermore, they are severely constrained by EWPO, to the point where they are not expected to affect gluon fusion (or  $t\bar{t}H$ ).

Another relevant loop-induced process in Higgs production is  $gg \to ZH$ , which, while being formally an NNLO QCD contribution to ZH, accounts for a significant portion of the inclusive cross section [214], and has a harder  $p_T$  spectrum, which could be especially relevant for EFT interpretations [215]. Recent STXS definitions include dedicated bins for this contribution, although we are not aware of any explicit extraction of the cross sections. Notably, this process was shown to be sensitive to top quark interactions with the Z, which are relatively poorly measured elsewhere [216, 217]. However, we leave to future work the inclusion of this loop-induced sensitivity to EW top operators, also present in other processes such as  $gg \rightarrow VV$  [218], and EW Higgs production and decay [219].

#### 4.3 Gauge boson and Higgs decays

Many of the measurements included in our fit involve the on-shell production and decay of SM gauge bosons and/or the Higgs boson. As mentioned above, we employ the NWA to factorise production and decay, such that, at the linear level in the SMEFT, the modification to the cross section of a given process is a combination of the  $a_i^X$  for production, the partial width to the decay channel and the total width of the parent particle. The total width shifts of the W and Z bosons, have been determined as follows, where  $\Lambda$  has been fixed to 1 TeV:

$$\frac{\Gamma_Z}{\Gamma_Z^{\text{SM}}} = 1 - 0.05 C_{HWB} - 0.041 C_{HD} + 0.082 C_{ll} - 0.13 C_{Hl}^{(3)} - 0.012 C_{Hl}^{(1)} - 0.012 C_{He} + 0.077 C_{Hq}^{(3)} + 0.0078 C_{Hq}^{(1)} + 0.016 C_{Hu} - 0.012 C_{Hd} + 0.021 C_{HQ}^{(3)} + 0.021 C_{HQ}^{(1)}, \\
\frac{\Gamma_W}{\Gamma_W^{\text{SM}}} = 1 - 0.14 C_{HWB} - 0.065 C_{HD} + 0.10 C_{ll} - 0.16 C_{Hl}^{(3)} + 0.081 C_{Hq}^{(3)}.$$
(4.4)

Higgs boson decays present a richer structure, due to the importance of four-fermion decay modes that are mediated in the SM by the H couplings to the W and Z. Although the Higgs mass is too small for both gauge bosons to be on-shell, it is often assumed that at least one of the gauge bosons is, simplifying the decay to a three-body,  $H \rightarrow V f' \bar{f}$ , final state with the NWA applied to on-shell vector boson, V = W, Z. This neglects certain interference effects between, e.g., neutral-current and charged-current mediated four-fermion decays. The SMEFT introduces tree-level  $H\gamma\gamma$ ,  $H\gamma Z$  and Hgg interactions, all of which can contribute to four-fermion decay modes, and degrade the accuracy of the NWA. In practice, however, experiments often make invariant mass cuts around  $m_Z$  in, e.g.,  $h \rightarrow 4\ell$  analyses, that could largely mitigate this effect.

Ref. [173] performs an in-depth calculation and analysis of Higgs decays to four fermions beyond the NWA, from which we take our predictions. We also include the contributions to  $h \to gg$  mediated by operators that modify the top-quark loop contribution. These have the same relative impact as they do on  $gg \to H$ , discussed in the previous section.<sup>8</sup> The results of ref. [173] are given for the flavour-universal SU(3)<sup>5</sup> scenario, which is more restrictive than our top-specific one, singling out operators affecting the left-handed *b*-quark couplings to the Z,  $C_{HQ}^{(1)}$  and  $C_{HQ}^{(3)}$ . In order to account for four-fermion decays involving *b* quarks, it was necessary to adapt these results, complemented with some predictions from

<sup>&</sup>lt;sup>8</sup>However, for reasons of consistency, we do not include the contributions to  $h \rightarrow \gamma \gamma$  and  $Z \gamma$  mediated by operators that modify the top-quark loop contribution, which are formally of the same order as other NLO electroweak corrections that we do not include in general, and would be similarly relevant for, e.g., Z-pole data.

SMEFT@NLO. The contributions to the total Higgs width were found to be

$$\frac{\Gamma_{H}}{\Gamma_{H}^{SM}} = 1 + 0.18 C_{HWB} + 0.018 C_{HD} + 0.045 C_{ll} + 0.11 C_{H\Box} + 1.7 C_{HG} - 0.075 C_{HW} 
- 0.093 C_{HB} - 0.089 C_{Hl}^{(3)} - 0.000059 C_{Hl}^{(1)} - 0.000039 C_{He} - 0.000051 C_{Hd} 
+ 0.0037 C_{Hq}^{(3)} - 0.00025 C_{Hq}^{(1)} + 0.000055 C_{Hu} - 0.00014 C_{HQ}^{(3)} + 0.00038 C_{HQ}^{(1)} 
- 0.73 C_{\tau H} - 0.0057 C_{tH} - 4.0 C_{bH} - 0.043 C_{\mu H} + 0.044 C_{tG}.$$
(4.6)

In any case, the only experimentally accessible four-fermion decay modes are the leptonic ones, so quark current operators are only practically relevant via their effect on the total width, which is clearly very small. For comparison, we computed the Higgs decays using the NWA for  $h \to VV^* \to 4f$  processes, noting that the two approaches can have significantly different predictions for some operators. However, we do not find any appreciable differences in the results of the global fit, indicating that Higgs decays are not the primary source of constraints for the operators whose predictions are sensitive to whether or not this approximation is used. Finally, we note that a recent EFT interpretation of Higgs production measurements in the  $h \to 4\ell$  channel [220] pointed out significant acceptance differences in the kinematical selection between the SM and the EFT due to the additional off-shell photon contributions from  $C_{HW}$ ,  $C_{HB}$  and  $C_{HWB}$ . We do not take any acceptance corrections into account in our analysis, but also do not expect these to have a large impact on the global fit, given the fact that our results are unaffected by whether or not we use the NWA.

#### 4.4 Diboson data

We have obtained predictions for the fiducial signal strengths in bins of leading-lepton  $p_T$  for the ATLAS WW [70] analysis, the Z-boson  $p_T$  for the CMS and ATLAS WZ analyses [67, 69], and in bins of  $\Delta \varphi_{jj}$  for the ATLAS Zjj analysis [71], following the general strategy outlined above. For the WW and WZ analyses, we find good agreement of the total SM fiducial cross-section, whereas in the Zjj cases we have validated our analysis by comparison with the binned signal strengths for the operator coefficients reported by ATLAS and CMS. In particular, as pointed out in ref. [71], we find that this channel is the most sensitive to the interference term that is linear in the triple gauge boson operator  $\mathcal{O}_W$ . For LEP we used the WW results of ref. [107] that are provided for the total and differential cross-sections at different centre of mass energies. We note that this analysis uses a restricted set of angular distribution bins to mitigate the effects of unknown correlations in those bins.

#### 4.5 Top data

Since we only make use of parton-level unfolded data, all top predictions are generated with stable top quarks. The data assume SM-like decay chains for the top quarks, and we do not take into account the small modifications to the W boson branching fractions in top decays due to current operators. For top production in association with a gauge or Higgs boson, we do take into account the modified branching fraction of the associated boson in the measured decay channel (usually leptonic for W and Z). However, this choice is not expected to have a significant impact on the results, given that top data is mainly sensitive to top-related operators, with those that modify gauge or Higgs boson decays being better constrained elsewhere. Much of the data we use overlaps with the data used in the top sector fit of ref. [168], in which predictions are also obtained with SMEFT@NLO, but including parton shower effects. We make use of the linear, LO parts of these in our work<sup>9</sup> and generate new, parton-level samples for these measurements, as including parton showering was not found to be very significant in determining the  $a_i^X$ . Our work also includes some top quark asymmetry measurements, namely those of the forward-backward and charge asymmetries from the Tevatron and LHC, respectively. Splitting a measured cross section,  $\sigma$ , into two regions,  $\sigma_F$  and  $\sigma_B$ , the asymmetry, A, is defined as  $A \equiv (\sigma_F - \sigma_B)/(\sigma_F + \sigma_B)$ . The linearised contribution of a given operator is proportional to the difference between its relative contributions to the cross sections in the two regions,  $a_i^{\sigma_F}$  and  $a_i^{\sigma_B}$ ,

$$a_{A}^{i} = \frac{1 - A_{\rm SM}^{2}}{2A_{\rm SM}} (a_{i}^{\sigma_{F}} - a_{i}^{\sigma_{B}}), \qquad (4.7)$$

highlighting how asymmetry measurements may be useful in breaking parameter space degeneracies in total cross section measurements that would, instead, be sensitive to the sum,  $a_i^{\sigma_F} + a_i^{\sigma_B}$ . We use the latest NNLO QCD + NLO EW theory predictions for the top asymmetries from ref. [221]. The impact of the asymmetries is quantified in section 6.2.

# 5 Fitting procedure

We perform a  $\chi^2$  fit for a vector of observables,  $\vec{y}$ , with covariance matrix,<sup>10</sup> V, and theory predictions for those observables,  $\vec{\mu}(C_i)$ , using a  $\chi^2$  function defined as

$$\chi^2(C_i) = (\vec{y} - \vec{\mu}(C_i))^T \mathbf{V}^{-1} (\vec{y} - \vec{\mu}(C_i)) .$$
(5.1)

The predictions are functions of the dimension-6 operator coefficients  $C_i$ , as defined in eq. (2.7), and are truncated at the linear level so as to include only the interference term with the SM. The quadratic dependence on  $C_i$  is generically of the same order as linear interference terms with coefficients of dimension-8 operators, though exceptions exist in some specific UV completions [222]. For example, in a UV completion with a single particle and a single coupling the quadratic dimension-6 contributions can be larger than the linear dimension-6 ones in the strong-coupling regime, though not more generally. The importance of their effect is therefore a model-dependent question. We note also that sensitivity of a linear fit to quadratic contributions is an indicator for a possible breakdown in the regime of validity of the SMEFT, so care must be taken in the interpretation of the fit.<sup>11</sup>

 $<sup>^{9}</sup>$ We thank the authors of ref. [168] for sharing the predictions with us.

<sup>&</sup>lt;sup>10</sup>In the case of a non-symmetric covariance matrix  $\tilde{\mathbf{V}}$ , we symmetrise it by defining  $\mathbf{V}^{-1} = \frac{1}{2} \left( \tilde{\mathbf{V}}^{-1} + (\tilde{\mathbf{V}}^{-1})^T \right)$ . We have verified in a numerical fit to Higgs data that this does not affect significantly the results.

<sup>&</sup>lt;sup>11</sup>This issue is illustrated in the context of Higgs measurements in appendix B.

The least-squares estimators  $\hat{C}_i$  that extremise the  $\chi^2$  function can be obtained analytically in the case of a linear fit (see, e.g., ref. [223] for a review). We may write the linear theory prediction in terms of a matrix **H** that characterises the modification of the SM predictions  $\vec{\mu}^{\text{SM}}$  at linear order:

$$\mu_{\alpha}(C_i) = \mu_{\alpha}^{\rm SM} + \mathbf{H}_{\alpha i} C_i \,. \tag{5.2}$$

A summation over repeated indices is implied; the index  $\alpha$  ranges over the number of observables and *i* ranges over the number of dimension-6 coefficients. Solving  $\partial \chi^2 / \partial C_i = 0$  gives the best fit values as

$$\hat{\vec{C}} = \left(\mathbf{H}^T \mathbf{V}^{-1} \mathbf{H}\right)^{-1} \mathbf{H}^T \mathbf{V}^{-1} (\vec{y} - \vec{\mu}^{\mathrm{SM}}) \equiv \mathbf{F}^{-1} \vec{\omega} .$$
(5.3)

It is convenient to define the symmetric Hessian matrix **F**, also known as the Fisher information matrix, and the  $\chi^2$  gradient vector  $\vec{\omega}$  as

$$\mathbf{F} \equiv \mathbf{H}^T \mathbf{V}^{-1} \mathbf{H}, \qquad \vec{\omega} \equiv \mathbf{H}^T \mathbf{V}^{-1} (\vec{y} - \vec{\mu}^{\text{SM}}), \qquad (5.4)$$

in terms of which the  $\chi^2$  function eq. (5.1) can be written as

$$\chi^{2}(C_{i}) = \chi^{2}_{\rm SM} - 2\vec{C}^{T}\vec{\omega} + \vec{C}^{T}\mathbf{F}\vec{C} = \chi^{2}_{\rm min} + (\vec{C} - \hat{\vec{C}})^{T}\mathbf{F}(\vec{C} - \hat{\vec{C}}), \qquad (5.5)$$

where  $\mathbf{F} \equiv \mathbf{U}^{-1}$  is the inverse of the covariance matrix of the least-squares estimators,  $\vec{C}$ .

Splitting the coefficients into  $\vec{C} = \{\vec{C}_A, \vec{C}_B\}$ , we may profile over a subset of coefficients  $\vec{C}_A$  to obtain the least-squares estimators  $\hat{\vec{C}}_B$  for the remaining coefficients  $\vec{C}_B$ . For this purpose, the Fisher information matrix may be decomposed into the sub-matrices

$$\mathbf{F} = \begin{pmatrix} \mathbf{F}_A & \mathbf{F}_{AB} \\ \mathbf{F}_{AB}^T & \mathbf{F}_B \end{pmatrix}, \qquad (5.6)$$

and the gradient vectors as  $\vec{\omega} = \{\vec{\omega}_A, \vec{\omega}_B\}$ . The profiled best fit values are then given by

$$\hat{\vec{C}}_B = \left(\mathbf{F}_B - \mathbf{F}_{AB}^T \mathbf{F}_A^{-1} \mathbf{F}_{AB}\right)^{-1} \left(\vec{\omega}_B - \mathbf{F}_{AB}^T \mathbf{F}_A^{-1} \vec{\omega}_A\right) \,. \tag{5.7}$$

In cases where a prior on the coefficients needs to be imposed, for example when the magnitude-squared of couplings cannot go negative, as when matching to specific UV models in section 7, or when including quadratic dependences on the coefficients, this analytic method may no longer be used. A numerical MCMC method using MultiNest has therefore also been implemented in Fitmaker, as described in appendix B.

# 6 Global results

#### 6.1 Higgs, diboson and electroweak fit

The main emphasis of this section is on the improvements in the Higgs data since the Run 1 and early Run 2 data that were analysed in the SMEFT framework in [160]. However, as has been emphasised previously, e.g., in [97, 105], there is considerable overlap between the

sets of operators whose coefficients are constrained by both electroweak and Higgs data,<sup>12</sup> as visualised in figure 1. Therefore we present in this section results from a joint fit to the combined Higgs, diboson and electroweak data, including the 20 operators of relevance listed in eq. (2.8). We recall that the latter are dominated by data from LEP, with the most important LHC contribution coming from an ATLAS measurement of  $M_W$ .

As already mentioned, this joint fit is carried out to linear order in the dimension-6 SMEFT operator coefficients, neglecting quadratic dimension-6 contributions to the LHC measurements and linear dimension-8 contributions, which, as discussed above, are *a priori* of similar order in the scale  $\Lambda$  of high-mass BSM physics. Details of the fit procedure are described for the analytic method in section 5 and a numerical MCMC method in appendix B. We use mainly the former, but have verified in representative cases that fit results do not depend significantly on the method used. Appendix B also discusses the importance of effects that are quadratic in the dimension-6 operators, and we refer the interested reader to [208] for a discussion of possible dimension-8 effects in Higgs measurements. We emphasise again the importance of using, as well as total Higgs production and decay rates, kinematic measurements of Higgs production as encapsulated in STXS measurements, due to the different  $p_T$  dependences of dimension-6 contributions to production amplitudes, whose relative importances are generally enhanced at higher  $p_T$  [160], as seen in figure 2.

The top panel in figure 3 shows the 95% CL intervals for the coefficients  $C_i(1 \text{ TeV})^2/\Lambda^2$ of the 20 dimension-6 SMEFT operators contributing to the joint Higgs, diboson and precision electroweak data analysis when each operator is analysed individually. We note that certain coefficients have been scaled as indicated by the labels on the x-axis. In particular, the Yukawa operator  $C_{tH}$ , the triple-gauge boson operators  $C_W$  and  $C_G$ , and the Higgs-only operator  $C_{H\Box}$  have all been rescaled by 2 orders of magnitude to appear on the same scale. Dotted grey vertical lines separate the sets of operators that contribute mostly to electroweak precision observables (EWPO), that contribute mostly to Higgs and diboson measurements (Bosonic), and that modify the Yukawa couplings (Yukawa). These categories are indicated as guides to aid the reader; however, as discussed in more detail later, there are correlations between these sectors.

The two right-most bars for each operator are from fits using all the available Higgs and diboson data from Run 2 of the LHC together with the precision electroweak constraints, differing only in whether they include the STXS constraints. The intermediate blue bars demonstrate the effect of replacing the combined ATLAS STXS dataset with the latest combined ATLAS signal strengths, in which case there are slightly weaker constraints for some of the operator coefficients. These results do not differ significantly whether predictions in the on-shell approximation are used, where the  $H \to 4f$  process is assumed to originate from an underlying  $H \to VV^* \to 4f$  decay with one gauge boson taken to be on-shell, or whether the SMEFT dependence of the full  $H \to 4f$  matrix elements is used (see [173] and section 5). For comparison, the left-most bar (light blue) is from a fit in which only the Higgs and diboson data analysed in [160] are used. In the case of  $C_W$ , we

 $<sup>^{12}\</sup>mathrm{We}$  show later in figure 11 the effect on the marginalised fit of removing the LEP EWPO and WW datasets.



JHEP04 (2021)279

Figure 3. Constraints on the indicated individual and marginalised operator coefficients  $C_i(1 \text{ TeV})^2/\Lambda^2$  (top and third panels) and the corresponding scales  $\Lambda$  for the indicated values of the  $C_i$  at the 95% confidence level (second and bottom panels), from a combined linear fit to the Higgs, diboson and electroweak precision observables. In the top two panels, the bars show the 95% CL ranges from the LHC Run 1 and early Run 2 data (light blue), current data without using the STXS measurements (intermediate blue), and current data including STXS using either the on-shell vector boson approximation or the full  $1 \rightarrow 4$  matrix elements for the 4-fermion Higgs decay modes taken from ref. [173] (dark blue). In the bottom two panels, the corresponding marginalised results are indicated by yellow, orange and red bars, respectively. We also show in purple in the individual case (grey in the marginalised case) the effect of dropping the ATLAS Z+ jets measurement.

also show (in purple) the effect of dropping the ATLAS  $Z_{jj}$  measurement [71], which does not impact significantly the constraints on the other operator coefficients.

In general, the individual fit ranges for the first 10 operator coefficients of figure 3 counting from the left, i.e., up to and including  $C_{Hu}$ , are very similar in fits including all the LHC Run 2 data to those found using the earlier set of Higgs, diboson and electroweak data. This is because the precision electroweak data provide the strongest constraints on these individual operator coefficients. The impacts of the Higgs and diboson data are more apparent for the rest of the operator coefficients, i.e., from  $C_{H\Box}$  rightwards, particularly for the Yukawa operators that have benefited from improved sensitivity of the Higgs couplings to the tau and bottom. The relative constraining power of datasets, as measured by the Fisher Information, is given in table 7 in the appendix, and confirms the points discussed above. It also quantifies the importance of the Zjj data (84%) in pinning down  $C_W$ , compared to  $W^{\pm}Z$  (13%) and LEP 2  $W^+W^-$  (3%)

Using the same colours, the bars in the second panel show the 95% CL lower limits on the  $\Lambda_i$  on logarithmic scales in units of TeV, for different values of  $C_i$ . These reaches are estimated by taking half the width of the 95% CL ranges of  $C_i/\Lambda_i^2$  as the typical sensitivity of the measurement. Here and in subsequent analogous panels, the darker (lighter) coloured histograms are for  $C_i = 0.01(1)$  and the histograms with dashed outlines are for the strongcoupling perturbativity limit  $C_i = (4\pi)^2$ . In general, the  $\Lambda_i$  scales would be modified by a factor  $\sqrt{C_i}$ , which would depend on whether the Wilson coefficient is induced by stronglyor weakly-coupled new physics, at tree- or loop-level.

The corresponding 95% CL constraints for the marginalised case, where we include simultaneously all operators in the analysis and then profile the likelihood over all coefficients except one, as described in section 5, are shown in the lower pair of panels in figure 3. The yellow, orange and red bars from left to right are the fits to the old data, the new data without the STXS measurements, and including them, respectively. There is again no significant difference for different treatments of  $h \to 4f$ . The marginalised constraints are weaker overall than the individual constraints, as the fit is allowed to explore all possible variations in the space of coefficients. We also note that the STXS measurements play a key role in the marginalised constraints for some operators, e.g.,  $C_{HG}, C_G$ and  $C_{tH}$ . As discussed in section 4.2, removing them causes a degeneracy in the parameter space that prevents meaningful constraints in these directions. We show in dark brown the impact on  $C_W$  of dropping the Z<sub>jj</sub> constraint, which still does not significantly affect the other operators. The more traditional diboson measurements suffer from suppressed SM interference at high energy due to helicity selection [224]. This is particularly so for  $W^+W^-$ , while  $W^{\pm}Z$  appears to retain some sensitivity. This is why the bound without Zjj changes significantly, becoming dominated by the  $W^{\pm}Z$  and/or LEP 2 data, with the high mass  $W^+W^-$  distributions from the LHC yielding no significant improvement. The  $Z_{ij}$  observable is therefore extremely useful for constraining anomalous gauge boson selfinteractions at linear level, overcoming the non-interference issue and accessing the leading contributions in the SMEFT expansion.

We see that most of the 95% CL ranges are reduced when the full Run 2 data are included, some quite substantially, the only exceptions being  $C_G$  and  $C_{HG}$ . This occurs



Figure 4. Left panel: breakdown of the impacts of various top datasets on the 2-dimensional subspace of the four-fermion operators  $(C_{Qq}^{1,8}, C_{tq}^8)$ , setting all other operator coefficients to zero. Right panel: similar breakdown of the impacts of various top datasets on the 2-dimensional subspace of the four-fermion operators  $(C_{Qu}^8, C_{Qd}^8)$ , setting all other operator coefficients to zero.

despite the individual constraints improving in both parameter space directions. We attribute the slight worsening of the marginalised bounds to the presence of a highly boosted  $H \rightarrow b\bar{b}$  measurement in the 2018 data [225] that selects Higgs  $p_T > 450$  GeV, which is significantly higher than the highest,  $p_T > 200$  GeV of the stage 1.0 STXS bins used in our 2020 dataset. Removing this observable degrades the 2018 bounds below our most recent results. We expect this sensitivity to be recovered once the stage 1.2 measurements, which probe a similarly high  $p_T$  region, are incorporated. We find  $\chi^2/dof = 0.94$  (p = 0.72) for our flavour-universal global fit, to be compared with  $\chi^2/dof = 0.93$  (p = 0.76) for the SM. Among the 20 operators considered in the fit to the Higgs, diboson and electroweak data, the most weakly constrained operator coefficients are all constrained so that  $\Lambda/\sqrt{C_i} \gtrsim 500$  (400) GeV in the individual (marginalised) fits, suggesting that the linear SMEFT treatment may be adequate in this sector [226].

# 6.2 Top fit

We consider now the information provided by top quark data, in a dedicated fit to a subset of relevant 'top quark operators', i.e., those involving at least one top quark field (the top Yukawa operator, and the top 2F and top 4F categories in equation (2.12)) plus the triple gluon operator,  $C_G$ , which globally affects most QCD-induced processes. We also include EW precision observables, which close a single blind direction  $(C_{HQ}^{(1)} + C_{HQ}^{(3)})$  affecting the left-handed  $Zb\bar{b}$  coupling, as well as the latest  $t\bar{t}H$  signal strength measurements from the two CMS and ATLAS Higgs combination papers included in the final fit, to constrain the top Yukawa operator. Overall, the  $\chi^2/n_{dof}$  for the SM is 0.92. However, there are a number of observables that exhibit some tension with the SM predictions, such as the recent 13 TeV measurement of the differential  $p_T$  distribution in t-channel single top production [56]  $(\chi^2/n_{d.o.f} = 5.3), t\bar{t}W$  cross section measurements by CMS at 8 TeV [26] and ATLAS at 8



Figure 5. Constraints on the indicated individual and marginalised operator coefficients at the 95% confidence level (upper and lower figures, respectively), from a combined linear fit to the top data and electroweak precision observables. The impact of  $t\bar{t}$  data is highlighted by the evolution of the constraints starting from no  $t\bar{t}$  data (light blue/yellow) adding Run 1  $t\bar{t}$  total and differential cross-section data (blue/pink), the corresponding Run 2  $t\bar{t}$  data (purple/orange), and finally  $t\bar{t}$  asymmetry measurements  $A_{FB}$  from the Tevatron and  $A_C$  from the LHC (green/red).

and 13 TeV [23, 58] ( $\chi^2/n_{\rm d.o.f} = 2.6, 2.1$  and 1.9), and CMS  $t\bar{t}$  differential distributions at 8 TeV ( $\frac{d\sigma}{dm_{t\bar{t}}y_{t\bar{t}}}$  in the l+jets channel) and 13 TeV ( $\frac{d\sigma}{dm_{t\bar{t}}}$  in the dilepton channel), both with  $\chi^2/n_{\rm d.o.f} = 1.6$ . These tensions may lead to a preference for non-zero Wilson coefficients, though this depends on whether other, more consistent observables also constrain the operators in question.

Figure 5 shows the 95% CL intervals for the coefficients individually (top two panels) and after marginalisation (bottom two panels). The pale blue bars in the top two panels are obtained without using any  $t\bar{t}$  data, the intermediate blue bars are obtained including the Run 1 tt total and differential cross-section data, the dark blue bars include also the corresponding Run 2  $t\bar{t}$  data, and finally the green bars include also the  $t\bar{t}$  asymmetry measurements  $A_{FB}$  from the Tevatron and  $A_C$  from the LHC. When removing the  $t\bar{t}$ data entirely, a closed fit is not possible, so marginalised constraints do not exist. The corresponding colours for the latter three sets of data (Run 1  $t\bar{t}$ , including Run 2, and including  $A_{FB}$  asymmetries) in the bottom two panels are yellow, orange and red. We also show in dark brown the impact of removing electroweak precision observables for the two operators most affected,  $C_{HQ}^{(1)}$  and  $C_{Ht}$ . As previously, the constraints on the scales  $\Lambda/\sqrt{C_i}$  are estimated by taking half the width of the 95% CL ranges of  $C_i/\Lambda_i^2$  for C=1as the typical sensitivity of the measurement. We note that many of the constraints on the quantities  $C_i(1\text{TeV})^2/\Lambda^2$  in the individual and marginalised cases differ by an order of magnitude, as shown in the top and third panels of figure 5, and that the corresponding constraints on the scales  $\Lambda/\sqrt{C_i}$  are typically a factor ~ 3 stronger in the individual analysis, as seen in the second and bottom panels.

The impact of the tension with the SM for the aforementioned observables can be seen from the individual constraints of figure 5. We see that  $C_{tH}$  and the 'top 2F' category appear consistent with the SM. Although it disagrees the most with the SM, the single top differential  $p_T$  data does not lead to significant pulls for the operators that can affect it,  $C_{Qq}^{3,1}$ ,  $C_{HQ}^3$  and  $C_{tW}$ . This is due to the fact that single operators are not able to improve the fit significantly, given that the disagreement predominantly comes from the lowest  $p_T$  bin, combined with the relatively good agreement of most other single-top data. Instead, the effect of the  $t\bar{t}W$  data can be seen in the individual constraints without  $t\bar{t}$ data, that cause the deviation of the best-fit values for  $C_{Qq}^{1,8}$  and  $C_{Qq}^{3,8}$  away from zero in the positive and negative directions, respectively. This is consistent with a relative minus sign in the linear dependence of the  $t\bar{t}W$  process on these two operators. On the other hand,  $t\bar{t}$  data depend on 9 of the operators in question, namely all of the 'top 4F' category except  $C_{Qq}^{3,1}$ , as well as  $C_{tG}$  and  $C_G$ . Gradually adding the differential  $t\bar{t}$ data draws the coefficients towards negative values, resulting in particularly large pulls, especially for  $C_G$ . Finally, the  $t\bar{t}$  asymmetry observables add an orthogonal constraint on the four-fermion operators that restore consistency with the SM in this sector. Since  $C_G$ does not produce an angular asymmetry in the  $t\bar{t}$  matrix element, its significant, non-zero best-fit value remains. Considering now the marginalised results, we see that more  $t\bar{t}$  data and the inclusion of asymmetries has a significant impact on the global sensitivity, even indirectly affecting sensitivity to EW top quark couplings  $C_{Ht}$  and  $C_{tB}$  by constraining the allowed four-fermion contributions to  $t\bar{t}Z/\gamma$ . We also show the importance of EW precision observables in closing the parameter space for neutral top quark couplings, by noting the large significant degradation of the limits in this space when removing them from the fit. All of the potentially large pulls are washed out by the marginalisation, except for  $C_G$  and  $C_{tG}$ , which lie 4.1 and 3.2 standard deviations away from zero, respectively.

The interplay between different  $t\bar{t}$  measurements in constraining the four fermion sector is shown in figure 4, inspired by a similar analysis in ref. [166]. Two pairs of operators are selected, setting all other coefficients to zero:  $C_{Qq}^{1,8}$  and  $C_{tq}^{8}$ , which couple to left- and right-handed top quarks, respectively, in the left panel, and  $C_{Qu}^{1,8}$  and  $C_{Qd}^{8}$ , which couple left handed top quarks to the up and down quarks, respectively, in the right panel. We see explicitly the complementarity between the  $t\bar{t}$  cross-section measurements, which strongly constrain one linear combination of  $C_{Qq}^{1,8}$  and  $C_{tq}^{8}$ , and the asymmetry measurements, which constrain an orthogonal direction. The combination of these measurements constrains each of  $C_{Qq}^{1,8}$  and  $C_{tq}^{8}$  quite tightly, though less so in the marginalised case, indicated by the dashed lines. The fact that forward-backward asymmetries are sensitive to the chiral structure of the  $t\bar{t}$  matrix element explains why they excel at distinguishing modified interactions for left- and right-handed tops but less so the isospin of the initial state quark (u or d). In this case the combination of  $t\bar{t}$  measurements constrains a highly correlated combination of the coefficients, though each is only weakly constrained in the marginalised fit, as seen in the bottom two panels of figure 5. The large differences between their individual and marginalised limits that we observe indicate strong correlations among the top quark four fermion operators constraints and that the overall marginalised sensitivity is set by the less precise  $t\bar{t}X$  data.

Overall, the best-constrained coefficients in figure 5 are  $C_{tG}$ ,  $C_{HQ}^{(3)}$ ,  $C_{HQ}^{(1)}$ ,  $C_{Qq}^{3,1}$  and  $C_G$ , with one direction being driven by electroweak precision observables. The large negative values of  $C_G$  can be traced back to  $t\bar{t}$  differential cross-section measurements, and we discuss a specific example in section 6.3.3 below, where we also consider the possibility that  $C_G$  is very small, as suggested on the basis of a quadratic analysis of multijet data. Four-fermion operator coefficients are less well constrained, with scales  $\Lambda$  between 800 - 1500 GeV in the individual analysis and 300 - 500 GeV in the marginalised analysis when the corresponding  $C_i = 1$ , in which case the validity of the global SMEFT interpretation for these operators could be questioned for weakly-coupled UV completions, given that some of the  $t\bar{t}$  data extends up to TeV energies. We therefore expect the differences between our top data analysis and those performed at quadratic level to be especially significant for the top quark four-fermion operators, as shown in ref. [168].

The neutral top quark operator coefficients  $C_{Ht}, C_{tB}$  are also particularly hard to constrain. Production of  $t\bar{t}Z/\gamma$  and tZ are the main handles we have on these couplings, and these are still not so well measured and only beginning to produce differential data. The  $t\bar{t}\gamma$  differential distributions in photon  $p_T$  turn out to provide the best handle on  $C_{tB}$ . Unfortunately,  $C_{Ht}$  does not predict any effects that grow in energy in either of these processes, and  $C_{tB}$  has a suppressed interference with the SM, meaning that one does not expect spectacular gains from differential measurements, especially in a linear analysis [216, 227]. Other rare EW top processes, such as  $t\bar{t}Wj$  and tWZ have been shown to be sensitive to such unitarity-violating behaviour and will therefore provide useful constraining power, once they are measured at the LHC [228, 229]. However, in all cases we see that the dashed histograms extend beyond a TeV even in the marginalised case, indicating that the SMEFT analysis should be a good approximation in the strong-coupling limit  $C_i = (4\pi)^2$ .

# 6.3 Combined top, Higgs, diboson and electroweak fit

As we discussed in section 2, there are different possible treatments of the flavour degrees of freedom within the SMEFT framework. Specifically, in this paper we assume either an  $SU(3)^5$  symmetry in the operator coefficients (broken by the Yukawa operators) or allow this symmetry to be broken to an  $SU(2)^2 \times SU(3)^3$  symmetry by the coefficients of third-generation fermions in a top-specific flavour scenario.

We find that, among the 20 flavour-universal operators in the SU(3)<sup>5</sup> scenario, only the marginalised constraints on  $C_G$ ,  $C_{HG}$  and  $C_{tH}$  are improved significantly by the additional top measurements beyond the constraints that are already provided by the electroweak, diboson and Higgs data, while the sensitivity to  $C_{Hq}^{(1)}$  and  $C_{Hq}^{(3)}$  decreases, since their thirdgeneration flavour components have been separated into separately constrained degrees of freedom. This indicates the robustness of the fit despite the increase in the number of parameters. However, there are significant correlations between the datasets. Accordingly, in the following we focus attention on the results from a simultaneous global fit to the 34 operators of the top-specific flavour scenario that was described in section 2.2.2.

#### 6.3.1 Operator sensitivities in individual and marginalised fits

Figures 6 shows the results from this combined fit to all the available Higgs, electroweak, diboson and top data, switching on one operator at a time (top two panels) and marginalising over all other coefficients (bottom two panels), respectively. In each case, the upper panel shows the 95% CL ranges for the operator coefficients  $C_i$  normalising the corresponding new physics scales to 1 TeV. As indicated on the x-axis labels, certain operator coefficients have been rescaled for the sake of convenience. The bars in the lower panels show the 95% CL lower limits on the  $\Lambda_i$  on logarithmic scales in units of TeV. As previously, these reaches are estimated by taking half the width of the 95% CL ranges of  $C_i/\Lambda_i^2$  for C = 1as the typical sensitivity of the measurement.

The differences in the constraints on the 20 operators entering the flavour-independent  $SU(3)^5$  fit between including top data or not are small in the individual case (dark vs light blue), except for  $C_G$ . For a more detailed breakdown of the relative constraining power of different datasets on each individual coefficient, we refer the reader to table 7. Marginalising widens the ranges allowed by the fit, but the effect of marginalising over a larger set of coefficients — 34 compared to 20 in the top-specific flavour scenario compared to the flavour-universal case (red vs yellow), introduces noticeable differences for only a few operator coefficients, namely  $C_G, C_{HG}$  and  $C_{tH}$ . The differences in the constraints on the top operators (shown in green in the individual case and pink in the marginalised case) when the electroweak, diboson and Higgs data are included in the fit are generally small, apart in the case of the Yukawa operator,  $C_{tH}$  and the top chromomagnetic dipole operator,  $C_{tG}$ . The loop-level constraints from Higgs production via gluon fusion are clearly



Figure 6. Results from the global fit to the electroweak, diboson, Higgs and top data in the top-specific  $SU(2)^2 \times SU(3)^3$  scenario. Top two panels: fits to the individual operators, showing the 95% CL ranges for the operator coefficients  $C_i$  normalising the new physics scale  $\Lambda$  to 1 TeV, and the ranges for the scales  $\Lambda$  for different values of the  $C_i$ . Bottom two panels: similar, but marginalising over the other operators. In the top panels, fit results found using only electroweak, diboson and Higgs data are shown in light blue, those found in the combination with the top data are shown in dark blue, and those using only top data are shown in green. In the bottom panels, the corresponding marginalised fit results are shown in yellow, red and pink.

very powerful in constraining these operator individually, but this sensitivity is diluted by marginalisation, which allows the other operators affecting this process to float.

Overall, the data are sufficient for a closed fit with no flat directions, and we find  $\chi^2/dof = 0.81$  (p = 0.99) for our top-specific global fit. The sensitivities to the scale of new physics in the operator coefficients in the individual case are generally several hundred GeV or more for  $C_i = 1$ . We also note that the  $C_i$  for the Yukawa operators would be expected to contain a Yukawa factor in the MFV hypothesis, as discussed in section 2, with the bounds then weakened appropriately. The scale sensitivities in the marginalised case still reach a TeV for  $C_i = 1$  for most of the electroweak precision observables set and some of the operators. We emphasise, however, that specific UV completions each generate only a subset of operators, for which they have correspondingly improved reaches. The individual and marginalised fits can therefore be taken as optimistic and pessimistic sensitivity estimates, respectively, with realistic cases living somewhere between the two.

#### 6.3.2 Sensitivities in 'Higgs-only' operator planes

In order to assess the potential impact of the interplay between top and Higgs data, we may consider the following subset of 'Higgs-only' operators:

$$\{C_{H\Box}, C_{HG}, C_{HW}, C_{HB}, C_{tH}, C_{bH}, C_{\tau H}, C_{\mu H}\}$$
(6.1)

together with  $C_G$  and  $C_{tG}$ , which do not modify Higgs interactions directly but can impact gluon fusion. Performing a fit to this subset, figure 7 displays the result for the 95% CL constraints when top data are combined with Higgs data in planes showing different pairs of the operator coefficients  $C_{HG}, C_{tG}, C_{tH}$  and  $C_G$ , marginalised over the other coefficients in (6.1). This is the relevant set of operators in which the interplay between Higgs and top physics is most evident, taking place in the gluon fusion and  $t\bar{t}$  associated Higgs production processes. It is well known that there is a degeneracy in gluon fusion between  $C_{HG}$  and  $C_{tH}$  that prevents it from being used as a robust indirect constraint on the top Yukawa coupling, or conversely, heavy coloured particles that couple to the Higgs. This strongly motivates the direct measurement of the top Yukawa, via  $t\bar{t}H$ . In figure 7, the yellow ellipses show results for Higgs data, without ttH, while the green ellipses show the sensitivity for Higgs data including it, indicating how a relative flat direction in this plane (top left panel) is lifted by the inclusion of  $t\bar{t}H$ . However, despite not being directly related to Higgs couplings, both  $C_{tG}$  and  $C_G$  can also affect Higgs(+jet) production in gluon fusion. Thankfully, these can be constrained by top data, particularly  $t\bar{t}$  (and multi-jet data at order  $\Lambda^{-4}$ , for the latter). The results for the combination of Higgs with top data are shown as mauve ellipses, where we see in each plane a very substantial reduction of the area allowed at the 95% CL. The difference between the two sensitivities underlines the strength and importance of this data in indirectly pinning down BSM interactions of the Higgs, where now the  $C_{tG}$  and  $C_{G}$  directions are squeezed down by an order of magnitude. We also see that several (anti)correlations between pairs of operator coefficients are suppressed when top data are included, most noticeably in the  $(C_{HG}, C_G)$  plane. However, using top data to constrain only two operators is not in keeping with the global spirit of



Figure 7. Constraints on the indicated pairs of operator coefficients at the 95% confidence level, marginalised over the remaining degrees of freedom in the 'Higgs only' operator set. The green and mauve shaded areas correspond to combined linear fits to Higgs data and Higgs + top data, respectively. The blue ellipses indicate the marginalised constraints from Higgs + top data after introducing top-quark four-fermion operators into the fit, and the yellow ellipses are obtained from a fit dropping the  $t\bar{t}H$  data.

SMEFT interpretations, especially given the large number of degrees of freedom discussed in section 2.2.2 that could potentially dilute its power to bound  $C_{tG}$  and  $C_G$ . We address this question by increasing our operator subset to include the 7 four-fermion operators that impact  $t\bar{t}$  production, with the new marginalised constraints shown as blue ellipses. Surprisingly, there is little further change when adding the four-fermion operators, indicating a very limited dilution effect and underlining the robustness of the complementarity of top data in indirectly constraining Higgs couplings. This is especially encouraging given the fact that, as discussed in section 6.2, our constraints on this set of operators are significantly weakened by the linear approximation used in our analysis, allowing for larger marginalisation effects than a quadratic-level fit would.

Figure 8 displays the constraints on the same pairs of operator coefficients at the 95% confidence level when the coefficients of other operators are set to zero, with more fine-grained information on the constraints provided by the different datasets. The shaded regions are the results of linear fits to Higgs signal strengths and 0 jet STXS bins (blue),  $t\bar{t}H$  signal strengths (mauve),  $\geq 1$  jet STXS bins (orange),  $t\bar{t}$  data (green),  $t\bar{t}V$  data (red) and their combination (grey). The dashed ellipses show the constraints in the corresponding



Figure 8. Constraints on the indicated pairs of operator coefficients at the 95% confidence level, setting the other operator coefficients to zero. The shaded regions correspond to linear fits to Higgs signal strengths and 0 jet STXS bins (blue),  $t\bar{t}H$  signal strengths (mauve),  $\geq 1$  jet STXS bins (orange)  $t\bar{t}$  data (green),  $t\bar{t}V$  data (red) and their combination (grey). The dashed ellipses show the constraints obtained by marginalising over the remaining Wilson coefficients of the full fit.

parameter planes when marginalising over the remaining Wilson coefficients of the full fit, as shown in figure 7. As was to be expected, the constraints obtained when the other operator coefficients are set to zero are significantly stronger. The complementarity between ggF,  $t\bar{t}H$  and  $t\bar{t}$  is again evident, with  $t\bar{t}V$  data also providing some additional information on  $C_G$ . We also see that  $\geq 1$  jet STXS bins have not yet reached the level of precision needed to offer significant complementary information in this parameter space. However, we expect this to improve as increasingly fine-grained STXS binnings are measured.

# 6.3.3 The triple-gluon operator $C_G$

The operator  $\mathcal{O}_G$  consists of triple-gluon field strengths [230–232] and so affects any observables sensitive to jets [174–176, 233]. This includes many of the Higgs and top processes in our global fit, as shown in figure 12 that we discuss in the next section, where we see sizeable correlations of  $C_G$  with the operators  $C_{HG}$ ,  $C_{tH}$ ,  $C_{tG}$ ,  $C_{Qq}^{3,8}$ ,  $C_{Qq}^{1,8}$ ,  $C_{Qu}^8$ ,  $C_{tq}^8$ ,  $C_{tu}^8$ , and  $C_{td}^8$ , spanning both the Higgs and top sectors. Therefore, the gluonic operator complicates the SMEFT interpretation of the measurements in these sectors.

It has been argued in refs. [174–176] that a very strong constraint on  $C_G$  is provided by multijet data that are not included in our default data set, and that one can set  $C_G = 0$ 



Figure 9. The measurement of the  $t\bar{t}$  invariant mass distribution in the lepton+jets channel at 13 TeV by the CMS experiment [53] compared to the SM prediction at the NNLO QCD + NLO EW level [234]. Also shown are predictions corresponding to the best-fit values for  $\hat{C}_G$  (green upward triangles),  $\hat{C}_{tG}$  (red downward triangles), their combination (purple circles) and the global best-fit point in the full parameter space (orange stars).

when analysing electroweak, Higgs or top data. However, this strong constraint relies on quadratic contributions to multijet observables, whereas our global fit is made to linear order. The linear contributions of  $C_G$  are small since the amplitude involving  $C_G$  in gluongluon scattering does not interfere with the SM, and in quark-gluon scattering it does so only proportionally to the quark masses. This has made it a challenge to constrain in past studies [230–232]. However, the top sector, with its large quark mass, provides an opportunity to recover sensitivity at linear order to  $C_G$ , as studied most recently in ref. [235]. Table 7 indicates that  $t\bar{t}$  (43%) and  $t\bar{t}V$  data (56%) provide the entirety of the individual sensitivity to  $C_G$ . This is confirmed by comparing figures 3, 5 and 6, which also emphasise that these bounds are robust when marginalising over the other operators in both the Higgs and top sectors.

The  $C_G$  fit also shows the strongest pull away from zero, with a significance ~ 3 and 4  $\sigma$  for the best fit in the individual and marginalised cases, respectively, as shown in figure 6. This effect is due to  $t\bar{t}$  differential data, an example of which is given by the 13-TeV invariant mass distribution data shown in figure 9. We see there that the  $m_{t\bar{t}}$  dependence of the cross-section data, normalised to the SM prediction, denoted by black points, differs quite significantly from that of the SM, represented by grey boxes at a value of 1, and it is this discrepancy that pulls  $C_G$  into negative territory, as shown by the green triangles, plotting the best-fit  $C_G$  contribution. We see that the agreement with the data is obtained by an interplay between  $C_G$  and  $C_{tG}$ , whose best fit prediction, shown by the red triangles, improves the agreement in the low mass bins while not too significantly affecting the high mass region. The sum of the best-fit  $C_G$  and  $C_{tG}$  predictions, shown by the purple circles, coincides with the global best-fit prediction, shown by the orange stars, demonstrating that the fit to the data is obtained primarily by this interplay. We emphasise, however, that the significance of this effect could be reduced if there were some important contribution to the  $m_{t\bar{t}}$  distribution close to threshold that has not been included in the SM calculation. Moreover,



Figure 10. Same as figure 7, but setting  $C_G = 0$ .

such a large pull away from the SM in this case is not meaningful, in view of the potentially important quadratic contributions from  $C_G$ . What the linear fit demonstrates is the size of the linear constraint on  $C_G$ , which is not known *a priori*, and its dependence on other operators. Significant indirect effects of  $C_G$  on the other operators may also then be questioned.

Accordingly, we have investigated the consequences of assuming that  $C_G$  can be better constrained by including dedicated QCD multi-jet data, and have analysed the effects of setting  $C_G = 0$  in our study of the impact of top data on Higgs coupling measurements, as well as in our marginalised global fits in the flavour-universal  $SU(3)^5$  and the topspecific  $SU(2)^2 \times SU(3)^3$  scenarios. Figure 10 shows that the global space of constraints is changed compared to figure 7, especially when only including Higgs data (green ellipse), with improved sensitivity (see figure 11) and a different pattern of correlations. Once the top data is included, the overall sensitivity is improved, and the best-fit point before including the four-fermion operators (mauve ellipse) is closer to the SM. The relative impact of adding the four-fermion operators (blue ellipse) is more noticeable when  $C_G = 0$ than when it is non-zero. When  $C_G = 0$ , the tension with the SM in the  $t\bar{t}$  data pulls the four-fermion operators and  $C_{tG}$ . These operators also affect  $t\bar{t}H$ , and  $C_{tG}$  also modifies gluon fusion, leading to a cascade of shifts in the ranges of these operators. Moving to the global results, we see in the top two panels of figure 11 that this constraint has little effect on the 95% ranges we find in our marginalised  $SU(3)^5$  fit, except that the ranges of  $C_{HG}$  and  $C_{tH}$  are reduced noticeably when we set  $C_G = 0$ ,<sup>13</sup> as expected from the previous discussion of figure 10. In the  $SU(2)^2 \times SU(3)^3$  (bottom two panels of figure 11) there are shifts in the central values of several top operator coefficients, with the four fermion

<sup>&</sup>lt;sup>13</sup>For completeness, we also show in the top two panels of figure 11 the effects of dropping the EWPOs (and LEP WW) from the marginalised SU(3)<sup>5</sup> fit (yellow bars). As could be expected, there are large effects on the constraints on the operators that contribute most to the EWPOs, but quite small effects in the bosonic and Yukawa sectors.

operators moving further away from the SM to absorb the aforementioned discrepancies in top data. Overall, no significant reductions in the ranges of any operator coefficients are observed, and thus our fit results are relatively insensitive to the treatment of  $C_G$ .

#### 6.3.4 Correlation matrix and principal component analysis

The full correlation matrix for the top-specific marginalised fit is shown in figure 12, colourcoded and labelled in percentages. Some of these correlations can be explained intuitively by the simple fact that two operators contribute to the same observable, while others occur more indirectly, through a chain of dependencies that is difficult to trace through the inversion of the Fisher information matrix. We see that there are substantial (anti) correlations between the coefficients of operators in the EWPO and Bosonic sectors, and there are also many (anti)correlations within the top 4F sector. On the other hand, there is only one large off-diagonal entry in the top 2F sector, namely a negative correlation between  $C_{HQ}^{(3)}$  and  $C_{HQ}^{(1)}$ . Along with  $C_{Ht}$ , that is mildly correlated with these two, these operators affect  $Zt\bar{t}$  couplings. As previously discussed, one linear combination,  $C_{HQ}^{(3)} + C_{HQ}^{(1)}$ modifies the  $Zb\bar{b}$  coupling, and hence has a very strong LEP constraint, while  $C_{Ht}$  and the other combination of  $C_{HQ}^{(3)}$  and  $C_{HQ}^{(1)}$  can only be probed in EW top processes. The bottom Yukawa operator  $(C_{bH})$  exhibits some correlations with those of the top  $(C_{tH})$ and the tau  $(C_{\tau H})$ , and in the bosonic sector some moderate correlations are observed, notably between  $C_{HW}$ ,  $C_{HB}$  and  $C_{H\Box}$ , and between  $C_{HG}$  and  $C_G$  (as expected from the discussion in section 6.3.2). Turning to correlations between operators in different sectors, we note substantial (negative) correlations between  $C_{bH}$  in the Yukawa sector and  $C_{HWB}, (C_{HD}), C_{Hl}^{(3)}, C_{Hq}^{(3)}$  and  $C_{He}$  in the EWPO sector, and  $(C_{H\Box}), C_{HW}$  and  $C_{HB}$  in the bosonic sector respectively, as well as substantial positive (negative) correlations between  $C_{tH}$  in the Yukawa sector and  $C_{H\square}, C_G$  and  $(C_{HG})$  in the Bosonic sector. Finally, there are several large (anti)correlations between operators in the top 4F and top 2F sectors, namely  $C_{Qq}^{3,1}$  and  $C_{HQ}^{(3)}$  (positive),  $C_{Qq}^{3,1}$  and  $C_{HQ}^{(1)}$  (negative),  $C_{Qu}^{8}$  and  $C_{Ht}$  (negative), and  $C_{tq}^{8}$  and  $C_{Ht}$  (positive). Overall, there are 22 correlation coefficients with magnitude  $\geq 0.2$ between operators in a top sector on the one hand and in a Yukawa, bosonic or electroweak sector on the other hand. These and the top sector may not be talking to each other very loudly, but they are starting to whisper to each other.

In the Gaussian approximation to the global likelihood that is used here, it is also informative to diagonalise the constraints on the operator coefficients in the orthonormal eigenvector basis and perform a principal component analysis. This tells us which directions are most constrained in the fit, and what operators contribute to those directions. We display the constraints on the eigenvectors graphically in figure 13: the rows in the centre panel correspond to the different operator coefficients, the columns correspond to the different eigenvectors, and the colour-coded squares represent the moduli-squared of the operator components in the eigenvectors. The latter are ordered such that the strengths of the global constraints decrease from left to right, as seen in the top panel of figure 13, where the 95% CL bounds on the scales  $\Lambda$  are calculated assuming that the linear combination of operator coefficients making up that particular eigenvector is set to unity. The bottom panel tabulates the respective relative constraining powers of the electroweak pre-


Figure 11. Comparison of the constraints on the indicated marginalised operator coefficients  $C_i(1 \text{ TeV})^2/\Lambda^2$  (top and third panels) and the corresponding scales  $\Lambda$  for the indicated values of the  $C_i$  at the 95% confidence level (second and bottom panels), found in a combined linear fit to the Higgs, diboson and electroweak precision observables (top two panels) and including in addition top data (bottom two panels), including  $C_G$  in the fit (orange) and setting  $C_G = 0$  (red). Also displayed in yellow in the top two panels is a fit without LEP (EWPO and WW) measurements.



Figure 12. The full  $34 \times 34$  correlation matrix for the marginalised top-specific fit. The operators are grouped into those affecting primarily electroweak precision observables, bosonic observables, Yukawa measurements and top electroweak measurements, as well as top-quark four-fermion operators. The entries in the correlation matrix are colour-coded according to the indicated magnitudes of the correlation coefficients.

cision data, LEP diboson data, Higgs coupling strength measurements from Runs 1 and 2, STXS measurements, LHC diboson and  $Z_{jj}$  measurements,  $t\bar{t}$  measurements, single top measurements and  $t\bar{t}V$  measurements. These are defined as the relative contribution of each dataset to the corresponding entry of the diagonalised Fisher information. Entries where there is no significant constraining power are indicated by "-".

We see that the largest component of the best-constrained eigenvector is  $\mathcal{O}_{HB}$ , and that the limit on its scale exceeds 20 TeV, with the most important contribution coming from the STXS measurements, followed by Higgs signal strength measurements. The scales of four more eigenvectors are constrained beyond the 10-TeV level, with the most important contributions coming from the electroweak precision measurements as well as STXS and Higgs signal strength measurements. They can broadly be associated with the powerful sensitivity we have obtained in constraining the  $H\gamma\gamma$  and Hgg interactions. The first 12 eigenvectors are constrained by a mixture of EWPO and Higgs data, showing that these two sets are providing complementary and competitive bounds in the multi-TeV range.

We next see a particularly strong constraint coming entirely from single top data on  $C_{Qq}^{3,1}$  alone. We note that other operators contribute along that eigenvector direction, which is given by  $-0.98C_{Qq}^{3,1} - 0.17C_{HW} + 0.08C_{HQ}^{(3)}$ , but with too small a magnitude to be visibly coloured.<sup>14</sup> However, this is partly responsible for the large (anti-)correlations between  $C_{Qq}^{3,1}$  and  $C_{HQ}^{(3)}(C_{HQ}^{(1)})$  shown in figure 12, and  $C_{Qq}^{3,1}$  also appears in other directions with a small contribution. Several other examples of relatively isolated operators can be found across the figure, identified by the columns dominated by a single, very dark spot. Here the eigen-directions nearly coincide with a particular operator, such that the rest of the fit should be relatively independent of whether these are included or not. Specific examples are  $C_{\mu H}$ , which is constrained in isolation by the  $H \to \mu \mu$  signal strength, and  $C_{tW}$ , which is constrained mainly by W-helicity fraction measurements in  $t\bar{t}$  data. We also see that  $Z_{ij}$  mostly constrains  $C_W$  with not much effect on the rest of the fit. The relation between measurements and constraints on operators can be indirect, illustrating the complementarity between the different datasets: for example, the LHC WW and WZ diboson data are responsible for 46% of the constraining power along the eigenvector direction principally aligned with  $C_{HW}$ , despite the lack of  $C_{HW}$  dependence in diboson data. However, their inclusion helps to close directions of limited sensitivity in the fit to Higgs and electroweak data that are then better able to constrain e.g.  $C_{HW}$ .

The least strongly constrained eigenvector is predominantly  $\mathcal{O}_{tu}^8$ , with a scale bounded just above 200 GeV when the operator coefficients are normalised to unity, mainly by  $t\bar{t}$ data. It is followed by three more eigenvectors with scales ~ 300 GeV, whose principal components are top operators. The most important constraints on these eigenvectors are in the top sector, principally from the  $t\bar{t}$  and  $t\bar{t}V$  data. As discussed in section 6.2, while the validity of the SMEFT may be questioned when the operator coefficients are normalised to unity, it should be reliable for all top operators in the strong-coupling limit. Also, we expect the SMEFT to be valid for the better-constrained eigenvectors even for unit-normalised coefficients, since these eigenvectors have relatively small top operator components, as seen in the middle panel of figure 12.

<sup>&</sup>lt;sup>14</sup>For completeness, the numerical values of the eigenvector components are provided in table 8.



Figure 13. Constraints on the eigenvectors of the global likelihood function. The top panel shows the strengths of the global constraints on the eigenvectors. The rows in the centre panel correspond to the different operator coefficients, the rows correspond to the different eigenvectors, and the colour-coded squares represent the moduli-squared of the operator components to the eigenvectors. The bottom panel shows the constraining powers of the electroweak precision data, LEP diboson data, Higgs coupling strength measurements from Runs 1 and 2, STXS, LHC diboson and Zjj,  $t\bar{t}$ , single-top and  $t\bar{t}W/Z/\gamma$  measurements, respectively. Instances where there is no significant constraining power are indicated by "-". Explicit expressions for the eigenvectors are given in table 8.

#### 7 Constraints on UV completions

So far our approach has been to use the SMEFT framework to combine all relevant data inputs and perform a global fit to the coefficients of the dimension-6 operators characterising possible modifications of the SM Lagrangian at leading order in the momentum expansion. The result of this fit, summarised in figure 6, provides a model-independent way to evaluate the compatibility of the SM with the available experimental data. This SMEFT analysis provides information on the level of new physics contributions compatible with the current data. We presented in the bottom panel of figure 6 limits on the dimensionful parameters  $\Lambda/\sqrt{C_i}$ , which can be interpreted as constraints on the possible scale of New Physics compatible with current measurements. The individual limits correspond to bounds on a single operator assuming all others are zero, while the marginalised limits allow all other coefficients to vary. These can respectively be taken as optimistic and pessimistic estimates of the sensitivity, and we expect realistic models to generate some intermediate subset.

In this section we go a step further in the interpretation of our fit and explore how specific UV completions of the SM Lagrangian are constrained by current measurements. In any given model, the global analysis we have presented is often not directly applicable, as typical models may generate more than just one of the dimension-6 SMEFT operators, but not all of them. Moreover, when a specific model contributes to more than a single operator, these contributions are often related, corresponding to a smaller subset of independent parameters.

To illustrate these model-dependent effects we have considered several model interpretations. In section 7.1 we discuss models in which SMEFT operators are induced at the tree level, and in section 7.2 we discuss classes of UV completions that share similar SMEFT patterns. We then analyze supersymmetric models with TeV-scale stops in which SMEFT operators are induced at the one-loop level in section 7.3. Finally, in section 7.4 we present results from a survey of the pulls for all fits with non-vanishing coefficients for combinations of 2, 3, 4 and 5 operators.

#### 7.1 Simple tree-level-induced SMEFTs

We first study the implications of our analysis for single-field extensions of the SM that contribute to the SMEFT at tree level. This exercise updates the one presented in [160] and is based on the dictionary provided in [143].<sup>15</sup>

We list in table 3 the  $SU(3) \times SU(2) \times U(1)$  quantum numbers and couplings of the new fields considered here. We assume flavour-universal couplings in all cases except T and TB, in which the new fields couple only to the third generation. These two models are taken from [146]. We consider only the renormalisable contributions from each single field extension. In the case of the models B and W (a Z' and W' respectively) we consider only their couplings to the Higgs doublet and set all fermion couplings to zero. In addition to evaluating the constraints on these single-field extensions, we also consider the following two combinations of the fields in table 3 that yield single-parameter models via cancellations

 $<sup>^{15}</sup>$ We note that these one-parameter extensions of the SM have been included among the BSM benchmark proposals made by the LHC Higgs Working Group [145].

Name	Spin	SU(3)	SU(2)	U(1)	Param.	Name	Spin	SU(3)	SU(2)	U(1)	Param.
S	0	1	1	0	$(M_S, \kappa_S)$	$\Delta_1$	$\frac{1}{2}$	1	2	$-\frac{1}{2}$	$(M_{\Delta_1}, \lambda_{\Delta_1})$
$S_1$	0	1	1	1	$(M_{S_1}, y_{S_1})$	$\Delta_3$	$\frac{1}{2}$	1	2	$-\frac{1}{2}$	$(M_{\Delta_3},\lambda_{\Delta_3})$
$\varphi$	0	1	2	$\frac{1}{2}$	$(M_{\varphi}, Z_6 \cos \beta)$	Σ	$\frac{1}{2}$	1	3	0	$(M_{\Sigma}, \lambda_{\Sigma})$
Ξ	0	1	3	0	$(M_{\Xi},\kappa_{\Xi})$	$\Sigma_1$	$\frac{1}{2}$	1	3	-1	$(M_{\Sigma_1},\lambda_{\Sigma_1})$
$\Xi_1$	0	1	3	1	$(M_{\Xi_1}, \kappa_{\Xi_1})$	U	$\frac{1}{2}$	3	1	$\frac{2}{3}$	$(M_U, \lambda_U)$
В	1	1	1	0	$(M_B, \hat{g}_H^B)$	D	$\frac{1}{2}$	3	1	$-\frac{1}{3}$	$(M_D, \lambda_D)$
$B_1$	1	1	1	1	$(M_{B_1}, g_{B_1})$	$Q_1$	$\frac{1}{2}$	3	2	$\frac{1}{6}$	$(M_{Q_1},\lambda_{Q_1})$
W	1	1	3	0	$(M_W, \hat{g}_H^W)$	$Q_5$	$\frac{1}{2}$	3	2	$-\frac{5}{6}$	$(M_{Q_5}, \lambda_{Q_5})$
$W_1$	1	1	3	1	$(M_{W_1}, \hat{g}_{W_1}^{\varphi})$	$Q_7$	$\frac{1}{2}$	3	2	$\frac{7}{6}$	$(M_{Q_7}, \lambda_{Q_7})$
N	$\frac{1}{2}$	1	1	0	$(M_N,\lambda_N)$	$T_1$	$\frac{1}{2}$	3	3	$-\frac{1}{3}$	$(M_{T_1},\lambda_{T_1})$
E	$\frac{1}{2}$	1	1	-1	$(M_E, \lambda_E)$	$T_2$	$\frac{1}{2}$	3	3	$\frac{2}{3}$	$(M_{T_2},\lambda_{T_2})$
Т	$\frac{1}{2}$	3	1	$\frac{2}{3}$	$(M_T, s_L^t)$	TB	$\frac{1}{2}$	3	2	$\frac{1}{6}$	$(M_{TB}, s_L^{t,b})$

Table 3. Single-field extensions of the SM constrained by our analysis.

(see ref. [145]), i.e., models that depend on only a single coupling ( $\lambda$  or  $g_H$ ), as well as a mass M: 1) Quark bidoublet model: { $Q_1, Q_7$ } with equal masses M and equal couplings  $\lambda$  to the top quark, and 2) Vector-singlet pair model: { $B, B_1$ } with equal masses M and Higgs couplings proportional to  $g_H$ . We exhibit in tables 4 and 5 the contributions made at tree level by exchanges of each of these fields to the SMEFT coefficients.<sup>16</sup> The numbers shown in the tables should each be multiplied by the appropriate squared coupling factors and divided by the square of the mass scale M.<sup>17</sup>

We show in figure 14 the results from our global fit for all of the one-parameter singlefield extensions of the SM. In each of these models we constrain a positive quantity:  $|\lambda|^2$ . The constraints in figure 14 are found using the numerical MCMC fitter described in appendix B. This method allows us to incorporate the constraint  $|\lambda|^2 > 0$  as a Heaviside prior  $\pi(|\lambda|^2 < 0) = 0$ . The 2- $\sigma$  constraints on the mass scales in TeV units, assuming that the corresponding couplings are set to unity, are shown as horizontal bars.<sup>18</sup> We note that most of these limits exceed 1 TeV for unit coupling and do not depend on kinematic distributions probing this region, in which case the SMEFT approach is self-consistent. The SU(2)-singlet VLQ top-partner model (T) and S are the most poorly constrained, with  $m_{S,T} > 900$ , 770 GeV. We also show in grey boxes the corresponding bounds on the squared couplings, assuming a mass scale of 1 TeV. Most of the bounds are < 1, justifying a tree-level treatment. We also list all the pulls that exceed 1- $\sigma$ , which is significant.

We can compare the mass limits for these models with the naive scale limits shown in figure 6. In a model-independent SMEFT analysis, one allows all the EFT operators

<sup>&</sup>lt;sup>16</sup>In general, the coloured, vector-like fermions contribute at one-loop order to  $C_{HG}$ . We include this contribution for the 2-parameter model TB, and verify in a representative example (the T field) that it has a negligible effect on the single-parameter model constraints.

<sup>&</sup>lt;sup>17</sup>We do not provide limits on the two-parameter model  $\Xi_1$ , which has a complex coupling  $g_{\Xi_1}$ , but note that it behaves similarly to models  $\Xi$  and  $S_1$ .

<sup>&</sup>lt;sup>18</sup>In the case of the T (vector-like quark) model, the mass limit has been obtained using the relation  $s_L^t \simeq \lambda v / \sqrt{2} M_T$  and setting  $\lambda = 1$  [146].

Model	$C_{HD}$	$C_{ll}$	$C_{Hl}^3$	$C_{Hl}^1$	$C_{He}$	$C_{H\square}$	$C_{\tau H}$	$C_{tH}$	$C_{bH}$
S						$-\frac{1}{2}$			
$S_1$		1							
Σ			$\frac{1}{16}$	$\frac{3}{16}$			$\frac{y_{\tau}}{4}$		
$\Sigma_1$			$-\frac{1}{16}$	$-\frac{3}{16}$			$\frac{y_{\tau}}{8}$		
N			$-\frac{1}{4}$	$\frac{1}{4}$					
E			$-\frac{1}{4}$	$-\frac{1}{4}$			$\frac{y_{\tau}}{2}$		
$\Delta_1$					$\frac{1}{2}$		$\frac{y_{\tau}}{2}$		
$\Delta_3$					$-\frac{1}{2}$		$\frac{y_{\tau}}{2}$		
$B_1$	1					$-\frac{1}{2}$	$-\frac{y_{\tau}}{2}$	$-\frac{y_t}{2}$	$-\frac{y_{b}}{2}$
Ξ	-2					$\frac{1}{2}$	$y_{\tau}$	$y_t$	$y_b$
$W_1$	$-\frac{1}{4}$					$-\frac{1}{8}$	$-\frac{y_{\tau}}{8}$	$-\frac{y_t}{8}$	$-\frac{y_{b}}{8}$
$\varphi$							$-y_{\tau}$	$-y_t$	$-y_b$
$\{B, B_1\}$						$-\frac{3}{2}$	$-y_{\tau}$	$-y_t$	$-y_b$
$\{Q_1, Q_7\}$								$y_t$	
Model	$C_{Hq}^3$	$C^1_{Hq}$	$(C^{3}_{Hq})_{33}$	$(C^{1}_{Hq})_{33}$	$C_{Hu}$	$C_{Hd}$	$C_{tH}$	$C_{bH}$	
U	$-\frac{1}{4}$	$\frac{1}{4}$	$-\frac{1}{4}$	$\frac{1}{4}$			$\frac{y_t}{2}$		
D	$-\frac{1}{4}$	$-\frac{1}{4}$	$-\frac{1}{4}$	$-\frac{1}{4}$				$\frac{y_b}{2}$	
$Q_5$						$-\frac{1}{2}$		$\frac{y_b}{2}$	
$Q_7$					$\frac{1}{2}$		$\frac{y_t}{2}$		
$T_1$	$-\frac{1}{16}$	$-\frac{3}{16}$	$-\frac{1}{16}$	$-\frac{3}{16}$			$\frac{y_t}{4}$	$\frac{y_b}{8}$	
$T_2$	$-\frac{1}{16}$	$\frac{3}{16}$	$-\frac{1}{16}$	$\frac{3}{16}$			$\frac{y_t}{8}$	$\frac{y_b}{4}$	
Т			$-\frac{1}{2}\frac{M_T^2}{v^2}$	$\frac{1}{2} \frac{M_T^2}{v^2}$			$y_t \frac{M_T^2}{v^2}$		

**Table 4.** Operators generated at tree level by the single-field extensions listed in the first column. Each extension depends on a single coupling (see table 3) as well as a new physics mass-scale M. The coefficients of the operators are each proportional to the squares of the corresponding coupling  $\lambda$  by the corresponding entry in the table and divided by  $M^2$ .  $y_t$ ,  $y_b$  and  $y_{\tau}$  denote the top, bottom and tau Yukawa couplings respectively, v denotes the electroweak scale and  $\alpha_s$  denotes the strong coupling.

Model	$C_{HD}$	$C_{H\square}$	$C_{\tau H}$	$C_{tH}$	$C_{bH}$
В	$-2a^{2}$	$-\frac{1}{2}(a^2-b^2)$	$-aby_{\tau}$	$-aby_t$	$-aby_b$
W	$\frac{1}{2}b^2$	$-\frac{1}{8}(3a^2+b^2)$	$-\frac{1}{4}y_{\tau}(a+b)^2$	$-\frac{1}{4}y_t(a+b)^2$	$-\frac{1}{4}y_b(a+b)^2$
Model	$C_{tH}$	$C_{bH}$	$C_{Ht}$	$C_{HG}$	
TB	$\frac{M_{TB}^2}{v^2}y_ta^2$	$rac{M_{TB}^2}{v^2}y_bb^2$	$-\frac{M_{TB}^2}{v^2}a^2$	$-\frac{M_{TB}^2}{v^2} \frac{\alpha_s(0.65)}{8\pi} b^2$	

**Table 5.** Operator coefficients generated by the tree-level single-field models B, W and TB, which each depend on two couplings a and b, with  $y_t$ ,  $y_b$  and  $y_\tau$  denoting the top, bottom and tau Yukawa couplings respectively, v denoting the electroweak scale and  $\alpha_s$  denoting the strong coupling. The coefficients of all operators are proportional to the corresponding entries in the table and divided by  $M^2$ .



Figure 14. The horizontal bars show the mass limits (in TeV) at the 95% CL for the models described in table 3, setting the corresponding couplings to unity. The coupling limits obtained when setting the mass to 1 TeV are shown in grey boxes. We also note in light blue the pulls that exceed  $1-\sigma$ .

to vary simultaneously. On the other hand, in specific models not all EFT coefficients are generated and those that do appear are related to each other in such a way that the number of free parameters of the model is matched to the number of independent EFT operators generated by the model. As an example, we discuss the following set of single-parameter models:

$$\Sigma, \Sigma_1, N \text{ and } E$$

which all span the same types of SMEFT operators. They are all characterized by non-zero values for the following set of operators involving electroweak precision lepton observables:

$$C_{H\ell}^{1,3} \neq 0$$

whereas the other operator coefficients are zero, or very mildly constrained (e.g.,  $C_{\tau H}$ , which is  $\propto y_{\tau}$ ). Interpreted in terms of these models, the global SMEFT fit leads to mass limits of the order of 5 TeV for unit couplings, or corresponding coupling limits of  $\mathcal{O}(10^{-1})$ for TeV resonances. Each of these particles generates a different relation between  $C_{H\ell}^1$  and  $C_{H\ell}^3$ , leading to slightly different limits, e.g., for the model with a new neutral fermion None expects  $C_{H\ell}^1 = -C_{H\ell}^3 = |\lambda_N|^2/4m_N^2$ . The sizes of the mass limits justify the SMEFT approach, and the small couplings for masses of a TeV justify working at tree level.

The constraints on particles beyond the SM would be weaker if their effects were not tree-level but loop-induced. These are typical of extensions of the SM where couplings to new states have to be in pairs, as would be the case if they carry a new conserved quantum number. We discuss in section 7.3 one particularly interesting example with such loop-induced effects, namely stops in an *R*-parity-conserving supersymmetric model.

#### 7.2 Tree-level SMEFT patterns

As already commented, and displayed in table 4, simple extensions of the SM exhibit specific patterns in the operators they generate. Many operators have vanishing coefficients and those that are non-zero are often related. For example, model  $B_1$  of table 3 generates the bosonic operators  $C_{H\square}$ ,  $C_{HD}$  and Yukawa operators such as  $C_{tH}$ . These three operators are related by  $C_{H\square} = C_{tH} = -\frac{1}{2}C_{HD}$ . Similarly, model  $W_1$  generates a pattern  $C_{H\square} = C_{tH} = \frac{1}{2}C_{HD}$ . Motivated by these patterns, we study the results of our fit in four subspaces of the SMEFT:

Boson-specific : 
$$(C_{HD}, C_{H\Box}, C_{tH})$$
,  
Lepton-specific :  $(C_{He}, C_{H\ell}^{(1,3)}, C_{\ell\ell})$ ,  
Quark-specific :  $(C_{Hu}, C_{Hd}, C_{Hq}^{(1,3)}, C_{tH})$ ,  
Top-specific :  $((C_{Ha}^{(1)})_{33}, (C_{Ha}^{(3)})_{33}, C_{HG}, C_{bH}, C_{tH}, C_{Ht})$ .

Results for the boson-specific scenario are shown in the top two panels of figure 15. We display 95% CL contours in the  $(C_{H\square}, C_{tH})$  plane as solid contours, marginalising over  $C_{HD}$ , and setting all other operator coefficients to zero. In the top panel we show how these operators are constrained in the cases of four specific UV models from table 3: S,  $\varphi$  [which may be derived from a 2-Higgs Doublet Model (2HDM)],  $Q_1\&Q_7$  and  $B\&B_1$ , showing a detail in the inset. As well as these 1-parameter models, we project two of the 2-parameter models shown in table 5 onto the  $(C_{H\square}, C_{tH})$  plane: B and W. These cases are vector bosons with couplings  $\hat{g}_H^B$  and  $\hat{g}_H^W$  to the Higgs doublet, also known as Z' and W' bosons, respectively. Model B projects onto a line in the  $(C_{H\square}, C_{tH})$  plane, illustrated in the top panel for  $r_B = 2, -2$ , where  $r_B = \text{Re}(\hat{g}_H^B)/\text{Im}(\hat{g}_H^B)$ . Model W also projects onto a line, illustrated here for  $r_W = -1, \frac{1}{3}$ . While the slope of the line generated by model B is free to take any value, the line generated by model W is constrained to lie within the wedge bounded by the  $r_W = -1, \frac{1}{3}$  lines shown.

In the middle panels of figure 15 we zoom in on the yellow ellipse shown in the top panel, so as to study the constraints in the  $(C_{H\square}, C_{tH})$  plane when  $C_{H\square} \propto C_{HD}$ . This



Figure 15. Constraints at the 95% CL on  $(C_{H\Box}, C_{tH})$  in the boson-specific scenario (upper two panels) and  $(C_{Hl}^{(1)}, C_{He})$  in the lepton-specific scenario (bottom panel). The lines correspond to the 2- $\sigma$  limits obtained when we restrict the operators to the relations generated by integrating out the indicated single-field extensions of the SM.

results in tighter constraints on  $C_{H\square}$  compared to when  $C_{HD}$  is treated as an independent parameter, as shown by the small ellipses in the left panel. The constants of proportionality (-2, -4, 2) are those found in the patterns generated by models  $B_1$ ,  $\Xi$  and  $W_1$ . Zooming in further in the right panel, these ellipses are squashed into near-vertical parallel lines showing the constraints on  $C_{H\square}$  and  $C_{tH}$  in the case of each of these 1-parameter models.

The lepton-specific scenario is shown in the bottom panel of figure 15, where the coefficients of the operators  $C_{ll}$ ,  $C_{Hl}^{(1)}$ ,  $C_{Hl}^{(3)}$  and  $C_{He}$  are studied. The ellipses show how the data constrain  $C_{Hl}^{(1)}$  and  $C_{He}$ , marginalising over  $C_{Hl}^{(3)}$ . In the yellow ellipse we also marginalise over  $C_{ll}$ , allowing for the nonzero values of  $C_{ll}$ . The constraints shrink when  $C_{ll} = 0$ , as in models  $\Sigma$ ,  $\Sigma_1$ , N, E,  $\Delta_1$  and  $\Delta_3$ .<sup>19</sup> The inset plot demonstrates how  $C_{Hl}^{(1)}$  and  $C_{He}$  are constrained when we restrict these operators to the patterns associated with each of these 1-parameter models.

In figure 16 we turn to quark-Higgs interactions. The flavour-universal quark-specific scenario is shown in the upper panel, where we constrain  $(C_{Hq}^{(1)}, C_{Hu})$ , marginalising over the remaining quark-Higgs operators  $C_{Hq}^{(3)}$ ,  $C_{Hd}$  and  $C_{tH}$ . Models that generate these operators often lead to the pattern  $C_{Hq}^{(1)} \propto C_{Hq}^{(3)}$ . The green and red ellipses show how such patterns narrow the constraints on  $C_{Hu}$  and  $C_{Hq}^{(1)}$  for two examples:  $C_{Hq}^{(3)} = \pm C_{Hq}^{(1)}$ . Specialising to the 1-parameter models  $U, D, T_1, T_2$  and  $Q_7$  further restricts the operators, leading to the 1-dimensional constraints shown in the inset plots.

Finally, the lower panel of figure 16 considers the flavour-non-universal top-specific scenario, where we consider the operators  $(C_{Hq}^{(1)})_{33}$ ,  $(C_{Hq}^{(3)})_{33}$ ,  $C_{HG}$ ,  $C_{bH}$ ,  $C_{tH}$  and  $C_{Ht}$ . These operators are generated by the vector-like quark models T and TB with couplings to the third generation quarks only. Integrating out the  $SU(2)_L$  singlet T generates the pattern  $(C_{Hq}^{(3)})_{33} = -(C_{Hq}^{(1)})_{33}$ ,  $C_{bH} = C_{Ht} = 0$ . The green ellipse demonstrates how this pattern tightens the constraints on  $C_{tH}$ . In contrast, the  $SU(2)_L$  doublet TB does not generate the  $(C_{Hq}^{(3)})_{33}$  operator. Setting  $(C_{Hq}^{(3)})_{33} = 0$  results in much narrower constraints in the  $(C_{Hq}^{(1)})_{33}$  direction, as shown by the red ellipse.

These patterns, and the results shown in figures 15 and 16, may be considered as more general explorations of the model parameter space than in the two previous sections. Readers exploring UV completions who are searching for the indirect LHC and LEP constraints on their models can match their scenario to the allowed ellipses in these figures. For example, models linked to neutrino physics could lead to the SMEFT pattern we have denoted as *lepton-specific*, whereas models with various additional scalars and gauge bosons would be contained among the *boson-specific scenarios*, and models with additional coloured particles could be included among the *quark-specific scenarios*.

#### 7.3 *R*-parity-conserving stop squarks at the 1-loop level

A particularly interesting loop-induced modification of the SM Lagrangian is *R*-parityconserving supersymmetry with a light stop sector. Whereas the discussion in the previous section of single-field tree-level models was motivated by simplicity, this scenario is motivated by the naturalness of the hierarchy between the electroweak scale and that of gravity

<sup>&</sup>lt;sup>19</sup>See also ref. [236] for a more detailed fit to these six vector-like lepton models.



**Figure 16.** Constraints at the 95% CL on  $(C_{Hq}^{(1)}, C_{Hu})$  in the quark-specific scenario and  $((C_{Hq}^{(1)})_{33}, C_{tH})$  in the top-specific scenario. The lines correspond to the  $2\sigma$  limits obtained when the operator coefficients are restricted to the relations generated by integrating out single-field extensions of the SM.



**Figure 17**. Limits from the global fit in the stop parameter plane,  $(\frac{X_t}{m_{\tilde{t}}}, m_{\tilde{t}})$ . The two panels correspond to the low and high tan  $\beta$  choices, 1 and 20 respectively.

or grand unification. A complete one-loop analysis of the light-stop scenario and a comparison with the SMEFT analysis was given in [150].

In presenting our results, we follow [148], assuming a common diagonal mass term  $m_{\tilde{t}}$ and denoting the stop mixing parameter by  $X_t$ . The constraints on degenerate stops are dominated by measurements of the  $H \to gg$  and  $H \to \gamma\gamma$  couplings, which constrain the dimension-6 operators  $C_{HG}$ ,  $C_{HB}$ ,  $C_{HW}$  and  $C_{HWB}$ . These constrain the stop parameters  $m_{\tilde{t}}$  and  $X_t$  through the following relations:

$$C_{HG} = \frac{g_s^2}{12} \frac{h_t^2}{(4\pi)^2} \left[ \left( 1 + \frac{1}{12} \frac{c_{2\beta}g'^2}{h_t^2} \right) - \frac{1}{2} \frac{X_t^2}{m_t^2} \right],$$

$$C_{HB} = \frac{17g'^2}{144} \frac{h_t^2}{(4\pi)^2} \left[ \left( 1 + \frac{31}{102} \frac{c_{2\beta}g'^2}{h_t^2} \right) - \frac{38}{85} \frac{X_t^2}{m_t^2} \right],$$

$$C_{HW} = \frac{g^2}{16} \frac{h_t^2}{(4\pi)^2} \left[ \left( 1 - \frac{1}{6} \frac{c_{2\beta}g'^2}{h_t^2} \right) - \frac{2}{5} \frac{X_t^2}{m_t^2} \right],$$

$$C_{HWB} = -\frac{gg'}{24} \frac{h_t^2}{(4\pi)^2} \left[ \left( 1 + \frac{1}{2} \frac{c_{2\beta}g^2}{h_t^2} \right) - \frac{4}{5} \frac{X_t^2}{m_t^2} \right],$$
(7.1)

where  $h_t \equiv \frac{m_t}{v}$ ,  $m_t$  denotes the top mass and  $\beta$  is related to the ratio of vacuum expectation values:  $\tan \beta = \frac{\langle H_u \rangle}{\langle H_d \rangle}$ . We calculate the constraints in the  $(\frac{X_t}{m_t}, m_t)$  plane for the representative values  $\tan \beta = 1$  and 20, which are shown in the left and right panels of figure 17, respectively.

One sees in both panels of figure 17 that current LHC data constrain the stop mass scale to  $\gtrsim 300 \text{ GeV}$ , except for  $|\frac{X_t}{m_t}| \sim 1.5$  [237], where partial cancellations reduce the sensitivity to the stop mass scale below 200 GeV. In these regions the SMEFT analysis gives only qualitative results. These blind directions could be eliminated with future measurements of the H+jet differential distribution [238].



Figure 18. Stacked histograms of the distribution of pulls obtained in fits to 2 (upper left), 3 (upper right), 4 (lower left) and 5 (lower right) parameter subsets. The subsets have been split into three categories: those that include only operators that affect  $t\bar{t}$  production (blue), those that do not include operators that affect  $t\bar{t}$  production (orange), and the rest (green). The dashed vertical lines mark the expected 95% ranges for the pull distributions.

# 7.4 Survey of combinations of multiple operators

In general, new physics beyond the Standard Model could be expected to contribute to the SMEFT via exchanges of more than just a single massive particle, just as, e.g., W and Z exchanges both contribute to the Fermi 4-fermion EFT of the weak interactions, and various mesons including vectors  $\rho$  and scalars  $\sigma$  contribute to the low-energy pionic EFT of QCD. Another example is provided by supersymmetry, where there might be a pair of relatively light stops, which contribute to four different dimension-6 operator coefficients, as discussed in the previous subsection.

With this motivation, we have surveyed all fits with contributions from any combination of two, three, four or five dimension-6 operators, namely 561, 5984, 46736 and 278256 combinations, respectively. For each combination  $\{\mathcal{O}_i\}$ , we calculate the pull that the corresponding fit exerts, given by:

$$P \equiv \sqrt{\chi_{\rm SM}^2 - \chi_{\{\mathcal{O}_i\}}^2} \,. \tag{7.2}$$



Figure 19. The percentage occurrence of a given operator in the combinations ranked in the top 10% of pulls found in fits to 2 (light blue), 3 (orange), 4 (green) and 5 (blue) parameter subsets.

Calculations of P for all of these combinations is possible only because, in the linear treatment that we have adopted in this paper, the calculations of the  $\chi^2_{\{O_i\}}$  are computationally undemanding.

figure 18 displays stacked histograms of the distributions of the pulls P obtained in fits to combinations of 2 (upper left), 3 (upper right), 4 (lower left) and 5 (lower right) operators  $\{O_i\}$ . In each panel, the blue histogram is for combinations that include only operators that affect  $t\bar{t}$  production, see figure 5, the orange histogram is for combinations that do not include any of these operators, and the green histogram is for the remaining combinations.

In all panels of figure 18 we see that the blue histograms for combinations that include operators affecting  $t\bar{t}$  production exhibit peaks at P > 2 that move to > 4 for combinations of 5 operators. On the other hand, the orange histograms for combinations of operators that do not affect  $t\bar{t}$  production are peaked at lower values of  $P \leq 2$  and do not have long tails extending to large values of P, while the peaks of the green histograms are intermediate. The vertical lines in the panels of figure 18 mark the 95% ranges expected for Gaussian distributions of the pulls in the cases of 2, 3, 4 and 5 operators. We find that the percentage of combinations above the 95% marks range from 9% (2 operators) to 14% (5 operators) of the total number of combinations: more than expected for Gaussian distributions, but not excessive. Neither the orange nor the green histograms provide any indication of a significant deviation from the SM that can be described by any combination of 2, 3, 4 or 5 operators.

Figure 19 shows how often any given operator appears in the combinations whose total pulls P ranked among the top 10% of those obtained in fits with combinations of 2 (light blue), 3 (orange), 4 (green) and 5 (dark blue) operators. The operators affecting  $t\bar{t}$ production generally appear more often among these top 10% combinations, particularly  $\mathcal{O}_G$ . The prevalence of these operators in the high-pull combinations is due to the relatively poor quality of the global linear fit in the top sector that we discussed in section 6.2.<sup>20</sup>

 $<sup>^{20}</sup>$ As discussed there, this issue might be mitigated by including quadratic terms in the SMEFT expansion, but an analysis of this possibility lies beyond the scope of the present analysis.

Overall, we see that  $\mathcal{O}_G$  appears in more than half of the top 10% of combinations of  $\leq 4$  operators, falling to somewhat over 40% for combinations of 5 operators. For comparison, almost 60% of the 5-operator combinations include  $\mathcal{O}_{Qq}^{3,8}$ , whereas this operator appears in smaller proportions of the 2-, 3- and 4-operator combinations.

This survey would suggest that the best prospects for BSM physics may be among the operators affecting  $t\bar{t}$  production, particularly  $\mathcal{O}_{Qq}^{3,8}$ . However, we would emphasise that the data in the top sector are currently the least precise, and that there may be an issue with the  $t\bar{t}$  mass distribution near threshold, as discussed in section 6.2. As the constraints on the scales of the top operators are relatively weak, the resolution of this issue may require including quadratic contributions of these operators, which are not yet all available. We recall in this connection that an analysis of  $C_G$  including quadratic contributions to multijet production found a strong constraint restricting  $C_G$  to small values below the sensitivity of our analysis.

There are many physics scenarios that suggest the appearance of BSM physics in the top sector, such as the light stop scenario discussed in section 7.3. However, as we see in eq. (7.1), the operators  $\{\mathcal{O}_{HG}, \mathcal{O}_{HB}, \mathcal{O}_{HW}, \mathcal{O}_{HWB}\}$  that are most constrained in this scenario contribute primarily in the EW and Higgs sectors, rather than the top sector. We find a pull P = 1.9 for the combination  $\{\mathcal{O}_{HG}, \mathcal{O}_{HB}, \mathcal{O}_{HW}, \mathcal{O}_{HWB}\}$ , which is typical for 4-operator combinations that do not include  $t\bar{t}$  operators (the orange histogram in the lower left panel of figure 18), and the maximum pull when a fifth operator is included is P = 3.2, which occurs in the combination with  $\mathcal{O}_G$ , and is typical of the green histogram in the lower right panel of figure 18. So we find no hint of light supersymmetry in the current data.

That said, this type of broad-brush survey of operator combinations may be a useful way to help optimise the search for BSM physics using the SMEFT in the future.

# 8 Summary and conclusions

In this paper we have introduced a new tool, Fitmaker, to make a global analysis of the available top, Higgs, diboson and electroweak data in the framework of the SMEFT with dimension-6 operators included to linear order. We have presented results for fits including each operator individually, and also when marginalising over all the other operators. In each case, we have presented results in an SU(3)<sup>5</sup> flavour-symmetric scenario and in an  $SU(2)^2 \times SU(3)^3$  top-specific scenario. Our results are displayed in figure 6, with numerical results for the  $SU(2)^2 \times SU(3)^3$  top-specific scenario presented in table 6 below. We find  $\chi^2/dof = 0.94$  for our flavour-universal global fit and 0.81 for our top-specific fit, to be compared with  $\chi^2/dof = 0.93$  and 0.91, respectively, in the SM. For C = 1, the constraints on the scales  $\Lambda$  of the coefficients of many operators contributing to Higgs and electroweak measurements are individually  $\mathcal{O}(10)$  TeV, but constraints in the top sector are currently less precise than the Higgs and electroweak data, falling to  $\sim 100$  GeV in the case of the operator  $\mathcal{O}_{tB}$ .<sup>21</sup> This restricts the interpretation in many cases to strongly-coupled models,

 $<sup>^{21}</sup>$ However, this operator is largely uncorrelated with the other operators in our fit, as found in our principal component analysis in section 6.3.4.

where we find that  $\Lambda \gtrsim 1 \text{ TeV}$  for all operators when  $C = (4\pi)^2$  at the perturbativity upper limit.

We do not find any significant discrepancy with the SM. However, the data in the top sector show a preference for a non-zero value for the coefficient  $C_G$  of the triple-gluon operator  $\mathcal{O}_G$ , which is mirrored by trends in the coefficients of other operators affecting  $t\bar{t}$  measurements. This deviation can be traced back to the behaviour near threshold of the  $t\bar{t}$  cross section, and we await with interest future experimental measurements and developments in their theoretical understanding. We note also that a fit to multijet data at quadratic order in  $C_G$  has constrained it to values so small that it would not contribute significantly to the measurements we consider. However, we have not included this constraint in our linear fit, for reasons of theoretical consistency.

We have presented the  $34 \times 34$  correlation matrix for the top-specific marginalised fit, grouping the operators into those affecting primarily electroweak precision observables, bosonic observables, Yukawa coupling measurements, top electroweak measurements and top-quark four-fermion operators. We find that the most important correlations are between operators in the electroweak, Higgs and Yukawa sectors, and between top electroweak and four-fermion operators. However, there are also some notable correlations between top four-fermion operators and bosonic operators, and between top electroweak and other electroweak operators. Overall, there are 24 instances of significant ( $\geq 20\%$ ) correlations between top operators and Yukawa, bosonic or electroweak operators, confirming the relevance of making a combined analysis of all sectors.

Analyzing data to linear order in the SMEFT operator coefficients, the global  $\chi^2$  function may be regarded as Gaussian, facilitating a principal component analysis, in which we diagonalise the  $\chi^2$  matrix, identify its eigenvectors and determine their eigenvalues, which are displayed in figure 13 and have the numerical values tabulated in table 8. The scale associated with the best-constrained eigenvector is > 20 TeV, with Higgs and STXS measurements playing the most important roles. Three other eigenvectors have scales > 10 TeV, with electroweak measurements also providing important constraints. The least-constrained eigenvector is essentially  $\propto \mathcal{O}_{tB}$ , whose scale may be as low as ~ 100 GeV but is largely uncorrelated with the other operators, and the scales of three other eigenvectors may be < 300 GeV. Care must therefore be taken regarding the validity of the EFT along those poorly-constrained directions.

We have analyzed the constraints our results provide on all the single-field extensions of the SM that contribute to SMEFT operator coefficients at the tree level. Normalising to unit couplings, the lower limits on the corresponding BSM particle masses range between > 1 TeV to > 10 TeV. The largest pulls  $P \equiv \sqrt{\chi^2_{\rm SM} - \chi^2_{\rm BSM}}$  are 1.6, and hence not significant. In some instances, a particular BSM particle may contribute to several operator coefficients, and we have analyzed the constraints in boson-, lepton-, quark- and top-specific subspaces of the SMEFT. We have also analyzed the constraints on low-mass stops, which contribute significantly to four operator coefficients at the one-loop level, finding they must weigh more than ~ 300 GeV when the stop mixing parameter  $X_t = 0$ .

Finally, we have surveyed the constraints on all possible 2-, 3-, 4- and 5-operator combinations, assuming that the other operators vanish. We find pulls that are insignificant for combinations that do not contain operators affecting  $t\bar{t}$  measurements, whereas the

pulls for combinations of  $t\bar{t}$  operators are larger. However, the same caveats apply to their interpretation as to the discussion of the  $t\bar{t}$  sector above. The full pull distributions including all operator combinations do not exhibit any significant features.

These examples indicate ways in which a global analysis of all the current data may be used to obtain the broadest possible, unbiased view on the nature of possible BSM physics within the assumptions of the SMEFT framework. If any specific model or pattern of SMEFT operators were to exert a significant pull, it would be a first indication of the direction of new physics, which could be followed up with a more focused study. However, the dataset we have included in our SMEFT analysis provides no significant indication of possible BSM physics: the only operator for which a non-zero coefficient is preferred is  $\mathcal{O}_G$ , and this preference is not very convincing. It is driven, in particular, by the threshold behaviour of the  $t\bar{t}$  production cross section, but an analysis of multijet data at quadratic order prefers much smaller values of  $C_G$ . More  $t\bar{t}$  data and theoretical understanding may be needed to resolve this discrepancy.

The fact that our analysis is restricted to linear order in the dimension-6 SMEFT operator coefficients is clearly a limitation, but this a consequence of our self-imposed consistency requirement, and not a limitation of the fitter methodology. Our Fitmaker code could be applied equally well at quadratic order in the dimension-6 SMEFT operator coefficients, but a consistent analysis to fourth order in the new physics scale would require including also the contributions of dimension-8 operators at linear order: see [208] for a discussion of their importance for Higgs measurements. There are many other extensions of our analysis that could be tackled with Fitmaker, including CP-violating effects, flavour observables, RGE running, and higher-order perturbative QCD and electroweak effects with the SMEFT at NLO.

The Fitmaker code can be obtained from the account @kenmimasu at the following Gitlab link: • Since it is built in a modular fashion, it can readily be expanded by the user adding more data.

### Acknowledgments

We thank Marcel Vos, Maria Moreno Llacer and Victor and Marco Miralles for suggesting the inclusion of  $t\bar{t}\gamma$ . K.M. thanks L. Mantani, O. Mattelaer, E. Vryonidou and C. Zhang for valuable discussions and help in obtaining the predictions. M.M. thanks B. C. Allanach for useful discussions. The work of J.E. was supported in part by the UK STFC via grants ST/P000258/1 and ST/T000759/1, and in part by the Estonian Research Council via grant MOBTT5. M.M. is supported by the University of Cambridge Schiff Foundation studentship and in part by the UK STFC via grants ST/P000681/1 and ST/T000694/1. K.M. is also supported by the UK STFC via grant ST/T000759/1. V.S. acknowledges support from the UK STFC via Grant ST/L000504/1 and Spanish national grant FPA2017-85985-P. T.Y. is supported by a Branco Weiss Society in Science Fellowship and partially by the UK STFC via the grant ST/P000681/1.

# Tables of numerical fit results

		Individual			Marginalised	
SMEFT	Best fit	95% CL	Scale	Best fit	95% CL	Scale
Coeff.	$[\Lambda = 1  \text{TeV}]$	range	$\frac{\Lambda}{\sqrt{C}}$ [TeV]	$[\Lambda = 1  {\rm TeV}]$	range	$\frac{\Lambda}{\sqrt{C}}$ [TeV]
$C_{HWB}$	0.00	[-0.0043, +0.0026]	17.0	0.18	[-0.36, +0.73]	1.4
$C_{HD}$	-0.01	[-0.023, +0.0027]	8.8	-0.39	[-1.6, +0.81]	0.91
$C_{ll}$	0.01	[-0.005, +0.019]	9.2	-0.03	[-0.084, +0.02]	4.4
$C_{Hl}^{(3)}$	0.00	[-0.01, +0.003]	12.0	-0.03	[-0.13, +0.055]	3.3
$C_{Hl}^{(1)}$	0.00	[-0.0044, +0.013]	11.0	0.11	[-0.19, +0.41]	1.8
$C_{He}$	0.00	[-0.015, +0.0071]	9.6	0.19	[-0.41, +0.79]	1.3
$C_{Hq}^{(3)}$	0.00	[-0.017, +0.012]	8.3	-0.05	[-0.11, +0.012]	4.1
$C_{Hq}^{(1)}$	0.02	[-0.1, +0.14]	2.9	-0.04	[-0.27, +0.18]	2.1
$C_{Hd}$	-0.03	[-0.13, +0.071]	3.1	-0.39	[-0.91, +0.13]	1.4
$C_{Hu}$	0.00	[-0.075, +0.073]	3.7	-0.19	[-0.63, +0.25]	1.5
$C_{H\square}$	-0.27	[-1, +0.47]	1.2	-0.9	[-3, +1.2]	0.69
$C_{HG}$	0.00	[-0.0034, +0.0032]	17.0	0.00	[-0.014, +0.0086]	9.4
$C_{HW}$	0.00	[-0.012, +0.006]	11.0	0.12	[-0.38, +0.62]	1.4
$C_{HB}$	0.00	[-0.0034, +0.002]	19.0	0.07	[-0.09, +0.22]	2.5
$C_W$	0.18	[-0.071, +0.42]	2.0	0.15	[-0.11, +0.4]	2.0
$C_G$	-0.46	[-0.77, -0.14]	1.8	-1.4	[-2.2, -0.72]	1.2
$C_{\tau H}$	0.01	[-0.015, +0.025]	7.1	0.01	[-0.016, +0.028]	6.7
$C_{\mu H}$	0.00	[-0.0057, +0.005]	14.0	0.00	[-0.0058, +0.005]	14.0
$C_{bH}$	0.00	[-0.016, +0.024]	7.1	0.01	[-0.034, +0.052]	4.8
$C_{tH}$	-0.09	[-1, +0.84]	1.0	1.5	[-2.8, +5.7]	0.48
$C_{HQ}^{(3)}$	0.01	[-0.032, +0.048]	5.0	-0.1	[-0.67, +0.46]	1.3
$C_{HQ}^{(1)}$	0.01	[-0.031, +0.049]	5.0	-0.01	[-0.59, +0.58]	1.3
$C_{Ht}$	0.87	[-1.2, +2.9]	0.7	6.6	[+2,+11]	0.47
$C_{tG}$	-0.01	[-0.1, +0.086]	3.2	0.36	[+0.12, +0.6]	2.0
$C_{tW}$	0.19	[-0.12, +0.51]	1.8	0.23	[-0.088, +0.55]	1.8
$C_{tB}$	-1.6	[-4.5, +1.2]	0.59	-1.4	[-5.2, +2.5]	0.51
$C_{Qq}^{3,1}$	0.06	[-0.043, +0.16]	3.2	0.05	[-0.071, +0.17]	2.9
$C_{Qq}^{3,8}$	-1.2	[-2.4, +0.036]	0.91	-6.8	[-18, +4.5]	0.3
$C_{Qq}^{1,8}$	-0.12	[ -0.56, +0.31 ]	1.5	-0.65	[-4.9, +3.6]	0.48
$C_{Qu}^8$	-0.6	[ -1.3, +0.06 ]	1.2	6.3	[-2.5, +15]	0.34
$C_{Qd}^8$	-1.4	[-2.9, +0.07]	0.83	1.8	[-9.5, +13]	0.3
$C_{tq}^8$	-0.4	[-0.85, +0.059]	1.5	-5.6	[-13, +2.2]	0.36
$C_{tu}^8$	-0.45	[-1.1, +0.23]	1.2	4.0	[-11, +19]	0.26
$C_{td}^8$	-1.0	$[\overline{-2.5, +0.38}]$	0.83	-0.42	[-12, +11]	0.29

**Table 6.** Table of numerical results in figure 6 from the global fit to the electroweak, diboson, Higgs and top data in the top-specific  $SU(2)^2 \times SU(3)^3$  scenario.

$C_i$	EWPO	LEP WW	Run 1 SS	Run 2 SS	STXS	LHC WW	WZ	Zjj	$t\bar{t}$	$W_{\text{hel.}}$	tX	$t\bar{t}V$
$C_{HWB}$	51	-	7	14	28	-	-	-	-	-	-	-
$C_{HD}$	100	-	_	_	-	-	-	—	-	-	-	-
$C_{ll}$	99	-	-	-	-	-	-	—	-	—	—	-
$C_{Hl}^{(3)}$	99	_	_	_	_	-	_	_	_	_	_	_
$C_{Hl}^{(1)}$	100	_	_	_	_	_	-	_	-	_	-	_
$C_{He}$	100	-	-	-	-	-	-	—	-	—	—	-
$C_{Hq}^{(3)}$	89	1	_	_	2	-	6	_	-	-	_	-
$C_{Hq}^{(1)}$	99	_	_	_	_	-	-	_	-	_	-	-
$C_{Hd}$	99	-	-	_	-	-	-	—	-	_	—	-
$C_{Hu}$	98	-	-	-	1	-	-	—	-	—	-	-
$C_{H\square}$	-	-	22	46	32	-	-	-	-	—	-	-
$C_{HG}$	-	-	22	42	36	-	-	—	-	_	—	-
$C_{HW}$	-	-	14	29	56	-	-	—	-	—	_	-
$C_{HB}$	-	-	14	29	57	-	-	—	-	—	—	-
$C_W$	-	3	—	_	-	-	13	84	-	—	-	-
$C_G$	-	-	-	_	-	-	-	—	43	—	—	56
$C_{\tau H}$	-	-	22	45	34	-	-	_	-	_	_	-
$C_{\mu H}$	-	-	5	95	_	-	-	—	-	—	—	-
$C_{bH}$	-	_	19	35	47	-	-	_	-	_	_	-
$C_{tH}$	-	-	21	45	34	-	-	_	-	_	_	-
$C_{HQ}^{(3)}$	99	_	_	_	-	_	-	_	-	_	_	-
$C_{HQ}^{(1)}$	100	_	_	_	-	_	-	_	-	_	_	-
$C_{Ht}$	-	-	_	_	-	-	-	_	-	_	_	100
$C_{tG}$	-	-	13	29	24	-	-	_	24	_	_	9
$C_{tW}$	-	-	_	-	-	-	-	_	-	84	15	_
$C_{tB}$	-	-	-	-	-	-	-	_	-	—	-	100
$C_{Qq}^{3,1}$	-	_	_	_	_	-	-	_	-	_	100	-
$C_{Qq}^{3,8}$	-	_	_	_	_	-	_	_	87	_	-	13
$C^{1,8}_{Qq}$	-	-	_	-	-	-	-	_	82	-	-	17
$C_{Qu}^8$	-	-	-		-	-	-		91			7
$C_{Qd}^8$	-	-	-	2	-	-	-	_	92	-	-	6
$C_{tq}^8$	-	-	-	1		-	-	-	89		-	10
$C_{tu}^8$	-	-	—	_	-	-	_	_	96		_	3
$C_{td}^8$	_	_	_	2	_	_	_	_	92	_	_	5

**Table 7.** Relative constraining power in percent of different datasets on each coefficient of the global fit individually. Entries below 1% are not displayed. 'SS',  $W_{\text{hel.}}$  and tX refer to Higgs signal strength, W-helicity fraction and single top data, respectively.

$2\sigma[\Lambda = 1 \text{TeV}]$	$\Lambda[{\rm TeV}]$	Eigenvector
0.0021	22	$+0.73C_{HB} - 0.51C_{HWB} - 0.37C_{HG} + 0.22C_{HW} - 0.07C_{Hl}^{(3)} - 0.06C_{bH}$
0.0036	17	$-0.79C_{HG} + 0.45C_{HWB} + 0.30C_{Hl}^{(3)} - 0.17C_{ll} + 0.15C_{HD} - 0.10C_{bH} - 0.09C_{He} + 0.09C_{\mu H}$
0.0042	15	$+0.51C_{HB}+0.46C_{Hl}^{(3)}+0.46C_{HG}+0.35C_{HWB}-0.25C_{ll}+0.22C_{HD}-0.16C_{He}+0.15C_{HW}-0.10C_{\mu H}-0.09C_{Hq}^{(3)}$
0.0054	14	$+0.99C_{\mu H}+0.12C_{HG}$
0.0066	12	$ -0.75C_{Hl}^{(1)} + 0.42C_{He} + 0.31C_{Hq}^{(3)} - 0.23C_{Hl}^{(3)} + 0.21C_{HWB} + 0.14C_{HB} - 0.12C_{ll} + 0.11C_{HD} + 0.08C_{HQ}^{(1)} + 0.08C_{HQ}^{(3)} + 0.06C_{Hu} $
0.015	8.3	$-0.56C_{Hq}^{(3)} + 0.49C_{He} + 0.37C_{Hl}^{(3)} - 0.35C_{HWB} - 0.26C_{ll} - 0.19C_{Hl}^{(1)} - 0.18C_{HB} - 0.10C_{Hu} - 0.10C_{HQ}^{(3)} - 0.10C_{HQ}^{(1)} + 0.07C_{Hd} + 0.06C_{HD} - 0.05C_{HW}^{(1)} - 0.05C_{Hq}^{(1)}$
0.019	7.3	$-0.62C_{He} - 0.51C_{Hl}^{(1)} - 0.50C_{HD} - 0.23C_{ll} - 0.14C_{Hq}^{(3)} - 0.13C_{HWB} - 0.07C_{\tau H} - 0.07C_{HB} + 0.06C_{Hl}^{(3)} - 0.012C_{HB} - 0.012C_{HB} - 0.000C_{HB} - 0.$
0.019	7.2	$-0.96C_{\tau H} + 0.27C_{bH}$
0.03	5.8	$-0.52 C_{HQ}^{(3)} - 0.52 C_{HQ}^{(1)} + 0.48 C_{Hq}^{(3)} - 0.34 C_{bH} + 0.17 C_{Hl}^{(3)} - 0.16 C_{ll} - 0.14 C_{HWB} - 0.13 C_{\tau H} - 0.09 C_{HB} + 0.06 C_{Hd} - 0.00 C_{HB} + 0.00 C_{Hd} - 0.00 C_{HB} + 0.00 C_{Hd} - 0.00 C_{HB} - 0$
0.035	5.3	$+0.88C_{bH} - 0.27C_{HQ}^{(3)} - 0.27C_{HQ}^{(1)} + 0.24C_{\tau H} + 0.12C_{Hq}^{(3)} - 0.10C_{HG}$
0.057	4.2	$-0.85C_{ll} + 0.29C_{Hl}^{(1)} - 0.26C_{Hl}^{(3)} + 0.18C_{Hq}^{(3)} + 0.17C_{HQ}^{(3)} + 0.17C_{HQ}^{(1)} - 0.11C_{HWB} + 0.07C_{bH} - 0.07C_{HB}$
0.086	3.4	$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$
0.1	3.2	$-0.98C_{Qq}^{3,1} - 0.17C_{HW} + 0.08C_{HQ}^{(3)}$
0.11	3	$ + 0.66C_{HW} - 0.39C_{Hl}^{(3)} - 0.37C_{Hq}^{(3)} - 0.21C_{HQ}^{(1)} + 0.20C_{HD} - 0.20C_{HQ}^{(3)} + 0.19C_{Hu} - 0.18C_{Qq}^{3,1} + 0.15C_{Hq}^{(1)} - 0.14C_{HB} + 0.12C_{HWB} - 0.10C_{He} - 0.07C_{bH} + 0.07C_{W} - 0.07C_{Hd} $
0.14	2.7	$+0.93C_{tG}+0.30C_{G}+0.12C_{Qq}^{1,8}+0.11C_{tq}^{8}+0.07C_{Qu}^{8}+0.06C_{tu}^{8}$
0.2	2.2	$+0.97C_{Hq}^{(1)} + 0.13C_{Hl}^{(3)} - 0.08C_{HD} - 0.08C_{Hd} + 0.08C_{HQ}^{(3)} + 0.07C_{HQ}^{(1)}$
0.25	2	$-0.99C_W + 0.07C_{Hu}$
0.28	1.9	$-0.92C_{Hu} + 0.24C_{HD} - 0.19C_{Hl}^{(3)} + 0.13C_{Hd} - 0.12C_{He} - 0.08C_{HQ}^{(3)} - 0.08C_{Hl}^{(1)} - 0.07C_{HQ}^{(1)} - 0.06C_{W} + 0.05C_{Hq}^{(1)} - 0.08C_{Hq}^{(1)} - 0.0$
0.31	1.8	$+0.57C_{Qq}^{1,8}-0.53C_{tq}^{8}+0.39C_{tu}^{8}-0.37C_{Qu}^{8}+0.21C_{Qq}^{3,8}+0.17C_{td}^{8}-0.16C_{Qd}^{8}$
0.32	1.8	$+1.00C_{tW}$
0.38	1.6	$+0.82C_G-0.35C_{tG}+0.27C_{tq}^8+0.24C_{Qq}^{1,8}+0.15C_{Qu}^8+0.10C_{Ht}+0.09C_{tu}^8+0.08C_{Qd}^8-0.06C_{HBox}+0.06C_{tH}+0.05C_{td}^8+0.05C_{td}^8+0.08C_{Qd}^8+0.06C_{HBox}+0.06C_{tH}+0.06C_{td$
0.51	1.4	$+0.97C_{Hd} + 0.17C_{Hu} + 0.10C_{HQ}^{(3)} + 0.09C_{Hq}^{(1)} - 0.06C_{Hl}^{(3)} + 0.05C_{HD}$
0.59	1.3	$-0.49C_{tq}^8 - 0.47C_{Qq}^{1,8} + 0.43C_G - 0.39C_{Qu}^8 - 0.31C_{tu}^8 - 0.17C_{Qd}^8 - 0.15C_{td}^8 - 0.14C_{Qq}^{3,8} + 0.07C_{Ht} - 0.07C_{HBox} + 0.06C_{tH}^{(3)} - 0.05C_{HQ}^{(1)} - 0.05C_{HQ}^$
0.77	1.1	$+0.70C_{HQ}^{(1)} - 0.69C_{HQ}^{(3)} + 0.11C_G - 0.09C_{Qq}^{3,1} + 0.06C_{Hd} - 0.06C_{Qq}^{1,8} - 0.05C_{Ht}$
1.1	0.96	$ + 0.59C_{HBox} - 0.58C_{HD} + 0.29C_{He} + 0.27C_{HWB} + 0.23C_{HW} - 0.19C_{Hu} + 0.14C_{Hl}^{(1)} - 0.10C_{tH} + 0.09C_{Ht} + 0.08C_{Hd} + 0.08C_{Hg} - 0.07C_{Qq}^{3,8} - 0.06C_{HQ}^{(1)} $
1.7	0.78	$-0.64C_{Qq}^{3,8} + 0.51C_{Qq}^{1,8} - 0.40C_{tu}^{8} + 0.29C_{Ht} - 0.16C_{td}^{8} - 0.12C_{Qu}^{8} - 0.12C_{HBox} - 0.12C_{G} + 0.08C_{HQ}^{(1)} - 0.07C_{HQ}^{(3)} - 0.06C_{tB} + 0.05C_{HD}$
2.1	0.7	$+0.73C_{HBox} + 0.44C_{HD} - 0.31C_{tH} - 0.22C_{He} - 0.20C_{HWB} - 0.19C_{HW} + 0.12C_{Hu} - 0.11C_{Hl}^{(1)} + 0.10C_G - 0.06C_{tB} - 0.06C_{Qu}^8 - 0.06C_{HB}$
2.8	0.6	$+0.85C_{tB}-0.31C_{Ht}-0.20C_{Qd}^{8}+0.19C_{Qu}^{8}+0.17C_{tH}-0.16C_{Qq}^{3,8}+0.13C_{HBox}-0.11C_{tq}^{8}-0.10C_{tu}^{8}+0.09C_{Qq}^{1,8}+0.09C_{QQ}^{1,8}+0.09C_{QQ}^{1,8}+0.09C_{QQ}^{1,8}+0.09C_{QQ}^{1,8}+0.09C_{QQ}^{1,8}+0.09C_{QQ}^{1,8}+0.09C_{QQ}^{1,8}+0.09C_{QQ}^{1,8}$
3.4	0.54	$-0.71C_{Ht} - 0.45\overline{C_{tB} + 0.40C_{tH} + 0.16C_{Qq}^{1,8} - 0.14C_{tu}^8 + 0.13C_{HBox} + 0.13C_{Qu}^8 - 0.13C_{Qq}^{3,8} - 0.12C_{tq}^8 - 0.10C_{td}^8 + 0.06C_{Qd}^8 + 0.06C_G$
4.4	0.48	$+0.82C_{tH}+0.46C_{Ht}+0.25C_{HBox}+0.10C_{HD}+0.09C_{tu}^8-0.09C_G+0.07C_{Qq}^{3,8}+0.07C_{tq}^8-0.06C_{Qd}^8$
9.0	0.33	$+0.55C_{Qu}^{8}-0.46C_{td}^{8}-0.40C_{Qd}^{8}+0.36C_{Qq}^{3,8}-0.26C_{tq}^{8}+0.19C_{Ht}-0.19C_{tu}^{8}+0.15C_{Qq}^{1,8}-0.13C_{tH}-0.12C_{tB}$
9.6	0.32	$+0.70C_{td}^8 - 0.40C_{Qd}^8 + 0.39C_{Qu}^8 - 0.31C_{Qq}^{3,8} - 0.21C_{tB} - 0.17C_{tq}^8 - 0.13C_{Qq}^{1,8} + 0.11C_{Ht}$
13.0	0.28	$+0.75C_{Qd}^8 - 0.48C_{tq}^8 + 0.33C_{Qu}^8 + 0.18C_{Ht} + 0.16C_{td}^8 - 0.16C_{tu}^8 + 0.09C_{tB}$
21.0	0.22	$+0.69C_{tu}^8 - 0.50\overline{C_{Qq}^{3,8}} - 0.41C_{td}^8 + 0.20C_{Qu}^8 - 0.18C_{Qq}^{1,8} - 0.17C_{tq}^8 + 0.06C_{Qd}^8$

**Table 8.** Components of the eigenvectors found in the principal component analysis of the globalfit displayed in figure 13. Components with coefficients of magnitude less than 0.05 are omitted.

# A Datasets

The following tables summarise the observables that have been encoded into the Fitmaker database. Those that are not included in the final fit are greyed out. This is usually because they are not statistically independent from other data that we include.

EW precision observables	$n_{\mathbf{obs}}$	Ref.
Precision electroweak measurements on the $Z$ resonance.	12	[1]
$\Gamma_Z, \sigma_{\text{had.}}^0, R_{\ell}^0, A_{FB}^{\ell}, A_{\ell}(\text{SLD}), A_{\ell}(\text{Pt}), R_b^0, R_c^0, A_{FB}^b, A_{FB}^c, A_b \&$		
$A_c$		
Combination of CDF and D0 W-Boson Mass Measurements	1	[6]
LHC run 1 W boson mass measurement by ATLAS	1	[65]
Diboson LEP & LHC	$n_{\mathbf{obs}}$	Ref.
$W^+W^-$ angular distribution measurements at LEP II.	8	[5]
$W^+W^-$ total cross section measurements at L3 in the $\ell\nu\ell\nu$ , $\ell\nu qq$	24	[3]
& $qqqq$ final states for 8 energies		
$W^+W^-$ total cross section measurements at OPAL in the $\ell\nu\ell\nu$ ,	21	[4]
$\ell \nu q q \& q q q q$ final states for 7 energies		
$W^+W^-$ total cross section measurements at ALEPH in the $\ell\nu\ell\nu$ ,	21	[2]
$\ell \nu q q \& q q q q$ final states for 8 energies		
ATLAS $W^+ W^-$ differential cross section in the $e\nu\mu\nu$ channel,	1	[66]
$\frac{d\sigma}{dp_{\ell_1}^T}, p_T > 120 \text{GeV}$ overflow bin		
ATLAS $W^+W^-$ fiducial differential cross section in the $e\nu\mu\nu$	14	[70]
channel, $\frac{d\sigma}{dp_{\ell_1}^T}$		
ATLAS $W^{\pm} Z$ fiducial differential cross section in the $\ell^+ \ell^- \ell^{\pm} \nu$	7	[69]
channel, $\frac{d\sigma}{dp_Z^T}$		
CMS $W^{\pm} Z$ normalised fiducial differential cross section in the	11	[67]
$\ell^+ \ell^- \ell^\pm \nu$ channel, $\frac{1}{\sigma} \frac{d\sigma}{dp_Z^T}$		
ATLAS $Zjj$ fiducial differential cross section in the $\ell^+\ell^-$ channel,	12	[71]
$\frac{d\sigma}{d\Delta \varphi_{ij}}$		

LHC Run 1 Higgs	$n_{\mathbf{obs}}$	Ref.
ATLAS and CMS LHC Run 1 combination of Higgs signal	21	[9]
strengths.		
Production: $ggF$ , $VBF$ , $ZH$ , $WH$ & $ttH$		
Decay: $\gamma\gamma$ , $ZZ$ , $W^+W^-$ , $\tau^+\tau^-$ & $b\bar{b}$		
ATLAS inclusive $Z\gamma$ signal strength measurement	1	[8]

LHC Run 2 Higgs (new)	$n_{\mathbf{obs}}$	Ref.
ATLAS combination of signal strengths and stage 1.0 STXS in	16 19  25	[12]
$H \to 4\ell$ including ratios of branching fractions to $\gamma\gamma$ , $WW^*$ , $\tau^+\tau^-$		
$\& b\bar{b}$		
Signal strengths   coarse STXS bins   fine STXS bins		
CMS LHC combination of Higgs signal strengths.	23	[15]
Production: $ggF$ , $VBF$ , $ZH$ , $WH$ & $ttH$		
Decay: $\gamma\gamma$ , ZZ, $W^+W^-$ , $\tau^+\tau^-$ , $b\bar{b}$ & $\mu^+\mu^-$		
CMS stage 1.0 STXS measurements for $H \to \gamma \gamma$ .	13 7	[14]
13 parameter fit $\mid$ 7 parameter fit		
CMS stage 1.0 STXS measurements for $H \to \tau^+ \tau^-$	9	[13]
CMS stage 1.1 STXS measurements for $H \to 4\ell$	19	[10]
CMS differential cross section measurements of inclusive Higgs	5 6	[11]
production in the $WW^* \to \ell \nu \ell \nu$ final state.		
$\left  rac{d\sigma}{dn_{ m jet}}  ight  \left  rac{d\sigma}{dp_{H}^{T}}  ight $		
ATLAS $H \to Z\overline{\gamma}$ signal strength.	1	[16]
ATLAS $H \to \mu^+ \mu^-$ signal strength.	1	[17]

Tevatron & Run 1 top	$n_{\mathbf{obs}}$	Ref.
Tevatron combination of differential $t\bar{t}$ forward-backward asym-	4	[7]
metry, $A_{FB}(m_{t\bar{t}})$ .		
ATLAS $t\bar{t}$ differential distributions in the dilepton channel.	6	[31]
$\frac{d\sigma}{dm_{t\bar{t}}}$		
ATLAS $t\bar{t}$ differential distributions in the $\ell$ +jets channel.	7 5  8  5	[24]
$rac{d\sigma}{dm_{tar{t}}} ~~ ~~ rac{d\sigma}{d y_{tar{t}} }~~ ~~ rac{d\sigma}{dp_t^T}~~ ~~ rac{d\sigma}{d y_t }~.$		
CMS $t\bar{t}$ differential distributions in the $\ell$ +jets channel.	7 10  8	[25.
$\frac{d\sigma}{d\sigma} \mid \frac{d\sigma}{d\sigma} \mid \frac{d\sigma}{\sigma} \mid \frac{d\sigma}{\sigma}$	110	<b>3</b> 4]
$\frac{dm_{t\bar{t}}}{dt} = \frac{dy_{t\bar{t}}}{dt} = \frac{dy_{t}}{dt} = \frac{dy_{t}}{dt}$		[00]
CMS measurement of differential tt charge asymmetry, $A_C(m_{t\bar{t}})$	3	[33]
in the dilepton channel.		[00]
ATLAS inclusive measurement to charge asymmetry, $A_C(m_{t\bar{t}})$ in	1	[32]
the dilepton channel.	C	[20]
AT LAS & UNIS combination of differential to charge asymmetry, $A_{\pi}(m_{\pi})$ in the $\ell$ jets shapped	0	႞ၖၓႄ႞
$A_C(m_{t\bar{t}})$ , in the $t$ +jets channel.	10 10	[10
CMS tt double differential distributions in the dilepton channel. $dz = dz = dz$	16  16	[18,
$\left  rac{dv}{dm_{tar{t}}dy_t}  ight  = \left  rac{dv}{dm_{tar{t}}dy_{tar{t}}}  ight  = \left  rac{dv}{dm_{tar{t}}dp_{tar{t}}^T}  ight  = \left  rac{dv}{dy_t dp_t^T}  ight .$	16 16	35]
ATLAS & CMS Run 1 combination of W-boson helicity fractions	3	[40]
in top decay. $f_0, f_L \& f_R$		
ATLAS measurement of $W$ -boson helicity fractions in top decay.	3	[30]
$f_0, f_L \& f_R$		
CMS measurement of $W$ -boson helicity fractions in top decay.	3	[29]
$f_0, f_L \& f_R$		
ATLAS $t\bar{t}W$ & $t\bar{t}Z$ cross section measurements. $\sigma_{t\bar{t}W} \sigma_{t\bar{t}Z}$	2	[23]
CMS $t\bar{t}W$ & $t\bar{t}Z$ cross section measurements. $\sigma_{t\bar{t}W} \sigma_{t\bar{t}Z}$	2	[26]
ATLAS $t\bar{t}\gamma$ cross section measurement in the $\ell$ + jets channel.	1	[36]
CMS $t\bar{t}\gamma$ cross section measurement in the $\ell$ + jets channel.	1	[37]
ATLAS $t$ -channel single-top differential distributions.	4  4 4  5	[39]
$\left  rac{d\sigma}{dp_t^T} ~~ \left  ~~ rac{d\sigma}{dp_{ar t}^T} ~~ \left  ~~ rac{d\sigma}{d y_t } ~~  ight  ~~ rac{d\sigma}{d y_{ar t} }  ight $		
CMS <i>s</i> -channel single-top cross section measurement.	1	[28]
CMS <i>t</i> -channel single-top differential distributions.	6 6	[19]
$\frac{d\sigma}{dp_{T+\bar{t}}^T} \mid \frac{d\sigma}{d y_{t+\bar{t}} }$		
CMS measurement of the <i>t</i> -channel single-top and anti-top cross	1 1 1  1	[20]
sections. $\sigma_t   \sigma_{\bar{t}}   \sigma_{t+\bar{t}}   R_t.$		
ATLAS <i>s</i> -channel single-top cross section measurement.	1	[27]
CMS $tW$ cross section measurement.	1	[21]
ATLAS $tW$ cross section measurement in the single lepton chan-	1	[41]
nel.		
ATLAS $tW$ cross section measurement in the dilepton channel.	1	[22]

Run 2 top	$n_{\mathbf{obs}}$	Ref.
CMS $t\bar{t}$ differential distributions in the dilepton channel.	6	[46,
$\frac{d\sigma}{dm_{\star\bar{\tau}}}$		50]
$\overline{\text{CMS}} t\bar{t}$ differential distributions in the $\ell$ +jets channel.	10	[53]
$\frac{d\sigma}{dm_{\star\bar{\tau}}}$		
ATLAS measurement of differential tt charge asymmetry,	5	[55]
$A_C(m_{tar{t}}).$		
ATLAS $t\bar{t}W$ & $t\bar{t}Z$ cross section measurements. $\sigma_{t\bar{t}W} \sigma_{t\bar{t}Z}$	2	[58]
CMS $t\bar{t}W$ & $t\bar{t}Z$ cross section measurements. $\sigma_{t\bar{t}W} \mid \sigma_{t\bar{t}Z}$	1   1	[48]
CMS $t\bar{t}Z$ differential distributions.	4 4	[60]
$\frac{d\sigma}{dp_{r}^{T}} \mid \frac{d\sigma}{d\cos\theta^{*}}$		
ATLAS $t\bar{t}\gamma$ differential distribution.	11	[62]
$\frac{d\sigma}{dn^T}$		
$\frac{\alpha_{P\gamma}}{CMS}$ measurement of differential cross sections and charge ratios	5 5	[56]
for <i>t</i> -channel single-top quark production.	I.	
$\left  rac{d\sigma}{dp_{t+ au}^T}  ight  = R_t \left( p_{t+ au}^T  ight)$		
CMS measurement of t-channel single-top and anti-top cross sec-	4	[42]
tions.		
$\sigma_t,  \sigma_{\bar{t}},  \sigma_{t+\bar{t}} \& R_t.$		
CMS measurement of the $t$ -channel single-top and anti-top cross	$1 1  \ 1 1$	[45]
sections. $\sigma_t   \sigma_{\bar{t}}   \sigma_{t+\bar{t}}   R_t$ .		
CMS $t$ -channel single-top differential distributions.	4 4	[44]
$\left  rac{d\sigma}{dp_{t\perpar{t}}^T}  ight  \left  rac{d\sigma}{d y_{t+ar{t}} }  ight $		
$\operatorname{ATLAS} tW$ cross section measurement.	1	[43]
CMS $tZ$ cross section measurement.	1	[47]
CMS $tW$ cross section measurement.	1	[52]
ATLAS $tZ$ cross section measurement.	1	[49]
CMS $tZ(Z \to \ell^+ \ell^-)$ cross section measurement	1	[54]
ATLAS four-top search in the multi-lepton and same-sign dilepton	1	[63]
channels.		
ATLAS four-top search in the single-lepton and opposite-sign	1	[51]
dilepton channels.		
CMS four-top search in the multi-lepton and same-sign dilepton	1	[61]
channels.		
CMS four-top search in the single-lepton and opposite-sign dilep-	1	[59]
ton channels.		
CMS $ttbb$ cross section measurement in the all-jet channel.	1	[57]
CMS <i>ttbb</i> cross section measurement in the dilepton channel.	1	64

### **B** Numerical fits with nested sampling

Throughout this paper we make use of the fitting procedure outlined in section 5, in which we minimise a  $\chi^2$  function and determine the least-squares estimator for each coefficient  $\hat{\vec{C}}$ . This procedure has the advantage of being analytic, as it is linearised in the coefficients  $\vec{C}$ , and therefore fast to implement. However, it relies on a number of key assumptions: the linear approximation to the SMEFT predictions  $\mu(\vec{C})$  must be accurate, and the experimental covariance matrix V must be symmetrised. Even if these assumptions are satisfied and we can write down a Gaussian likelihood  $\mathcal{L}(\vec{C}|D)$  where D denotes the dataset, the analytic fitting procedure excludes the possibility that a nontrivial prior  $\pi(\vec{C})$  may lead to a non-Gaussian posterior  $p(\vec{C}|D)$  through Bayes' theorem:  $p(\vec{C}|D) \propto \mathcal{L}(\vec{C}|D)\pi(\vec{C})$ .

With this motivation, we implement in Fitmaker an option to constrain parameters numerically using the method of nested sampling. This is provided through MultiNest [239], incorporated into our code using Pymultinest [240].

As in traditional Markov Chain Monte Carlo (MCMC) parameter estimation, nested sampling is a method of sampling from the posterior distribution. MultiNest uses the ellipsoidal nested sampling algorithm, described in more detail in [239], in which the samples are drawn from ellipsoids in parameter space. As the algorithm progresses the ellipsoids close in on the regions of high likelihood. Overlapping and distinct ellipsoids allow for the possibility of degeneracies and multiple modes in the posterior respectively. This is an advantage over traditional MCMC methods, in which the Markov chain may get stuck and fail to explore more than one mode of the posterior. The importance of this feature in the context of quadratic SMEFT contributions will be seen below.

We apply nested sampling in section 7 to find constraints on the 1-parameter UV models in table 3. In many of these models we are constraining a positive quantity  $|\lambda|^2$ , where  $\lambda$  denotes a coupling of the new field to the SM. Nested sampling allows us to produce the constraints due to positivity bounds on  $|\lambda|^2$  using a Heaviside prior:  $\pi(|\lambda|^2 < 0) = 0$ . The constraints in figure 14 are found in this way.

As a proof-of-concept of the capabilities of nested sampling, we investigate in this section the effects of including quadratic contributions from dimension-6 operators in SMEFT predictions from Higgs data. We perform a Higgs-only fit using just the Run 2 signal strength measurements from ATLAS [12] and CMS [15], and constrain 7 operators:  $C_{H\square}$ ,  $C_{HG}$ ,  $C_{HW}$ ,  $C_{HB}$ ,  $C_{\tau H}$ ,  $C_{tH}$  and  $C_{bH}$ . For the purpose of this proof-of-concept fit, we take our SMEFT predictions from [241], rotating the SILH basis operators into the Warsaw basis using the Rosetta code [242]. This fit differs from the results in the main text: as well as the differences in the SMEFT predictions, we use just a subset of the full dataset and use only signal strengths, not STXS measurements.

Figure 20 shows the distribution of 30,000 samples produced by nested sampling (with a sampling efficiency of 0.8 and an evidence tolerance of 0.5, taking approximately 1 hour). Each 2-dimensional distribution is a projection of the full 7-dimensional posterior distribution onto a 2-parameter subspace. These plots highlight the non-Gaussianity of the posterior distributions when quadratic contributions are included. In particular, we see multiple modes in the distribution of  $C_{\tau H}$  as well as highly skewed distributions in  $C_{tH}$  and  $C_{bH}$ .



Figure 20. Samples produced by nested sampling, projected onto 2- and 1-dimensional subspaces of the 7-dimensional parameter space. Along the diagonal we compare the distributions found with and without quadratic SMEFT contributions in blue and purple, respectively.

The histograms along the diagonal in figure 20 show the distributions of samples in each of the 7 parameters. Here we compare the results of nested sampling with and without quadratic contributions in blue and purple respectively. We see that although the distributions are generally peaked close to the same value, the shapes of the distributions differ, with the quadratic contributions leading to more asymmetric and multimodal distributions. For example, the distribution of  $C_{HG}$  is more skewed towards the negative region when quadratic corrections are included, and  $C_{bH}$  is much more narrowly constrained than in the linear-only case. These differences in distributions translate into differences in the marginalised 95 % credible intervals shown in figure 21, computed as highest posterior density intervals.



Figure 21. Marginalised 95 % credible intervals for each of the indicated operator coefficients resulting from a fit to Higgs signal strength data using nested sampling. We compare the effects of including quadratic contributions from dimension-6 operators (blue) to the case of linearised SMEFT predictions (purple).

Although there are visible differences between the credible regions found in the linear and quadratic SMEFT fits, they are sufficiently similar that one may consider the linear approximation to be usefully robust. The most notable difference is in  $C_{\tau H}$ , in which a distinct second mode is found in the quadratic SMEFT fit, while the first mode is in good agreement with the linear SMEFT fit. There is only one other instance, namely  $C_{HG}$ , where the mode in the quadratic fit lies outside the 95% CL range found in the linear approximation, and only one instance, namely  $C_{bH}$ , where the size of the quadratic 95% credible interval is much smaller than the linear 95% CL range. Apart from these exceptions, the ranges estimated in the linear fit are encouraging approximations to the results from the quadratic fit. We note that a global quadratic fit would require calculations of many currently unknown quadratic operator contributions and, for consistency, a full treatment of the linear contributions of dimension-8 operators as discussed in the Higgs sector in [208].

**Open Access.** This article is distributed under the terms of the Creative Commons Attribution License (CC-BY 4.0), which permits any use, distribution and reproduction in any medium, provided the original author(s) and source are credited.

# References

- ALEPH, DELPHI, L3, OPAL, SLD, LEP ELECTROWEAK WORKING GROUP, SLD ELECTROWEAK GROUP and SLD HEAVY FLAVOUR GROUP collaborations, *Precision* electroweak measurements on the Z resonance, *Phys. Rept.* 427 (2006) 257 [hep-ex/0509008] [INSPIRE].
- [2] ALEPH collaboration, Measurement of W-pair production in e<sup>+</sup>e<sup>-</sup> collisions at centre-of-mass energies from 183-GeV to 209-GeV, Eur. Phys. J. C 38 (2004) 147 [INSPIRE].

- [3] L3 collaboration, Measurement of the cross section of W-boson pair production at LEP, Phys. Lett. B 600 (2004) 22 [hep-ex/0409016] [INSPIRE].
- [4] OPAL collaboration, Measurement of the  $e^+e^- \rightarrow W^+W^-$  cross section and W decay branching fractions at LEP, Eur. Phys. J. C 52 (2007) 767 [arXiv:0708.1311] [INSPIRE].
- [5] ALEPH, DELPHI, L3, OPAL and LEP ELECTROWEAK collaborations, *Electroweak Measurements in Electron-Positron Collisions at W-Boson-Pair Energies at LEP, Phys. Rept.* 532 (2013) 119 [arXiv:1302.3415] [INSPIRE].
- [6] CDF and D0 collaborations, Combination of CDF and D0 W-Boson Mass Measurements, Phys. Rev. D 88 (2013) 052018 [arXiv:1307.7627] [INSPIRE].
- [7] CDF and D0 collaborations, Combined Forward-Backward Asymmetry Measurements in Top-Antitop Quark Production at the Tevatron, Phys. Rev. Lett. 120 (2018) 042001
   [arXiv:1709.04894] [INSPIRE].
- [8] ATLAS collaboration, Measurements of the Higgs boson production and decay rates and coupling strengths using pp collision data at  $\sqrt{s} = 7$  and 8 TeV in the ATLAS experiment, Eur. Phys. J. C 76 (2016) 6 [arXiv:1507.04548] [INSPIRE].
- [9] ATLAS and CMS collaborations, Measurements of the Higgs boson production and decay rates and constraints on its couplings from a combined ATLAS and CMS analysis of the LHC pp collision data at  $\sqrt{s} = 7$  and 8 TeV, JHEP **08** (2016) 045 [arXiv:1606.02266] [INSPIRE].
- [10] CMS collaboration, Measurements of properties of the Higgs boson in the four-lepton final state in proton-proton collisions at  $\sqrt{s} = 13 \text{ TeV}$ , Tech. Rep. CMS-PAS-HIG-19-001 (2019).
- [11] CMS collaboration, Measurements of differential Higgs boson production cross sections in the leptonic WW decay mode at  $\sqrt{s} = 13$  TeV, Tech. Rep. CMS-PAS-HIG-19-002 (2019).
- [12] ATLAS collaboration, Combined measurements of Higgs boson production and decay using up to 80 fb<sup>-1</sup> of proton-proton collision data at  $\sqrt{s} = 13$  TeV collected with the ATLAS experiment, Phys. Rev. D 101 (2020) 012002 [arXiv:1909.02845] [INSPIRE].
- [13] CMS collaboration, Measurement of Higgs boson production and decay to the  $\tau\tau$  final state, Tech. Rep. CMS-PAS-HIG-18-032 (2019).
- [14] CMS collaboration, Measurements of Higgs boson production via gluon fusion and vector boson fusion in the diphoton decay channel at  $\sqrt{s} = 13$  TeV, Tech. Rep. CMS-PAS-HIG-18-029 (2019).
- [15] CMS collaboration, Combined Higgs boson production and decay measurements with up to 137fb-1 of proton-proton collision data at sqrts = 13 TeV, Tech. Rep. CMS-PAS-HIG-19-005 (2020).
- [16] ATLAS collaboration, A search for the  $Z\gamma$  decay mode of the Higgs boson in pp collisions at  $\sqrt{s} = 13$  TeV with the ATLAS detector, Phys. Lett. B 809 (2020) 135754 [arXiv:2005.05382] [INSPIRE].
- [17] ATLAS collaboration, A search for the dimuon decay of the Standard Model Higgs boson with the ATLAS detector, Phys. Lett. B 812 (2021) 135980 [arXiv:2007.07830] [INSPIRE].
- [18] CMS collaboration, Measurement of the  $t\bar{t}$  production cross section in the dilepton channel in pp collisions at  $\sqrt{s} = 8$  TeV, JHEP **02** (2014) 024 [Erratum ibid. **02** (2014) 102] [arXiv:1312.7582] [INSPIRE].
- [19] CMS collaboration, Single top t-channel differential cross section at 8 TeV, Tech. Rep. CMS-PAS-TOP-14-004 (2014).

- [20] CMS collaboration, Measurement of the t-channel single-top-quark production cross section and of the  $|V_{tb}|$  CKM matrix element in pp collisions at  $\sqrt{s} = 8$  TeV, JHEP **06** (2014) 090 [arXiv:1403.7366] [INSPIRE].
- [21] CMS collaboration, Observation of the associated production of a single top quark and a W boson in pp collisions at  $\sqrt{s} = 8$  TeV, Phys. Rev. Lett. **112** (2014) 231802 [arXiv:1401.2942] [INSPIRE].
- [22] ATLAS collaboration, Measurement of the production cross-section of a single top quark in association with a W boson at 8 TeV with the ATLAS experiment, JHEP 01 (2016) 064 [arXiv:1510.03752] [INSPIRE].
- [23] ATLAS collaboration, Measurement of the  $t\bar{t}W$  and  $t\bar{t}Z$  production cross sections in pp collisions at  $\sqrt{s} = 8 \text{ TeV}$  with the ATLAS detector, JHEP **11** (2015) 172 [arXiv:1509.05276] [INSPIRE].
- [24] ATLAS collaboration, Measurements of top-quark pair differential cross-sections in the lepton+jets channel in pp collisions at  $\sqrt{s} = 8$  TeV using the ATLAS detector, Eur. Phys. J. C 76 (2016) 538 [arXiv:1511.04716] [INSPIRE].
- [25] CMS collaboration, Measurement of the differential cross section for top quark pair production in pp collisions at  $\sqrt{s} = 8$  TeV, Eur. Phys. J. C **75** (2015) 542 [arXiv:1505.04480] [INSPIRE].
- [26] CMS collaboration, Observation of top quark pairs produced in association with a vector boson in pp collisions at  $\sqrt{s} = 8$  TeV, JHEP **01** (2016) 096 [arXiv:1510.01131] [INSPIRE].
- [27] ATLAS collaboration, Evidence for single top-quark production in the s-channel in proton-proton collisions at  $\sqrt{s} = 8$  TeV with the ATLAS detector using the Matrix Element Method, Phys. Lett. B **756** (2016) 228 [arXiv:1511.05980] [INSPIRE].
- [28] CMS collaboration, Search for s channel single top quark production in pp collisions at  $\sqrt{s} = 7$  and 8 TeV, JHEP 09 (2016) 027 [arXiv:1603.02555] [INSPIRE].
- [29] CMS collaboration, Measurement of the W boson helicity fractions in the decays of top quark pairs to lepton + jets final states produced in pp collisions at  $\sqrt{s} = 8 \text{ TeV}$ , Phys. Lett. B 762 (2016) 512 [arXiv:1605.09047] [INSPIRE].
- [30] ATLAS collaboration, Measurement of the W boson polarisation in  $t\bar{t}$  events from pp collisions at  $\sqrt{s} = 8$  TeV in the lepton + jets channel with ATLAS, Eur. Phys. J. C 77 (2017) 264 [Erratum ibid. 79 (2019) 19] [arXiv:1612.02577] [INSPIRE].
- [31] ATLAS collaboration, Measurement of top quark pair differential cross-sections in the dilepton channel in pp collisions at  $\sqrt{s} = 7$  and 8 TeV with ATLAS, Phys. Rev. D 94 (2016) 092003 [Addendum ibid. 101 (2020) 119901] [arXiv:1607.07281] [INSPIRE].
- [32] ATLAS collaboration, Measurements of the charge asymmetry in top-quark pair production in the dilepton final state at  $\sqrt{s} = 8$  TeV with the ATLAS detector, Phys. Rev. D 94 (2016) 032006 [arXiv:1604.05538] [INSPIRE].
- [33] CMS collaboration, Measurements of  $t\bar{t}$  charge asymmetry using dilepton final states in pp collisions at  $\sqrt{s} = 8$  TeV, Phys. Lett. B 760 (2016) 365 [arXiv:1603.06221] [INSPIRE].
- [34] CMS collaboration, Measurements of the tt production cross section in lepton+jets final states in pp collisions at 8 TeV and ratio of 8 to 7 TeV cross sections, Eur. Phys. J. C 77 (2017) 15 [arXiv:1602.09024] [INSPIRE].

- [35] CMS collaboration, Measurement of double-differential cross sections for top quark pair production in pp collisions at  $\sqrt{s} = 8$  TeV and impact on parton distribution functions, Eur. Phys. J. C 77 (2017) 459 [arXiv:1703.01630] [INSPIRE].
- [36] ATLAS collaboration, Measurement of the  $t\bar{t}\gamma$  production cross section in proton-proton collisions at  $\sqrt{s} = 8$  TeV with the ATLAS detector, JHEP **11** (2017) 086 [arXiv:1706.03046] [INSPIRE].
- [37] CMS collaboration, Measurement of the semileptonic  $t\bar{t} + \gamma$  production cross section in pp collisions at  $\sqrt{s} = 8$  TeV, JHEP 10 (2017) 006 [arXiv:1706.08128] [INSPIRE].
- [38] ATLAS and CMS collaborations, Combination of inclusive and differential  $t\bar{t}$  charge asymmetry measurements using ATLAS and CMS data at  $\sqrt{s} = 7$  and 8 TeV, JHEP 04 (2018) 033 [arXiv:1709.05327] [INSPIRE].
- [39] ATLAS collaboration, Fiducial, total and differential cross-section measurements of t-channel single top-quark production in pp collisions at 8 TeV using data collected by the ATLAS detector, Eur. Phys. J. C 77 (2017) 531 [arXiv:1702.02859] [INSPIRE].
- [40] CMS and ATLAS collaborations, Combination of the W boson polarization measurements in top quark decays using ATLAS and CMS data at  $\sqrt{s} = 8$  TeV, JHEP **08** (2020) 051 [arXiv:2005.03799] [INSPIRE].
- [41] ATLAS collaboration, Measurement of single top-quark production in association with a W boson in the single-lepton channel at  $\sqrt{s} = 8$  TeV with the ATLAS detector, arXiv:2007.01554 [INSPIRE].
- [42] CMS collaboration, Cross section measurement of t-channel single top quark production in pp collisions at  $\sqrt{s} = 13 \text{ TeV}$ , Phys. Lett. B 772 (2017) 752 [arXiv:1610.00678] [INSPIRE].
- [43] ATLAS collaboration, Measurement of the cross-section for producing a W boson in association with a single top quark in pp collisions at  $\sqrt{s} = 13$  TeV with ATLAS, JHEP **01** (2018) 063 [arXiv:1612.07231] [INSPIRE].
- [44] CMS collaboration, Measurement of the differential cross section for t-channel single-top-quark production at  $\sqrt{s} = 13$  TeV, Tech. Rep. CMS-PAS-TOP-16-004 (2016).
- [45] ATLAS collaboration, Measurement of the inclusive cross-sections of single top-quark and top-antiquark t-channel production in pp collisions at  $\sqrt{s} = 13$  TeV with the ATLAS detector, JHEP 04 (2017) 086 [arXiv:1609.03920] [INSPIRE].
- [46] CMS collaboration, Measurement of normalized differential  $t\bar{t}$  cross sections in the dilepton channel from pp collisions at  $\sqrt{s} = 13 \text{ TeV}$ , JHEP **04** (2018) 060 [arXiv:1708.07638] [INSPIRE].
- [47] CMS collaboration, Measurement of the associated production of a single top quark and a Z boson in pp collisions at  $\sqrt{s} = 13$  TeV, Phys. Lett. B 779 (2018) 358 [arXiv:1712.02825] [INSPIRE].
- [48] CMS collaboration, Measurement of the cross section for top quark pair production in association with a W or Z boson in proton-proton collisions at  $\sqrt{s} = 13$  TeV, JHEP 08 (2018) 011 [arXiv:1711.02547] [INSPIRE].
- [49] ATLAS collaboration, Measurement of the production cross-section of a single top quark in association with a Z boson in proton-proton collisions at 13 TeV with the ATLAS detector, Phys. Lett. B 780 (2018) 557 [arXiv:1710.03659] [INSPIRE].

- [50] CMS collaboration, Measurement of the  $t\bar{t}$  production cross section, the top quark mass, and the strong coupling constant using dilepton events in pp collisions at  $\sqrt{s} = 13$  TeV, Eur. Phys. J. C 79 (2019) 368 [arXiv:1812.10505] [INSPIRE].
- [51] ATLAS collaboration, Search for four-top-quark production in the single-lepton and opposite-sign dilepton final states in pp collisions at  $\sqrt{s} = 13$  TeV with the ATLAS detector, *Phys. Rev. D* **99** (2019) 052009 [arXiv:1811.02305] [INSPIRE].
- [52] CMS collaboration, Measurement of the production cross section for single top quarks in association with W bosons in proton-proton collisions at  $\sqrt{s} = 13$  TeV, JHEP 10 (2018) 117 [arXiv:1805.07399] [INSPIRE].
- [53] CMS collaboration, Measurement of differential cross sections for the production of top quark pairs and of additional jets in lepton+jets events from pp collisions at  $\sqrt{s} = 13$  TeV, Phys. Rev. D 97 (2018) 112003 [arXiv:1803.08856] [INSPIRE].
- [54] CMS collaboration, Observation of Single Top Quark Production in Association with a Z Boson in Proton-Proton Collisions at  $\sqrt{s} = 13$  TeV, Phys. Rev. Lett. **122** (2019) 132003 [arXiv:1812.05900] [INSPIRE].
- [55] ATLAS collaboration, Inclusive and differential measurement of the charge asymmetry in  $t\bar{t}$  events at 13 TeV with the ATLAS detector, Tech. Rep. ATLAS-CONF-2019-026 (2019).
- [56] CMS collaboration, Measurement of differential cross sections and charge ratios for t-channel single top quark production in proton-proton collisions at  $\sqrt{s} = 13 \text{ TeV}$ , Eur. Phys. J. C 80 (2020) 370 [arXiv:1907.08330] [INSPIRE].
- [57] CMS collaboration, Measurement of the ttbb production cross section in the all-jet final state in pp collisions at  $\sqrt{s} = 13$  TeV, Phys. Lett. B 803 (2020) 135285 [arXiv:1909.05306] [INSPIRE].
- [58] ATLAS collaboration, Measurement of the  $t\bar{t}Z$  and  $t\bar{t}W$  cross sections in proton-proton collisions at  $\sqrt{s} = 13$  TeV with the ATLAS detector, Phys. Rev. D **99** (2019) 072009 [arXiv:1901.03584] [INSPIRE].
- [59] CMS collaboration, Search for the production of four top quarks in the single-lepton and opposite-sign dilepton final states in proton-proton collisions at  $\sqrt{s} = 13$  TeV, JHEP 11 (2019) 082 [arXiv:1906.02805] [INSPIRE].
- [60] CMS collaboration, Measurement of top quark pair production in association with a Z boson in proton-proton collisions at √s = 13 TeV, JHEP 03 (2020) 056
   [arXiv:1907.11270] [INSPIRE].
- [61] CMS collaboration, Search for production of four top quarks in final states with same-sign or multiple leptons in proton-proton collisions at  $\sqrt{s} = 13$  TeV, Eur. Phys. J. C 80 (2020) 75 [arXiv:1908.06463] [INSPIRE].
- [62] ATLAS collaboration, Measurements of inclusive and differential cross-sections of combined  $t\bar{t}\gamma$  and  $tW\gamma$  production in the eµ channel at 13 TeV with the ATLAS detector, JHEP 09 (2020) 049 [arXiv:2007.06946] [INSPIRE].
- [63] ATLAS collaboration, Evidence for  $t\bar{t}t\bar{t}$  production in the multilepton final state in proton-proton collisions at  $\sqrt{s} = 13$  TeV with the ATLAS detector, Tech. Rep. ATLAS-CONF-2020-013 (2020).
- [64] CMS collaboration, Measurement of the cross section for tt production with additional jets and b jets in pp collisions at  $\sqrt{s} = 13$  TeV, JHEP **07** (2020) 125 [arXiv:2003.06467] [INSPIRE].

- [65] ATLAS collaboration, Measurement of the W-boson mass in pp collisions at  $\sqrt{s} = 7$  TeV with the ATLAS detector, Eur. Phys. J. C 78 (2018) 110 [Erratum ibid. 78 (2018) 898] [arXiv:1701.07240] [INSPIRE].
- [66] ATLAS collaboration, Measurement of the  $W^+W^-$  production cross section in pp collisions at a centre-of-mass energy of  $\sqrt{s} = 13$  TeV with the ATLAS experiment, Phys. Lett. B 773 (2017) 354 [arXiv:1702.04519] [INSPIRE].
- [67] CMS collaboration, Measurements of the  $pp \rightarrow WZ$  inclusive and differential production cross section and constraints on charged anomalous triple gauge couplings at  $\sqrt{s} = 13$  TeV, JHEP 04 (2019) 122 [arXiv:1901.03428] [INSPIRE].
- [68] CMS collaboration, Search for anomalous triple gauge couplings in WW and WZ production in lepton + jet events in proton-proton collisions at  $\sqrt{s} = 13 \text{ TeV}$ , JHEP 12 (2019) 062 [arXiv:1907.08354] [INSPIRE].
- [69] ATLAS collaboration, Measurement of  $W^{\pm}Z$  production cross sections and gauge boson polarisation in pp collisions at  $\sqrt{s} = 13$  TeV with the ATLAS detector, Eur. Phys. J. C 79 (2019) 535 [arXiv:1902.05759] [INSPIRE].
- [70] ATLAS collaboration, Measurement of fiducial and differential  $W^+W^-$  production cross-sections at  $\sqrt{s} = 13$  TeV with the ATLAS detector, Eur. Phys. J. C 79 (2019) 884 [arXiv:1905.04242] [INSPIRE].
- [71] ATLAS collaboration, Differential cross-section measurements for the electroweak production of dijets in association with a Z boson in proton-proton collisions at ATLAS, Eur. Phys. J. C 81 (2021) 163 [arXiv:2006.15458] [INSPIRE].
- [72] T. Appelquist and J. Carazzone, Infrared Singularities and Massive Fields, Phys. Rev. D 11 (1975) 2856 [INSPIRE].
- [73] S. Weinberg, Baryon and Lepton Nonconserving Processes, Phys. Rev. Lett. 43 (1979) 1566 [INSPIRE].
- [74] W. Buchmüller and D. Wyler, Effective Lagrangian Analysis of New Interactions and Flavor Conservation, Nucl. Phys. B 268 (1986) 621 [INSPIRE].
- [75] T. Cohen, N. Craig, X. Lu and D. Sutherland, Is SMEFT Enough?, JHEP 03 (2021) 237
   [arXiv:2008.08597] [INSPIRE].
- [76] G. Buchalla, O. Catà, A. Celis and C. Krause, Fitting Higgs Data with Nonlinear Effective Theory, Eur. Phys. J. C 76 (2016) 233 [arXiv:1511.00988] [INSPIRE].
- [77] I. Brivio, J. Gonzalez-Fraile, M.C. Gonzalez-Garcia and L. Merlo, The complete HEFT Lagrangian after the LHC Run I, Eur. Phys. J. C 76 (2016) 416 [arXiv:1604.06801] [INSPIRE].
- [78] J. de Blas, O. Eberhardt and C. Krause, Current and Future Constraints on Higgs Couplings in the Nonlinear Effective Theory, JHEP 07 (2018) 048 [arXiv:1803.00939]
   [INSPIRE].
- [79] R. Alonso, E.E. Jenkins and A.V. Manohar, A Geometric Formulation of Higgs Effective Field Theory: Measuring the Curvature of Scalar Field Space, Phys. Lett. B 754 (2016) 335 [arXiv:1511.00724] [INSPIRE].
- [80] A. Helset, A. Martin and M. Trott, The Geometric Standard Model Effective Field Theory, JHEP 03 (2020) 163 [arXiv:2001.01453] [INSPIRE].
- [81] B. Grzadkowski, M. Iskrzynski, M. Misiak and J. Rosiek, Dimension-Six Terms in the Standard Model Lagrangian, JHEP 10 (2010) 085 [arXiv:1008.4884] [INSPIRE].

- [82] L. Lehman, Extending the Standard Model Effective Field Theory with the Complete Set of Dimension-7 Operators, Phys. Rev. D 90 (2014) 125023 [arXiv:1410.4193] [INSPIRE].
- [83] L. Lehman and A. Martin, Low-derivative operators of the Standard Model effective field theory via Hilbert series methods, JHEP 02 (2016) 081 [arXiv:1510.00372] [INSPIRE].
- [84] B. Henning, X. Lu, T. Melia and H. Murayama, 2, 84, 30, 993, 560, 15456, 11962, 261485, ...: Higher dimension operators in the SM EFT, JHEP 08 (2017) 016 [Erratum ibid. 09 (2019) 019] [arXiv:1512.03433] [INSPIRE].
- [85] B. Gripaios and D. Sutherland, DEFT: A program for operators in EFT, JHEP 01 (2019) 128 [arXiv:1807.07546] [INSPIRE].
- [86] J.C. Criado, BasisGen: automatic generation of operator bases, Eur. Phys. J. C 79 (2019) 256 [arXiv:1901.03501] [INSPIRE].
- [87] C.W. Murphy, Dimension-8 operators in the Standard Model Eective Field Theory, JHEP 10 (2020) 174 [arXiv:2005.00059] [INSPIRE].
- [88] H.-L. Li, Z. Ren, J. Shu, M.-L. Xiao, J.-H. Yu and Y.-H. Zheng, Complete Set of Dimension-8 Operators in the Standard Model Effective Field Theory, arXiv: 2005.00008 [INSPIRE].
- [89] H.-L. Li, Z. Ren, M.-L. Xiao, J.-H. Yu and Y.-H. Zheng, Complete Set of Dimension-9 Operators in the Standard Model Effective Field Theory, arXiv: 2007.07899 [INSPIRE].
- [90] Y. Liao and X.-D. Ma, An explicit construction of the dimension-9 operator basis in the standard model effective field theory, JHEP 11 (2020) 152 [arXiv:2007.08125] [INSPIRE].
- [91] E.E. Jenkins, A.V. Manohar and M. Trott, Renormalization Group Evolution of the Standard Model Dimension Six Operators I: Formalism and lambda Dependence, JHEP 10 (2013) 087 [arXiv:1308.2627] [INSPIRE].
- [92] E.E. Jenkins, A.V. Manohar and M. Trott, Renormalization Group Evolution of the Standard Model Dimension Six Operators II: Yukawa Dependence, JHEP 01 (2014) 035
   [arXiv:1310.4838] [INSPIRE].
- [93] R. Alonso, E.E. Jenkins, A.V. Manohar and M. Trott, Renormalization Group Evolution of the Standard Model Dimension Six Operators III: Gauge Coupling Dependence and Phenomenology, JHEP 04 (2014) 159 [arXiv:1312.2014] [INSPIRE].
- [94] Z. Han and W. Skiba, Effective theory analysis of precision electroweak data, Phys. Rev. D 71 (2005) 075009 [hep-ph/0412166] [INSPIRE].
- [95] A. Pomarol and F. Riva, Towards the Ultimate SM Fit to Close in on Higgs Physics, JHEP 01 (2014) 151 [arXiv:1308.2803] [INSPIRE].
- [96] T. Corbett, O.J.P. Eboli, J. Gonzalez-Fraile and M.C. Gonzalez-Garcia, Robust Determination of the Higgs Couplings: Power to the Data, Phys. Rev. D 87 (2013) 015022 [arXiv:1211.4580] [INSPIRE].
- [97] J. Ellis, V. Sanz and T. You, The Effective Standard Model after LHC Run I, JHEP 03 (2015) 157 [arXiv:1410.7703] [INSPIRE].
- [98] ATLAS collaboration, Observation of a new particle in the search for the Standard Model Higgs boson with the ATLAS detector at the LHC, Phys. Lett. B 716 (2012) 1 [arXiv:1207.7214] [INSPIRE].
- [99] CMS collaboration, Observation of a New Boson at a Mass of 125 GeV with the CMS Experiment at the LHC, Phys. Lett. B 716 (2012) 30 [arXiv:1207.7235] [INSPIRE].

- [100] B. Dumont, S. Fichet and G. von Gersdorff, A Bayesian view of the Higgs sector with higher dimensional operators, JHEP 07 (2013) 065 [arXiv:1304.3369] [INSPIRE].
- [101] T. Corbett, O.J.P. Éboli, J. Gonzalez-Fraile and M.C. Gonzalez-Garcia, Determining Triple Gauge Boson Couplings from Higgs Data, Phys. Rev. Lett. 111 (2013) 011801 [arXiv:1304.1151] [INSPIRE].
- [102] W.-F. Chang, W.-P. Pan and F. Xu, Effective gauge-Higgs operators analysis of new physics associated with the Higgs boson, Phys. Rev. D 88 (2013) 033004 [arXiv:1303.7035]
   [INSPIRE].
- [103] J. Elias-Miro, J.R. Espinosa, E. Masso and A. Pomarol, *Higgs windows to new physics through d = 6 operators: constraints and one-loop anomalous dimensions*, *JHEP* 11 (2013) 066 [arXiv:1308.1879] [INSPIRE].
- [104] E. Boos, V. Bunichev, M. Dubinin and Y. Kurihara, Higgs boson signal at complete tree level in the SM extension by dimension-six operators, Phys. Rev. D 89 (2014) 035001
   [arXiv:1309.5410] [INSPIRE].
- [105] J. Ellis, V. Sanz and T. You, Complete Higgs Sector Constraints on Dimension-6 Operators, JHEP 07 (2014) 036 [arXiv:1404.3667] [INSPIRE].
- [106] A. Falkowski and F. Riva, Model-independent precision constraints on dimension-6 operators, JHEP 02 (2015) 039 [arXiv:1411.0669] [INSPIRE].
- [107] L. Berthier, M. Bjørn and M. Trott, Incorporating doubly resonant  $W^{\pm}$  data in a global fit of SMEFT parameters to lift flat directions, JHEP **09** (2016) 157 [arXiv:1606.06693] [INSPIRE].
- [108] S. Banerjee, R.S. Gupta, J.Y. Reiness, S. Seth and M. Spannowsky, Towards the ultimate differential SMEFT analysis, JHEP 09 (2020) 170 [arXiv:1912.07628] [INSPIRE].
- [109] A. Biekötter, R. Gomez-Ambrosio, P. Gregg, F. Krauss and M. Schönherr, Constraining SMEFT operators with associated hγ production in weak boson fusion, Phys. Lett. B 814 (2021) 136079 [arXiv:2003.06379] [INSPIRE].
- S. Banerjee, C. Englert, R.S. Gupta and M. Spannowsky, Probing Electroweak Precision Physics via boosted Higgs-strahlung at the LHC, Phys. Rev. D 98 (2018) 095012
   [arXiv:1807.01796] [INSPIRE].
- [111] A. Falkowski et al., Light quark Yukawas in triboson final states, JHEP 04 (2021) 023
   [arXiv:2011.09551] [INSPIRE].
- [112] A. Efrati, A. Falkowski and Y. Soreq, Electroweak constraints on flavorful effective theories, JHEP 07 (2015) 018 [arXiv:1503.07872] [INSPIRE].
- [113] L. Silvestrini and M. Valli, Model-independent Bounds on the Standard Model Effective Theory from Flavour Physics, Phys. Lett. B 799 (2019) 135062 [arXiv:1812.10913]
   [INSPIRE].
- [114] S. Descotes-Genon, A. Falkowski, M. Fedele, M. González-Alonso and J. Virto, *The CKM parameters in the SMEFT*, *JHEP* 05 (2019) 172 [arXiv:1812.08163] [INSPIRE].
- [115] J. Aebischer, J. Kumar, P. Stangl and D.M. Straub, A Global Likelihood for Precision Constraints and Flavour Anomalies, Eur. Phys. J. C 79 (2019) 509 [arXiv:1810.07698]
   [INSPIRE].
- [116] T. Hurth, S. Renner and W. Shepherd, Matching for FCNC effects in the flavour-symmetric SMEFT, JHEP 06 (2019) 029 [arXiv:1903.00500] [INSPIRE].

- [117] J. Aebischer, C. Bobeth, A.J. Buras and J. Kumar, *SMEFT ATLAS of*  $\Delta F = 2$ *Transitions, JHEP* **12** (2020) 187 [arXiv:2009.07276] [INSPIRE].
- [118] R. Aoude, T. Hurth, S. Renner and W. Shepherd, The impact of flavour data on global fits of the MFV SMEFT, JHEP 12 (2020) 113 [arXiv:2003.05432] [INSPIRE].
- [119] D.A. Faroughy, G. Isidori, F. Wilsch and K. Yamamoto, Flavour symmetries in the SMEFT, JHEP 08 (2020) 166 [arXiv:2005.05366] [INSPIRE].
- [120] J. Aebischer and J. Kumar, Flavour violating effects of Yukawa running in SMEFT, JHEP
   09 (2020) 187 [arXiv:2005.12283] [INSPIRE].
- [121] A. Falkowski and K. Mimouni, Model independent constraints on four-lepton operators, JHEP 02 (2016) 086 [arXiv:1511.07434] [INSPIRE].
- [122] A. Falkowski, M. González-Alonso and K. Mimouni, Compilation of low-energy constraints on 4-fermion operators in the SMEFT, JHEP 08 (2017) 123 [arXiv:1706.03783] [INSPIRE].
- [123] A. Falkowski, M. González-Alonso and O. Naviliat-Cuncic, Comprehensive analysis of beta decays within and beyond the Standard Model, JHEP 04 (2021) 126 [arXiv:2010.13797]
   [INSPIRE].
- [124] G. Panico, F. Riva and A. Wulzer, Diboson interference resurrection, Phys. Lett. B 776 (2018) 473 [arXiv:1708.07823] [INSPIRE].
- [125] J. Baglio, S. Dawson and I.M. Lewis, NLO effects in EFT fits to W<sup>+</sup>W<sup>-</sup> production at the LHC, Phys. Rev. D 99 (2019) 035029 [arXiv:1812.00214] [INSPIRE].
- [126] C. Grojean, M. Montull and M. Riembau, *Diboson at the LHC vs LEP*, *JHEP* 03 (2019)
   020 [arXiv:1810.05149] [INSPIRE].
- [127] R. Gomez-Ambrosio, Studies of Dimension-Six EFT effects in Vector Boson Scattering, Eur. Phys. J. C 79 (2019) 389 [arXiv:1809.04189] [INSPIRE].
- [128] S. Dawson and P.P. Giardino, Higgs decays to ZZ and Zγ in the standard model effective field theory: An NLO analysis, Phys. Rev. D 97 (2018) 093003 [arXiv:1801.01136]
   [INSPIRE].
- [129] P. Kozów, L. Merlo, S. Pokorski and M. Szleper, Same-sign WW Scattering in the HEFT: Discoverability vs. EFT Validity, JHEP 07 (2019) 021 [arXiv:1905.03354] [INSPIRE].
- [130] J. Baglio, S. Dawson and S. Homiller, QCD corrections in Standard Model EFT fits to WZ and WW production, Phys. Rev. D 100 (2019) 113010 [arXiv:1909.11576] [INSPIRE].
- [131] A. Azatov, D. Barducci and E. Venturini, Precision diboson measurements at hadron colliders, JHEP 04 (2019) 075 [arXiv:1901.04821] [INSPIRE].
- [132] J. Ellis, N.E. Mavromatos and T. You, Light-by-Light Scattering Constraint on Born-Infeld Theory, Phys. Rev. Lett. 118 (2017) 261802 [arXiv:1703.08450] [INSPIRE].
- [133] J. Ellis, S.-F. Ge, H.-J. He and R.-Q. Xiao, Probing the scale of new physics in the ZZγ coupling at e<sup>+</sup>e<sup>-</sup> colliders, Chin. Phys. C 44 (2020) 063106 [arXiv:1902.06631] [INSPIRE].
- [134] S. Alioli, R. Boughezal, E. Mereghetti and F. Petriello, Novel angular dependence in Drell-Yan lepton production via dimension-8 operators, Phys. Lett. B 809 (2020) 135703
   [arXiv:2003.11615] [INSPIRE].
- [135] J. Ellis, H.-J. He and R.-Q. Xiao, Probing new physics in dimension-8 neutral gauge couplings at e<sup>+</sup>e<sup>-</sup> colliders, Sci. China Phys. Mech. Astron. 64 (2021) 221062
   [arXiv:2008.04298] [INSPIRE].
- [136] B. Bellazzini and F. Riva, New phenomenological and theoretical perspective on anomalous ZZ and  $Z\gamma$  processes, Phys. Rev. D 98 (2018) 095021 [arXiv:1806.09640] [INSPIRE].
- [137] B. Bellazzini, F. Riva, J. Serra and F. Sgarlata, The other effective fermion compositeness, JHEP 11 (2017) 020 [arXiv:1706.03070] [INSPIRE].
- [138] G.N. Remmen and N.L. Rodd, Consistency of the Standard Model Effective Field Theory, JHEP 12 (2019) 032 [arXiv:1908.09845] [INSPIRE].
- [139] B. Fuks, Y. Liu, C. Zhang and S.-Y. Zhou, Positivity in electron-positron scattering: testing the axiomatic quantum field theory principles and probing the existence of UV states, Chin. Phys. C 45 (2021) 023108 [arXiv:2009.02212] [INSPIRE].
- [140] Q. Bonnefoy, E. Gendy and C. Grojean, Positivity bounds on Minimal Flavor Violation, JHEP 04 (2021) 115 [arXiv:2011.12855] [INSPIRE].
- [141] G.F. Giudice, C. Grojean, A. Pomarol and R. Rattazzi, The Strongly-Interacting Light Higgs, JHEP 06 (2007) 045 [hep-ph/0703164] [INSPIRE].
- [142] S. Dawson and C.W. Murphy, Standard Model EFT and Extended Scalar Sectors, Phys. Rev. D 96 (2017) 015041 [arXiv:1704.07851] [INSPIRE].
- [143] J. de Blas, J.C. Criado, M. Pérez-Victoria and J. Santiago, Effective description of general extensions of the Standard Model: the complete tree-level dictionary, JHEP 03 (2018) 109 [arXiv:1711.10391] [INSPIRE].
- [144] Anisha, S. Das Bakshi, J. Chakrabortty and S.K. Patra, A Step Toward Model Comparison: Connecting Electroweak-Scale Observables to BSM through EFT and Bayesian Statistics, arXiv:2010.04088 [INSPIRE].
- [145] D. Marzocca et al., BSM Benchmarks for Effective Field Theories in Higgs and Electroweak Physics, arXiv: 2009.01249 [INSPIRE].
- [146] S. Dawson, S. Homiller and S.D. Lane, Putting standard model EFT fits to work, Phys. Rev. D 102 (2020) 055012 [arXiv:2007.01296] [INSPIRE].
- [147] B. Mecaj and M. Neubert, Effective Field Theory for Leptoquarks, arXiv: 2012.02186
  [INSPIRE].
- [148] B. Henning, X. Lu and H. Murayama, What do precision Higgs measurements buy us?, arXiv:1404.1058 [INSPIRE].
- [149] R. Huo, Standard Model Effective Field Theory: Integrating out Vector-Like Fermions, JHEP 09 (2015) 037 [arXiv:1506.00840] [INSPIRE].
- [150] A. Drozd, J. Ellis, J. Quevillon and T. You, Comparing EFT and Exact One-Loop Analyses of Non-Degenerate Stops, JHEP 06 (2015) 028 [arXiv:1504.02409] [INSPIRE].
- [151] R. Huo, Effective Field Theory of Integrating out Sfermions in the MSSM: Complete One-Loop Analysis, Phys. Rev. D 97 (2018) 075013 [arXiv:1509.05942] [INSPIRE].
- [152] A. Drozd, J. Ellis, J. Quevillon and T. You, The Universal One-Loop Effective Action, JHEP 03 (2016) 180 [arXiv:1512.03003] [INSPIRE].
- [153] C.-W. Chiang and R. Huo, Standard Model Effective Field Theory: Integrating out a Generic Scalar, JHEP 09 (2015) 152 [arXiv:1505.06334] [INSPIRE].
- [154] H. Han, R. Huo, M. Jiang and J. Shu, Standard Model Effective Field Theory: Integrating out Neutralinos and Charginos in the MSSM, Phys. Rev. D 97 (2018) 095003 [arXiv:1712.07825] [INSPIRE].

- [155] S. Das Bakshi, J. Chakrabortty and S.K. Patra, CoDEx: Wilson coefficient calculator connecting SMEFT to UV theory, Eur. Phys. J. C 79 (2019) 21 [arXiv:1808.04403]
  [INSPIRE].
- [156] U. Haisch, M. Ruhdorfer, E. Salvioni, E. Venturini and A. Weiler, Singlet night in Feynman-ville: one-loop matching of a real scalar, JHEP 04 (2020) 164 [Erratum ibid. 07 (2020) 066] [arXiv:2003.05936] [INSPIRE].
- [157] V. Gherardi, D. Marzocca and E. Venturini, Matching scalar leptoquarks to the SMEFT at one loop, JHEP 07 (2020) 225 [Erratum ibid. 01 (2021) 006] [arXiv:2003.12525] [INSPIRE].
- [158] S.A.R. Ellis, J. Quevillon, P.N.H. Vuong, T. You and Z. Zhang, The Fermionic Universal One-Loop Effective Action, JHEP 11 (2020) 078 [arXiv:2006.16260] [INSPIRE].
- [159] M. Chala and A. Titov, One-loop matching in the SMEFT extended with a sterile neutrino, JHEP 05 (2020) 139 [arXiv:2001.07732] [INSPIRE].
- [160] J. Ellis, C.W. Murphy, V. Sanz and T. You, Updated Global SMEFT Fit to Higgs, Diboson and Electroweak Data, JHEP 06 (2018) 146 [arXiv:1803.03252] [INSPIRE].
- [161] E. da Silva Almeida, A. Alves, N. Rosa Agostinho, O.J.P. Éboli and M.C. Gonzalez-Garcia, Electroweak Sector Under Scrutiny: A Combined Analysis of LHC and Electroweak Precision Data, Phys. Rev. D 99 (2019) 033001 [arXiv:1812.01009] [INSPIRE].
- [162] A. Biekoetter, T. Corbett and T. Plehn, The Gauge-Higgs Legacy of the LHC Run II, SciPost Phys. 6 (2019) 064 [arXiv:1812.07587] [INSPIRE].
- [163] A. Falkowski and D. Straub, Flavourful SMEFT likelihood for Higgs and electroweak data, JHEP 04 (2020) 066 [arXiv:1911.07866] [INSPIRE].
- [164] ATLAS collaboration, Interpretations of the combined measurement of Higgs boson production and decay, Tech. Rep. ATLAS-CONF-2020-053 (2020).
- [165] A. Buckley et al., Constraining top quark effective theory in the LHC Run II era, JHEP 04 (2016) 015 [arXiv:1512.03360] [INSPIRE].
- [166] I. Brivio et al., O new physics, where art thou? A global search in the top sector, JHEP 02 (2020) 131 [arXiv:1910.03606] [INSPIRE].
- [167] S. Bißmann, J. Erdmann, C. Grunwald, G. Hiller and K. Kröninger, Constraining top-quark couplings combining top-quark and B decay observables, Eur. Phys. J. C 80 (2020) 136 [arXiv:1909.13632] [INSPIRE].
- [168] N.P. Hartland et al., A Monte Carlo global analysis of the Standard Model Effective Field Theory: the top quark sector, JHEP 04 (2019) 100 [arXiv:1901.05965] [INSPIRE].
- [169] G. Durieux et al., The electro-weak couplings of the top and bottom quarks Global fit and future prospects, JHEP 12 (2019) 98 [Erratum ibid. 01 (2021) 195] [arXiv:1907.10619]
  [INSPIRE].
- [170] S. van Beek, E.R. Nocera, J. Rojo and E. Slade, Constraining the SMEFT with Bayesian reweighting, SciPost Phys. 7 (2019) 070 [arXiv:1906.05296] [INSPIRE].
- [171] CMS Collaboration, Using associated top quark production to probe for new physics within the framework of effective field theory, tech. rep. CMS-PAS-TOP-19-001 (2020).
- [172] C. Degrande, G. Durieux, F. Maltoni, K. Mimasu, E. Vryonidou and C. Zhang, Automated one-loop computations in the SMEFT, arXiv:2008.11743 [INSPIRE].
- [173] I. Brivio, T. Corbett and M. Trott, The Higgs width in the SMEFT, JHEP 10 (2019) 056
  [arXiv:1906.06949] [INSPIRE].

- [174] F. Krauss, S. Kuttimalai and T. Plehn, LHC multijet events as a probe for anomalous dimension-six gluon interactions, Phys. Rev. D 95 (2017) 035024 [arXiv:1611.00767]
  [INSPIRE].
- [175] V. Hirschi, F. Maltoni, I. Tsinikos and E. Vryonidou, Constraining anomalous gluon self-interactions at the LHC: a reappraisal, JHEP 07 (2018) 093 [arXiv:1806.04696]
  [INSPIRE].
- [176] R. Goldouzian and M.D. Hildreth, LHC dijet angular distributions as a probe for the dimension-six triple gluon vertex, Phys. Lett. B 811 (2020) 135889 [arXiv:2001.02736]
  [INSPIRE].
- [177] S. Willenbrock and C. Zhang, Effective Field Theory Beyond the Standard Model, Ann. Rev. Nucl. Part. Sci. 64 (2014) 83 [arXiv:1401.0470] [INSPIRE].
- [178] A. Falkowski, Effective field theory approach to LHC Higgs data, Pramana 87 (2016) 39
  [arXiv:1505.00046] [INSPIRE].
- [179] I. Brivio and M. Trott, The Standard Model as an Effective Field Theory, Phys. Rept. 793 (2019) 1 [arXiv:1706.08945] [INSPIRE].
- [180] J.M. Gerard, Fermion mass spectrum in  $SU(2)_L \times U(1)$ , Z. Phys. C 18 (1983) 145 [INSPIRE].
- [181] R.S. Chivukula and H. Georgi, Composite Technicolor Standard Model, Phys. Lett. B 188 (1987) 99 [INSPIRE].
- [182] G. D'Ambrosio, G.F. Giudice, G. Isidori and A. Strumia, Minimal flavor violation: An effective field theory approach, Nucl. Phys. B 645 (2002) 155 [hep-ph/0207036] [INSPIRE].
- [183] D. Barducci et al., Interpreting top-quark LHC measurements in the standard-model effective field theory, arXiv:1802.07237 [INSPIRE].
- [184] O. Domenech, A. Pomarol and J. Serra, Probing the SM with Dijets at the LHC, Phys. Rev. D 85 (2012) 074030 [arXiv:1201.6510] [INSPIRE].
- [185] M. Farina, G. Panico, D. Pappadopulo, J.T. Ruderman, R. Torre and A. Wulzer, *Energy helps accuracy: electroweak precision tests at hadron colliders*, *Phys. Lett. B* 772 (2017) 210 [arXiv:1609.08157] [INSPIRE].
- [186] S. Dawson, P.P. Giardino and A. Ismail, Standard model EFT and the Drell-Yan process at high energy, Phys. Rev. D 99 (2019) 035044 [arXiv:1811.12260] [INSPIRE].
- [187] S. Alte, M. König and W. Shepherd, Consistent Searches for SMEFT Effects in Non-Resonant Dilepton Events, JHEP 07 (2019) 144 [arXiv:1812.07575] [INSPIRE].
- [188] J. Fuentes-Martin, A. Greljo, J. Martin Camalich and J.D. Ruiz-Alvarez, Charm physics confronts high-p<sub>T</sub> lepton tails, JHEP 11 (2020) 080 [arXiv:2003.12421] [INSPIRE].
- [189] R. Torre, L. Ricci and A. Wulzer, On the W&Y interpretation of high-energy Drell-Yan measurements, JHEP 02 (2021) 144 [arXiv:2008.12978] [INSPIRE].
- [190] C. Degrande, J.-M. Gerard, C. Grojean, F. Maltoni and G. Servant, Non-resonant New Physics in Top Pair Production at Hadron Colliders, JHEP 03 (2011) 125 [arXiv:1010.6304] [INSPIRE].
- [191] J. D'Hondt, A. Mariotti, K. Mimasu, S. Moortgat and C. Zhang, Learning to pinpoint effective operators at the LHC: a study of the ttbb signature, JHEP 11 (2018) 131
  [arXiv:1807.02130] [INSPIRE].

- [192] G. Banelli, E. Salvioni, J. Serra, T. Theil and A. Weiler, The Present and Future of Four Top Operators, JHEP 02 (2021) 043 [arXiv:2010.05915] [INSPIRE].
- [193] J. Ellis and T. You, Sensitivities of Prospective Future e<sup>+</sup>e<sup>-</sup> Colliders to Decoupled New Physics, JHEP 03 (2016) 089 [arXiv:1510.04561] [INSPIRE].
- [194] J. De Blas, G. Durieux, C. Grojean, J. Gu and A. Paul, On the future of Higgs, electroweak and diboson measurements at lepton colliders, JHEP 12 (2019) 117 [arXiv:1907.04311]
   [INSPIRE].
- [195] FCC collaboration, FCC-ee: The Lepton Collider: Future Circular Collider Conceptual Design Report Volume 2, Eur. Phys. J. ST 228 (2019) 261 [INSPIRE].
- [196] Z. Zhang, Time to Go Beyond Triple-Gauge-Boson-Coupling Interpretation of W Pair Production, Phys. Rev. Lett. 118 (2017) 011803 [arXiv:1610.01618] [INSPIRE].
- [197] B. Henning, D. Lombardo, M. Riembau and F. Riva, Measuring Higgs Couplings without Higgs Bosons, Phys. Rev. Lett. 123 (2019) 181801 [arXiv:1812.09299] [INSPIRE].
- [198] S. Bißmann, J. Erdmann, C. Grunwald, G. Hiller and K. Kröninger, Correlating uncertainties in global analyses within SMEFT matters, Phys. Rev. D 102 (2020) 115019 [arXiv:1912.06090] [INSPIRE].
- [199] M. Czakon, D. Heymes and A. Mitov, fastNLO tables for NNLO top-quark pair differential distributions, arXiv:1704.08551 [INSPIRE].
- [200] M.L. Czakon et al., NNLO versus NLO multi-jet merging for top-pair production including electroweak corrections, in 11th International Workshop on Top Quark Physics, in 11th International Workshop on Top Quark Physics, (2019) [arXiv:1901.04442] [INSPIRE].
- [201] M. Czakon et al., An exploratory study of the impact of CMS double-differential top distributions on the gluon parton distribution function, J. Phys. G 48 (2020) 015003 [arXiv:1912.08801] [INSPIRE].
- [202] I. Brivio, Y. Jiang and M. Trott, The SMEFTsim package, theory and tools, JHEP 12 (2017) 070 [arXiv:1709.06492] [INSPIRE].
- [203] R.D. Ball et al., Parton distributions with LHC data, Nucl. Phys. B 867 (2013) 244 [arXiv:1207.1303] [INSPIRE].
- [204] C. Englert and M. Spannowsky, Effective Theories and Measurements at Colliders, Phys. Lett. B 740 (2015) 8 [arXiv:1408.5147] [INSPIRE].
- [205] C. Hays, A. Helset, A. Martin and M. Trott, *Exact SMEFT formulation and expansion to*  $\mathcal{O}(v^4/\Lambda^4)$ , *JHEP* **11** (2020) 087 [arXiv:2007.00565] [INSPIRE].
- [206] A. Horne, J. Pittman, M. Snedeker, W. Shepherd and J.W. Walker, *Shift-Type SMEFT Effects in Dileptons at the LHC*, *JHEP* 03 (2021) 118 [arXiv:2007.12698] [INSPIRE].
- [207] E. Keilmann and W. Shepherd, Dijets at Tevatron Cannot Constrain SMEFT Four-Quark Operators, JHEP 09 (2019) 086 [arXiv:1907.13160] [INSPIRE].
- [208] C. Hays, A. Martin, V. Sanz and J. Setford, On the impact of dimension-eight SMEFT operators on Higgs measurements, JHEP 02 (2019) 123 [arXiv:1808.00442] [INSPIRE].
- [209] J. Baglio, S. Dawson, S. Homiller, S.D. Lane and I.M. Lewis, Validity of standard model EFT studies of VH and VV production at NLO, Phys. Rev. D 101 (2020) 115004 [arXiv:2003.07862] [INSPIRE].

- [210] V. Cirigliano, W. Dekens, J. de Vries and E. Mereghetti, Constraining the top-Higgs sector of the Standard Model Effective Field Theory, Phys. Rev. D 94 (2016) 034031
  [arXiv:1605.04311] [INSPIRE].
- [211] S. Carrazza, C. Degrande, S. Iranipour, J. Rojo and M. Ubiali, Can New Physics hide inside the proton?, Phys. Rev. Lett. 123 (2019) 132001 [arXiv:1905.05215] [INSPIRE].
- [212] ATLAS collaboration, Methodology for EFT interpretation of Higgs boson Simplified Template Cross-section results in ATLAS, Tech. Rep. ATL-PHYS-PUB-2019-042 (2019).
- [213] N. Berger et al., Simplified Template Cross Sections Stage 1.1, arXiv:1906.02754
  [INSPIRE].
- [214] O. Brein, A. Djouadi and R. Harlander, NNLO QCD corrections to the Higgs-strahlung processes at hadron colliders, Phys. Lett. B 579 (2004) 149 [hep-ph/0307206] [INSPIRE].
- [215] K. Mimasu, V. Sanz and C. Williams, Higher Order QCD predictions for Associated Higgs production with anomalous couplings to gauge bosons, JHEP 08 (2016) 039 [arXiv:1512.02572] [INSPIRE].
- [216] O. Bessidskaia Bylund, F. Maltoni, I. Tsinikos, E. Vryonidou and C. Zhang, Probing top quark neutral couplings in the Standard Model Effective Field Theory at NLO in QCD, JHEP 05 (2016) 052 [arXiv:1601.08193] [INSPIRE].
- [217] C. Englert, R. Rosenfeld, M. Spannowsky and A. Tonero, New physics and signal-background interference in associated pp → HZ production, EPL 114 (2016) 31001 [arXiv:1603.05304] [INSPIRE].
- [218] A. Azatov, C. Grojean, A. Paul and E. Salvioni, Resolving gluon fusion loops at current and future hadron colliders, JHEP 09 (2016) 123 [arXiv:1608.00977] [INSPIRE].
- [219] E. Vryonidou and C. Zhang, Dimension-six electroweak top-loop effects in Higgs production and decay, JHEP 08 (2018) 036 [arXiv:1804.09766] [INSPIRE].
- [220] ATLAS collaboration, Higgs boson production cross-section measurements and their EFT interpretation in the 4ℓ decay channel at √s = 13 TeV with the ATLAS detector, Eur. Phys. J. C 80 (2020) 957 [Erratum ibid. 81 (2021) 29] [arXiv:2004.03447] [INSPIRE].
- [221] M. Czakon, D. Heymes, A. Mitov, D. Pagani, I. Tsinikos and M. Zaro, Top-quark charge asymmetry at the LHC and Tevatron through NNLO QCD and NLO EW, Phys. Rev. D 98 (2018) 014003 [arXiv:1711.03945] [INSPIRE].
- [222] D. Liu, A. Pomarol, R. Rattazzi and F. Riva, Patterns of Strong Coupling for LHC Searches, JHEP 11 (2016) 141 [arXiv:1603.03064] [INSPIRE].
- [223] PARTICLE DATA GROUP collaboration, Review of Particle Physics, PTEP 2020 (2020) 083C01 [INSPIRE].
- [224] A. Azatov, R. Contino, C.S. Machado and F. Riva, *Helicity selection rules and noninterference for BSM amplitudes*, *Phys. Rev. D* 95 (2017) 065014 [arXiv:1607.05236]
  [INSPIRE].
- [225] CMS collaboration, Inclusive search for a highly boosted Higgs boson decaying to a bottom quark-antiquark pair, Phys. Rev. Lett. **120** (2018) 071802 [arXiv:1709.05543] [INSPIRE].
- [226] R. Contino, A. Falkowski, F. Goertz, C. Grojean and F. Riva, On the Validity of the Effective Field Theory Approach to SM Precision Tests, JHEP 07 (2016) 144 [arXiv:1604.06444] [INSPIRE].

- [227] C. Degrande, F. Maltoni, K. Mimasu, E. Vryonidou and C. Zhang, Single-top associated production with a Z or H boson at the LHC: the SMEFT interpretation, JHEP 10 (2018) 005 [arXiv:1804.07773] [INSPIRE].
- [228] J.A. Dror, M. Farina, E. Salvioni and J. Serra, Strong tW Scattering at the LHC, JHEP 01 (2016) 071 [arXiv:1511.03674] [INSPIRE].
- [229] F. Maltoni, L. Mantani and K. Mimasu, Top-quark electroweak interactions at high energy, JHEP 10 (2019) 004 [arXiv:1904.05637] [INSPIRE].
- [230] E.H. Simmons, Higher dimension gluon operators and hadronic scattering, Phys. Lett. B 246 (1990) 471 [INSPIRE].
- [231] H.K. Dreiner, A. Duff and D. Zeppenfeld, How well do we know the three gluon vertex?, Phys. Lett. B 282 (1992) 441 [INSPIRE].
- [232] P.L. Cho and E.H. Simmons, Searching for G3 in tt production, Phys. Rev. D 51 (1995)
  2360 [hep-ph/9408206] [INSPIRE].
- [233] D. Ghosh and M. Wiebusch, Dimension-six triple gluon operator in Higgs+jet observables, Phys. Rev. D 91 (2015) 031701 [arXiv:1411.2029] [INSPIRE].
- [234] M. Czakon, D. Heymes, A. Mitov, D. Pagani, I. Tsinikos and M. Zaro, Top-pair production at the LHC through NNLO QCD and NLO EW, JHEP 10 (2017) 186 [arXiv:1705.04105] [INSPIRE].
- [235] D. Bardhan, D. Ghosh, P. Jain and A.M. Thalapillil, Towards constraining triple gluon operators through tops, arXiv:2010.13402 [INSPIRE].
- [236] A. Crivellin, F. Kirk, C.A. Manzari and M. Montull, Global Electroweak Fit and Vector-Like Leptons in Light of the Cabibbo Angle Anomaly, JHEP 12 (2020) 166 [arXiv:2008.01113] [INSPIRE].
- [237] J.R. Espinosa, C. Grojean, V. Sanz and M. Trott, NSUSY fits, JHEP 12 (2012) 077 [arXiv:1207.7355] [INSPIRE].
- [238] A. Banfi, A. Bond, A. Martin and V. Sanz, Digging for Top Squarks from Higgs data: from signal strengths to differential distributions, JHEP 11 (2018) 171 [arXiv:1806.05598]
  [INSPIRE].
- [239] F. Feroz, M.P. Hobson and M. Bridges, MultiNest: an efficient and robust Bayesian inference tool for cosmology and particle physics, Mon. Not. Roy. Astron. Soc. 398 (2009) 1601 [arXiv:0809.3437] [INSPIRE].
- [240] J. Buchner et al., X-ray spectral modelling of the AGN obscuring region in the CDFS: Bayesian model selection and catalogue, Astron. Astrophys. 564 (2014) A125 [arXiv:1402.0004] [INSPIRE].
- [241] C. Hays, V. Sanz Gonzalez and G. Zemaityte, Constraining EFT parameters using simplified template cross sections, Tech. Rep. LHCHXSWG-2019-004 (2019).
- [242] A. Falkowski, B. Fuks, K. Mawatari, K. Mimasu, F. Riva and V. Sanz, Rosetta: an operator basis translator for Standard Model effective field theory, Eur. Phys. J. C 75 (2015) 583 [arXiv:1508.05895] [INSPIRE].